BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

In the Matter of the Joint Application of Sprint Communications Company L.P. (U-5112) and T-Mobile USA, Inc., a Delaware Corporation, For Approval of Transfer of Control of Sprint Communications Company L.P. Pursuant to California Public Utilities Code Section 854(a).

Application 18-07-011

In the Matter of the Joint Application of Sprint Spectrum L.P. (U-3062-C), and Virgin Mobile USA, L.P. (U-4327-C) and T-Mobile USA, Inc., a Delaware Corporation for Review of Wireless Transfer Notification per Commission Decision 95-10-032.

Application 18-07-012

MOTION OF JOINT APPLICANTS TO ADVISE THE COMMISSION OF DOJ PROPOSED FINAL JUDGMENT

Dave Conn Susan Lipper T-Mobile USA, Inc. 12920 SE 38th St. Bellevue, WA 98006 Telephone: 425.378.4000

Facsimile: 425.378.4040 Email: dave.conn@t-mobile.com

Email: susan.lipper@t-mobile.com

Suzanne Toller Davis Wright Tremaine LLP 505 Montgomery Street, Suite 800 San Francisco, CA 94111 Telephone: 415.276.6500

Email: suzannetoller@dwt.com

Leon M. Bloomfield Law Offices of Leon M. Bloomfield 1901 Harrison St., Suite 1400 Oakland, CA 94612

Telephone: 510.625.1164 Email: lmb@wblaw.net

Attorneys for T-Mobile USA, Inc.

Stephen H. Kukta Sprint 900 7th Street, NW, Suite 700 Washington, DC 20001 Telephone: 415.572.8358

Email: stephen.h.kukta@sprint.com

Earl Nicholas Selby Law Offices of Earl Nicholas Selby 530 Lytton Avenue, 2nd Floor Palo Alto, CA 94301 Telephone: 650.323.0990

Telephone: 650.323.0990 Facsimile: 650.325.9041

Email: selbytelecom@gmail.com

Attorneys for Sprint Spectrum L.P. (U-3062-C), Virgin Mobile USA, L.P. (U-4327-C)

BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

In the Matter of the Joint Application of Sprint Communications Company L.P. (U-5112) and T-Mobile USA, Inc., a Delaware Corporation, For Approval of Transfer of Control of Sprint Communications Company L.P. Pursuant to California Public Utilities Code Section 854(a).

Application 18-07-011

In the Matter of the Joint Application of Sprint Spectrum L.P. (U-3062-C), and Virgin Mobile USA, L.P. (U-4327-C) and T-Mobile USA, Inc., a Delaware Corporation for Review of Wireless Transfer Notification per Commission Decision 95-10-032.

Application 18-07-012

MOTION OF JOINT APPLICANTS TO ADVISE THE COMMISSION OF DOJ PROPOSED FINAL JUDGMENT

Pursuant to Rule 11.1 of the Commission's Rules of Practice and Procedure ("Rules"), Sprint Spectrum L.P. (U-3062-C), Virgin Mobile USA, L.P. (U-4327-C) (collectively, "Sprint"), and T-Mobile USA, Inc. ("T-Mobile USA") (collectively, the "Joint Applicants") respectfully submit this Motion to advise the Commission of the terms of a proposed Final Judgment ("Proposed Final Judgment") and related Stipulation and Order ("Stipulation & Order") filed by the U.S. Department of Justice ("DOJ") on July 26, 2019 in the U.S. District Court for the District of Columbia.

The Proposed Final Judgment and Stipulation & Order reflect conditions accepted by, among others, T-Mobile USA and Sprint Corporation (the "Parties") to resolve the potential competition-related questions raised by the DOJ in connection with its review of the proposed merger between the Parties (the "Transaction"). (A copy of the Proposed Final Judgment, which is subject to judicial review under the Tunney Act, 15 U.S.C. § 16(b), and is contingent on closing the Transaction, is attached hereto as Exhibit 1. A copy of the Stipulation & Order is attached

hereto as Exhibit 2.) As described more fully below, under the Proposed Final Judgment, DISH Network Corporation ("DISH") – a party to the Proposed Final Judgment and Order & Stipulation – will acquire Sprint's prepaid wireless businesses (excluding the Assurance Wireless LifeLine business¹) and obtain additional rights that will strengthen DISH's ability to compete in the retail mobile wireless services market. DISH also commits to offer consumers retail mobile wireless services (including postpaid wireless services) and to deploy a nationwide 5G broadband network.²

As reflected in Joint Applicants' briefs, the record overwhelmingly demonstrates that the Transaction, as originally structured, is pro-competitive and will benefit consumers and the public by enabling New T-Mobile to build a world-leading 5G wireless network with greater capacity, better coverage, and faster speeds, and that will give consumers better service and lower prices. Further enhancing those enormous public-interest benefits from the New T-Mobile network and its business plans, Joint Applicants have offered more than 50 voluntary commitments in this proceeding.³ Moreover, Joint Applicants have advised the Commission of additional nationwide commitments made to the Federal Communications Commission ("FCC") that are backed up by a robust enforcement mechanism.⁴ Based on those commitments and the record before the FCC, the FCC Chairman and two of the four other FCC Commissioners have stated that they intend to

_

¹ DISH's acquisition also excludes the prepaid wireless customers of Shenandoah Telecommunications Company and Swiftel Communications, Inc.

² See infra at 3-4.

³ See Joint Applicants' Post-Hearing Reply Brief on the Joint Application for Review of Wireless Transfer Notification Per Commission Decision 95-10-032 (filed May 10, 2019) ("Joint Applicants' Wireless Reply Brief") at 2-3; see also id., Appendix 1; Motion of Joint Applicants to Advise the Commission of New FCC Commitments (filed May 20, 2019) ("Motion to Advise Commission of FCC Commitments") at 3.

⁴ See generally Motion to Advise Commission of FCC Commitments. The FCC enforcement mechanism includes strong verification measures, substantial voluntary contributions for any missed deadlines, and continued accrual of the voluntary contributions until any unmet obligations are fulfilled. See id. at 2 & n.5.

approve the Transaction pursuant to the FCC's exclusive authority over national wireless licensing.⁵

The DOJ's Proposed Final Judgment further demonstrates that the wireless transfers involved in this proceeding⁶ are in the public interest and that the Commission should conclude its review of those transfers.⁷ The Proposed Final Judgment, including the divestures and other protections summarized below, only reinforces the conclusion that the Transaction is procompetitive.⁸ Among other things, the Proposed Final Judgment includes the following key features:

• As part of extensive mandated divestitures, DISH will acquire Sprint's Boost, ⁹ Virgin Mobile and Sprint-branded prepaid businesses and millions of Boost and Virgin Mobile customers. Ex. 1 at §§ II.L, IV. (The divestitures exclude the Assurance brand Lifeline business, and New T-

⁵ Chairman Pai Statement on T-Mobile/Sprint Transaction (May 20, 2019), available at https://docs.fcc.gov/public/attachments/DOC-357535A1.pdf; Commissioner Carr Announces Support for T-Mobile – Sprint Transaction (May 20, 2019), available at https://docs.fcc.gov/public/attachments/DOC-357536A1.pdf; May 20, 2019 2:16 p.m. Tweet by Commissioner Michael O'Rielly ("I am inclined to support T-Mobile/Sprint proposed merger, even if not convinced of the need for all the newly announced conditions being proposed."), available at https://twitter.com/mikeofcc/status/1130583036595179520.

⁶ See Joint Applicants' Post-Hearing Opening Brief on the Joint Application for Review of Wireless Transfer Notification Per Commission Decision 95-10-032 (filed Apr. 26, 2019) ("Joint Applicants' Wireless Opening Brief") at 8-10 (describing the "straightforward, parent-level transfer of control of the Sprint Wireless CA Entities," Sprint Spectrum L.P. and Virgin Mobile USA, L.P., to T-Mobile USA).

⁷ Nothing in the DOJ's Proposed Final Judgment will prevent or hinder New T-Mobile from fulfilling its buildout or other commitments.

⁸ See, e.g., Joint Applicants' Wireless Opening Brief at 51-80; Joint Applicants' Wireless Reply Brief at 30-57; see also Joint Applicants' Post-Hearing Opening Brief Requesting Immediate Approval of the Transfer of Sprint Communications Company L.P. to T-Mobile USA, Inc. (April 26, 2019) ("Joint Applicants' Wireline Opening Brief") at 2, 14-15, 17-19; Joint Applicants' Post-Hearing Reply Brief Requesting Immediate Approval of the Transfer of Sprint Communications Company L.P. to T-Mobile USA, Inc. (filed May 10, 2019") ("Joint Applicants' Wireline Reply Brief") at 2.

⁹ As Joint Applicants explained in their May 20, 2019 *Motion to Advise the Commission of New FCC Commitments*, "the FCC Commitment to divest and sell Boost does not impact its California commitment to the Boost Pilot program after the merger. T-Mobile reaffirms its strong support for that program and its willingness to work with the Commission staff to determine how to best support the pilot, including but not limited to having one of New T-Mobile's brands (e.g. MetroPCS) assume Boost's responsibilities under the pilot."

Mobile will continue to provide LifeLine service under that brand. *See* Ex. 1 at § II.L.) To facilitate DISH's emergence as a new, competing provider, T-Mobile and Sprint have agreed to provide DISH extensive support to ensure a smooth and orderly transition. Ex. 1 at § IV.A.

- DISH has the right to offer jobs to Sprint's Prepaid Asset Personnel (consistent with employee rights and employment laws), and New T-Mobile is obligated to facilitate that hiring process and the transition of employees. *Id.* at § IV.A.2.
- DISH also has agreed to buy all of Sprint's 800 MHz spectrum licenses. *Id.* at § IV.B
- DISH will have the option to assume select cell sites and retail locations that are decommissioned by New T-Mobile for five years following the closing of the divestiture transaction, subject to any assignment restrictions, and New T-Mobile will make available to DISH at least 20,000 decommissioned T-Mobile USA and Sprint cell sites, and at least 400 retail stores. *Id.* at § IV.C-D.
- DISH is obligated to use the divested assets to "offer retail mobile wireless services, including offering nationwide postpaid retail mobile wireless service" within one year of the sale of the prepaid assets. *Id.* at § IV.F.
- DISH is also obligated to deploy a nationwide 5G broadband network and comply with various additional FCC requirements to utilize its extensive spectrum holdings, subject to specific deadlines and backed by penalties. *Id.* at § VIII.A.
- Upon closing of the divestiture transaction, DISH and New T-Mobile will enter into a 7-year Mobile Virtual Network Operator ("MVNO") agreement that allows DISH to sell retail wireless services under any DISH owned brands using New T-Mobile's network. In addition, DISH will be entitled to transition the acquired Boost, Virgin Mobile and Sprint-branded prepaid customers to New T-Mobile's network and activate new customers on New T-Mobile's network. *Id.* at §§ IV.A, VI. The terms of the DISH MVNO must be "commercially reasonable and … acceptable to the [DOJ]." *Id.* at § VI.A.
- Moreover, New T-Mobile will offer standard commercial support arrangements to DISH via a Transition Services Agreement for up to 3 years following the close of the divestiture transaction. The transition services provided by New T-Mobile will result in the orderly transfer of prepaid customers to DISH and will also ensure the continued and seamless operation of the Boost Mobile, Virgin Mobile, and Sprint-branded prepaid businesses following transition to DISH's ownership. New T-Mobile must not unreasonably discriminate against any DISH subscribers, and must not

- unreasonably refuse to allow devices used by DISH customers to access the New T-Mobile network. *Id.* at § VI.B.
- New T-Mobile must honor all existing T-Mobile and Sprint MVNO agreements and agree to certain extensions of such agreements. *Id.* at § VII.
- New T-Mobile must support eSIM technology on smartphones (to the extent technically feasible), and allow "unlocking" of mobile devices. *Id.* at § VII.
- A Monitoring Trustee will be appointed to ensure compliance with the terms of the Proposed Final Judgment. *Id.* at § XII.

On July 26, 2019, T-Mobile USA entered into an Asset Purchase Agreement with Sprint Corporation and DISH that implements many of the terms of the Proposed Final Judgment. (The terms of the Asset Purchase Agreement are described in greater detail in a Form 8-K filed by T-Mobile USA with the Securities and Exchange Commission on July 26, 2019. A copy of the Form 8-K is attached hereto as Exhibit 3.)

Based on the extensive record before this Commission – now amplified by the FCC commitments and DOJ Proposed Final Judgment – Joint Applicants respectfully reiterate that the Commission should promptly conclude its review of the wireless transfers. ¹⁰ Joint Applicants note that this proceeding has now been pending for over a year, and that while the Commission's schedule called for a Proposed Decision by no later than June 30, 2019, ¹¹ no such Decision has yet been issued. Accordingly, Joint Applicants respectfully urge the Commission to issue a Proposed Decision concluding its review of the Wireless Transfer Notification without further delay.

¹⁰ See, e.g., Joint Applicants' Wireless Opening Brief at 14-16 (the Wireless Notification is subject to the Commission's review and not preapproval); Joint Applicants' Wireless Reply Brief at 7-12 (same).

¹¹ Amended Assigned Commissioner's Scoping Memo & Ruling (Oct. 4, 2018) at 4 (adopting schedule providing for a Proposed Decision by "2nd Quarter 2019").

Moreover, the DOJ's Proposed Final Judgment addresses only wireless-related issues and

has no bearing on the Wireline Approval Application that remains pending before the Commission.

As Joint Applicants have previously explained, there is no sound legal or policy reason to delay a

decision approving that Application. The record unequivocally shows that the Wireline Approval

Application readily satisfies the Commission's well-established standard for approving wireline

transfers under California Public Utilities Code § 854(a), and the Intervenors have not presented

any evidence or argument to the contrary. Accordingly, Joint Applicants respectfully reiterate that

the Wireline Approval Application – which raises distinct legal and factual issues from the

Wireless Notification – should be granted immediately.

Respectfully submitted this 26th day of July, 2019.

Suzanne Toller¹²

Davis Wright Tremaine LLP

505 Montgomery Street, Suite 800

San Francisco, CA 94111 Telephone: 415.276.6500

Email: suzannetoller@dwt.com

Attorneys for T-Mobile USA, Inc.

¹² Pursuant to Rule 1.8(d), this document is signed on behalf of Joint Applicants.

6

EXHIBIT 1

PROPOSED FINAL JUDGMENT

UNITED STATES DISTRICT COURT FOR THE DISTRICT OF COLUMBIA

UNITED STATES OF AMERICA et al.,

Plaintiffs,

v.

DEUTSCHE TELEKOM AG, T-MOBILE US, INC., SOFTBANK GROUP CORP., SPRINT CORPORATION, and DISH NETWORK CORPORATION,

Defendants.

Case No.

Filed:

[PROPOSED] FINAL JUDGMENT

WHEREAS, Plaintiffs, United States of America and the States of Kansas, Nebraska,
Ohio, Oklahoma, and South Dakota ("Plaintiff States"), filed their Complaint on July 26, 2019,
the Plaintiffs and Defendants Deutsche Telekom AG, T-Mobile US, Inc., SoftBank Group Corp.,
and Sprint Corp., by their respective attorneys, have consented to the entry of this Final
Judgment without trial or adjudication of any issue of fact or law, and without this Final
Judgment constituting any evidence against or admission by any party regarding any issue of fact
or law;

AND WHEREAS, pursuant to a Stipulation and Order among Deutsche Telekom AG, T-Mobile US, Inc., SoftBank Group Corp., Sprint Corp., and DISH Network Corp. (collectively, "Defendants") and the United States, the Court has joined DISH Network Corp. as a defendant to this action for the purposes of settlement and for the entry of this Final Judgment;

AND WHEREAS, Defendants agree to be bound by the provisions of this Final Judgment pending its approval by the Court;

AND WHEREAS, the purpose of this Final Judgment is to preserve competition by enabling the entry of another national facilities-based mobile wireless network operator;

AND WHEREAS, Plaintiffs require Divesting Defendants to make certain divestitures for the purpose of remedying the loss of competition alleged in the Complaint;

AND WHEREAS, Defendants have represented to Plaintiffs that the divestitures and other relief required by this Final Judgment can and will be made and carried out, and that Defendants will not later raise any claim of hardship or difficulty as grounds for asking the Court to modify any of the provisions contained below;

NOW THEREFORE, before any testimony is taken, without trial or adjudication of any issue of fact or law, and upon consent of the parties, it is ORDERED, ADJUDGED, AND DECREED:

I. <u>JURISDICTION</u>

The Court has jurisdiction over the subject matter of and each of the parties to this action. The Complaint states a claim upon which relief may be granted against Divesting Defendants and Parent Defendants under Section 7 of the Clayton Act, 15 U.S.C. § 18. Pursuant to the Stipulation and Order filed simultaneously with this Final Judgment joining DISH as a defendant to this action, DISH has consented to this Court's exercise of specific personal jurisdiction over DISH in this matter solely for the purposes of settlement and for the entry and enforcement of the Final Judgment.

II. DEFINITIONS

As used in this Final Judgment:

- A. "Acquiring Defendant" or "Acquirer" or "DISH" mean Defendant DISH Network Corporation, a Nevada corporation with its headquarters in Englewood, Colorado; its successors and assigns; and its subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- B. "Assurance Wireless" means the prepaid wireless business conducted by Virgin Mobile under the Assurance Lifeline brand.
- C. "Cell Site" or "Tower Site" mean any wireless communications towers, rooftops, water towers, or other wireless communications facilities owned or leased by Divesting Defendants and the physical location and wireless equipment thereto.
- D. "Decommissioned" or "Decommissioning" means, with respect to a Cell Site, when the Cell Site is no longer transmitting on Divesting Defendants' networks. With respect to Retail Locations, Decommissioned or Decommissioning means when Divesting Defendants cease customer service operations.
- E. "Deutsche Telekom" means Deutsche Telekom AG, a German corporation headquartered in Bonn, Germany, that is the controlling shareholder of T-Mobile; its successors and assigns; and its parents, subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
 - F. "Divesting Defendants" means T-Mobile and Sprint.
- G. "Divestiture Assets" means the Prepaid Assets, the 800 MHz Spectrum Licenses, the Decommissioned Retail Locations, and the Decommissioned Cell Sites.

- H. "Fifth Generation Broadband Services" or "5G Services" means at least 3GPP Release 15, capable of providing enhanced mobile broadband (eMBB) functionality.
- I. "Full MVNO Agreement" means an agreement that (1) provides the Acquiring Defendant the ability to sell retail mobile wireless services as an MVNO using the Divesting Defendants' wireless networks, (2) provides Acquiring Defendant the option to deploy its own core network with all associated service platforms to be offered in combination with services provided by Divesting Defendants' wireless networks, and (3) requires Divesting Defendants to provide network connectivity between Divesting Defendants and Acquiring Defendant's network for all traffic.
- J. "MVNO" means a mobile virtual network operator, such as TracFone and Google Fi, that obtains network access from facilities-based providers like T-Mobile and Sprint and resells that mobile wireless service to consumers under its own brand name.
 - K. "Parent Defendants" means Deutsche Telekom and SoftBank.
- L. "Prepaid Assets" means all tangible and intangible assets primarily used by the Boost Mobile, Sprint-branded prepaid, and Virgin Mobile businesses today, including but not limited to Boost and Virgin Mobile Retail Locations, licenses, personnel, facilities, data, and intellectual property, as well as all relationships and/or contracts with prepaid customers served by Sprint, Boost Mobile, and Virgin Mobile. Prepaid Assets do not include the Assurance Wireless business and the prepaid wireless customers of Shenandoah Telecommunications Company and Swiftel Communications, Inc.
- M. "Prepaid Assets Personnel" means all employees whose jobs currently focus on the support of the Prepaid Assets, or whose jobs have previously focused on supporting the

Prepaid Assets at any time between January 1, 2016 and the date on which the Prepaid Assets are divested to the Acquirer. Prepaid Assets Personnel shall include no fewer than 400 current employees of the Divesting Defendants, which shall include employees involved in sales management, marketing management, distribution support, sales support, and finance.

- N. "Retail Locations" means any retail locations owned or operated by Divesting

 Defendants and from which either T-Mobile or Sprint sells mobile wireless service under any of
 their affiliated brands, including Sprint, Boost Mobile, Virgin Mobile, T-Mobile, Metro by

 T-Mobile, and MetroPCS.
- O. "800 MHz Spectrum Licenses" means all of Sprint's 800 MHz spectrum holdings as listed and described in Attachment A to this Final Judgment.
- P. "600 MHz Spectrum Licenses" means all of DISH's 600 MHz spectrum holdings as listed and described in Attachment B to this Final Judgment.
- Q. "SoftBank" means SoftBank Group Corp., a Japanese corporation and controlling shareholder of Sprint; its successors and assigns; and its parents, subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- R. "Sprint" means Defendant Sprint Corporation, a Delaware corporation with its headquarters in Overland Park, Kansas; its successors and assigns; and its subsidiaries, divisions, groups, affiliates (other than SoftBank), partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- S. "T-Mobile" means Defendant T-Mobile US, Inc., a Delaware corporation with its headquarters in Bellevue, Washington; its successors and assigns; and its subsidiaries, divisions,

groups, affiliates (other than Deutsche Telekom), partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.

III. APPLICABILITY

- A. This Final Judgment applies to the Divesting Defendants, Parent Defendants, and Acquiring Defendant, as defined above, and all other persons in active concert or participation with any of them who receive actual notice of this Final Judgment by personal service or otherwise.
- B. If any of the terms of an agreement between (i) Divesting Defendants and the Acquiring Defendant to effectuate the divestitures required by the Final Judgment or (ii) Defendants and the Federal Communications Commission (FCC) to effectuate the divestitures required by the Final Judgment varies from the terms of this Final Judgment then, to the extent that Defendants cannot fully comply with both terms due to a conflict between the terms, this Final Judgment will determine Defendants' obligations. Provided, however, that if there is an inconsistency between this Final Judgment and any commitment any of the Defendants have made to the FCC, the more stringent obligations will control.

IV. DIVESTITURES

A. **Prepaid Assets**

1. The Divesting Defendants shall take all actions required to enable

Acquiring Defendant to have, within ninety (90) days after notice of the entry of this

Final Judgment by the Court, the ability to provision any new or existing customer of the

Prepaid Assets holding a compatible handset device onto the T-Mobile network pursuant
to the terms of any Full MVNO Agreement. Divesting Defendants are ordered and

directed, not more than fifteen (15) days after Divesting Defendants can provide

Acquiring Defendant the ability to provision any new or existing customer of the Prepaid

Assets holding a compatible handset device onto the T-Mobile network pursuant to the

terms of any Full MVNO Agreement, or the first business day of the month following the

later of the consummation of the merger of T-Mobile and Sprint and the receipt of any

approvals required for the divestiture of the Prepaid Assets from the FCC and any

material state public utility commission, or five (5) calendar days after notice of the entry

of this Final Judgment by the Court, whichever is later, to divest the Prepaid Assets to

Acquiring Defendant in a manner acceptable to the United States, in its sole discretion.

2. <u>Employees</u>

a. Within ten (10) business days following the filing of the Complaint in this matter, Divesting Defendants shall provide to Acquiring Defendant, the United States, the Plaintiff States, and the Monitoring Trustee, organization charts covering all Prepaid Assets Personnel for each year from January 1, 2016 to present. Within ten (10) business days of receiving a request from Acquiring Defendant, Divesting Defendants shall provide to Acquiring Defendant, the United States, the Plaintiff States, and the Monitoring Trustee, additional information related to identified Prepaid Assets Personnel, including name, job title, reporting relationships, past experience, responsibilities from January 1, 2016 through the date on which the Prepaid Assets are transferred to Acquirer, training and educational history, relevant certifications, job performance evaluations, and current salary and benefits information to enable Acquiring

Defendant to make offers of employment. If Divesting Defendants are barred by any applicable laws from providing any of this information to Acquiring Defendant, within ten (10) business days of receiving Acquiring Defendant's request, Divesting Defendants will provide the requested information to the greatest extent possible under applicable laws and also provide a written explanation of their inability to comply fully with Acquiring Defendant's request for information regarding Prepaid Assets Personnel.

b. Upon request, Divesting Defendants shall make Prepaid Assets
Personnel available for interviews with Acquiring Defendant during normal
business hours at a mutually agreeable location. Divesting Defendants will not
interfere with any negotiations by Acquiring Defendant to employ any Prepaid
Assets Personnel. Interference includes but is not limited to offering to increase
the salary or benefits of or offering bonuses to Prepaid Assets Personnel other
than as part of a company-wide increase in salary or benefits or company-wide
provision of bonuses granted in the ordinary course of business. If Divesting
Defendants have offered Prepaid Assets Personnel incentives to remain employed
with Divesting Defendants until a certain date (e.g., retention bonuses), Divesting
Defendant must warrant to those Prepaid Assets Personnel and the Acquiring
Defendant that the Prepaid Assets Personnel will receive all promised incentives
if they accept an offer of employment with the Acquiring Defendant and remain
employed with the Acquiring Defendant until the date contemplated by the

originally agreed-upon incentive. Divesting Defendants shall be responsible for reimbursing Acquiring Defendant the costs associated with such incentives.

- c. For any Prepaid Assets Personnel who elect employment with Acquiring Defendant, Divesting Defendants shall waive all non-compete and non-disclosure agreements, vest all unvested pension and other equity rights, and provide all benefits to which Prepaid Assets Personnel would be provided if transferred to a buyer of an ongoing business.
- d. For a period of two (2) years from the date of filing of the Complaint in this matter, Divesting Defendants may not solicit to hire, or hire, any Prepaid Assets Personnel who was hired by Acquiring Defendant, unless (a) such individual is terminated or laid off by Acquiring Defendant or (b) Acquiring Defendant agrees in writing that Divesting Defendants may solicit or hire that individual.
- e. Nothing in this Section prohibits Divesting Defendants from maintaining any reasonable restrictions on the disclosure by any employee who accepts an offer of employment with Acquiring Defendant of Divesting Defendants' proprietary non-public information that is (a) not otherwise required to be disclosed by this Final Judgment, (b) related solely to Divesting Defendant's businesses and clients, and (c) unrelated to the Divestiture Assets.
- f. Acquiring Defendant's right to hire Prepaid Assets Personnel pursuant to Paragraph IV(A)(2) and Divesting Defendants' obligations under

Paragraphs IV(A)(2)(a)-(c) lasts for a period of one hundred and eighty (180) days after the closing of the divestiture of the Prepaid Assets.

- 3. Divesting Defendants shall warrant to Acquiring Defendant that the Prepaid Assets will be fully operational on the date of transfer.
- 4. At the option of Acquiring Defendant, Divesting Defendants shall enter into one or more transition services agreements to provide billing, customer care, SIM card procurement, device provisioning, and all other services used by the Prepaid Assets prior to the date of their transfer to Acquirer for an initial period of up to two (2) years after the transfer of the Prepaid Assets. During the initial two-year term of the agreement, Divesting Defendants shall provide the transition services at no greater than cost to Acquiring Defendant. All other terms and conditions of any such agreement must be reasonably related to market conditions for the provision of the relevant services and must be acceptable to the United States in its sole discretion, after consultation with the affected Plaintiff States. Upon Acquiring Defendant's request, the United States, in its sole discretion, after consultation with the affected Plaintiff States, may approve one or more extensions of such agreement(s) for a total of up to an additional one (1) year.
- 5. At Acquiring Defendant's option, on or before the divestiture of the Prepaid Assets, Divesting Defendants shall assign or otherwise transfer to Acquiring Defendant all transferable or assignable agreements, or any assignable portions thereof, related to the Prepaid Assets, including, but not limited to, all supply contracts, licenses, and collaborations. Divesting Defendants shall use best efforts to expeditiously obtain from any third parties any consent necessary to transfer or assign to Acquiring Defendant

all agreements related to the Prepaid Assets. To the extent consent cannot be obtained and the agreement is not otherwise assignable, Divesting Defendants shall use best efforts to obtain or provide for Acquiring Defendant, as expeditiously as possible, the full benefits of any such agreement as it relates to the Prepaid Assets by assisting Acquiring Defendant to secure a new agreement and by taking any other steps necessary to ensure that Acquiring Defendant obtains the full benefit of the agreement as it relates to the Prepaid Assets. Divesting Defendants will not assert, directly or indirectly, any legal claim that would interfere with Acquiring Defendant's ability to obtain the full benefit from any transferred third-party agreement to the same extent enjoyed by Divesting Defendant prior to the transfer.

6. At Acquiring Defendant's option, on or before the divestiture of the Prepaid Assets, Divesting Defendants shall provide contact information and make introductions to distributors and suppliers that support the Prepaid Assets. Divesting Defendants shall not interfere with Acquiring Defendant's attempts to negotiate with these distributors or suppliers.

B. **800 MHz Spectrum License Transfer**

1. Divesting Defendants are ordered and directed, within three (3) years after the closing of the divestiture of the Prepaid Assets or within five (5) business days of the approval by the FCC of the transfer of the 800 MHz Spectrum Licenses, whichever is later, to divest the 800 MHz Spectrum Licenses in a manner acceptable to the United States, in its sole discretion, after consultation with the affected Plaintiff States. The United States, in its sole discretion, after consultation with the affected Plaintiff States,

may agree to one or more extensions of this time period not to exceed sixty (60) calendar days in total, and will notify the Court in such circumstances. Acquiring Defendant will make timely application to the FCC for the transfer of the spectrum to comply with this Paragraph.

- 2. Acquiring Defendant shall pay a penalty of \$360,000,000 to the United States if it elects not to purchase the 800 MHz Spectrum Licenses. The Acquiring Defendant shall pay the penalty within thirty (30) days of declining to purchase the 800 MHz Spectrum Licenses. Notwithstanding the foregoing, the Acquiring Defendant will not be required to pay such penalty if it has deployed a core network and offered 5G Service to at least 20% of the U.S. population over DISH's facilities-based network within three (3) years of the closing of the divestiture of the Prepaid Assets.
- 3. If, at the expiration of this Final Judgment, Acquiring Defendant has acquired the 800 MHz Spectrum Licenses, but has not deployed all of the 800 MHz Spectrum Licenses for use in the provision of retail mobile wireless services, Acquiring Defendant shall forfeit to the FCC, at the United States' sole discretion, after consultation with the affected Plaintiff States, all of the 800 MHz Spectrum Licenses that are not being used to provide retail mobile wireless services, unless Acquiring Defendant already is providing nationwide retail mobile wireless services over DISH's facilities-based network.
- 4. If the Acquiring Defendant does not purchase the 800 MHz Spectrum Licenses, Divesting Defendants shall conduct an auction of the 800 MHz Spectrum Licenses within six (6) months of Acquiring Defendant declining to purchase the

licenses. In such auction, Divesting Defendants will not divest the 800 MHz Spectrum Licenses to any other national facilities-based mobile wireless network operator, without the prior written approval of the United States, in its sole discretion, after consultation with the affected Plaintiff States, and will not be required to divest the 800 MHz Spectrum Licenses at a price that is lower than the price the Acquiring Defendant originally agreed to pay for such licenses. In addition, Divesting Defendants may apply to the United States to be relieved from the commitment to sell the 800 MHz Spectrum Licenses if (i) Acquiring Defendant declines to purchase the 800 MHz Spectrum License and (ii) the sale of the 800 MHz Spectrum Licenses is no longer needed fully to remedy the competitive harms of the merger, as determined by the United States in its sole discretion, after consultation with the affected Plaintiff States.

C. **Decommissioned Cell Sites**

- 1. Divesting Defendants shall make all Cell Sites Decommissioned by Divesting Defendants within five (5) years of the closing of the divestiture of the Prepaid Assets, which shall not be fewer than 20,000 Cell Sites, available to Acquiring Defendant immediately after such Decommissioning.
- 2. Divesting Defendants shall provide, no later than the closing of the Prepaid Assets divestiture, the Acquiring Defendant and Monitoring Trustee with a detailed schedule identifying, over the next five (5) years: (i) each Cell Site that the Divesting Defendants plan to Decommission; (ii) the forecasted date for Decommissioning; and (iii) whether a given Cell Site is freely transferrable. For a period of five (5) years following the closing of the divestiture of the Prepaid Assets, on the first day of each

month Divesting Defendants shall submit to the Acquiring Defendant and Monitoring

Trustee updated Cell Site Decommissioning schedules that include a rolling monthly

forecast projected out two hundred and seventy (270) days. All forecasted

Decommissionings within one hundred and eighty (180) days will be binding, subject to
any mandatory restrictions on transfer imposed by federal or state law, unless the

Monitoring Trustee determines that the Decommissioning was changed for good cause,
and the changes and justifications are reported by the Divesting Defendants to the United

States.

3. Divesting Defendants are ordered to pay to the United States, within ninety (90) days following the end of each fiscal quarter, \$50,000 multiplied by the total number of Cell Sites in excess of two (2) percent of Cell Sites in any 180-day Cell Site forecast:

(a) for which the Acquiring Defendant exercised its option to acquire such Cell Site that was Decommissioned more than ten (10) days after the date forecasted in the 180-day Cell Site forecast or (b) that were Decommissioned but did not appear on any 180-day Cell Site forecast. If Divesting Defendants are incorrect, and have not cured within ten (10) days, on more than ten (10) percent of Cell Sites in any three 180-day Cell Site forecasts, the penalty shall increase to \$100,000 per incorrect Cell Site for which the Acquiring Defendant exercised its option to acquire such Cell Site starting on the fourth 180-day Cell Site forecast that is incorrect on at least ten (10) percent of Cell Sites and continuing at that level for any penalties imposed pursuant to this Paragraph. If Divesting Defendants demonstrate that there was good cause for the forecast to have been inaccurate with regard to an individual Cell Site, the United States may, in its sole

discretion, after consultation with the affected Plaintiff States, waive some or all of the payments.

- 4. Divesting Defendants shall assign or transfer any rights that are assignable or transferrable and are useful for Acquiring Defendant to deploy infrastructure on the Decommissioned Cell Sites and will waive or terminate any rights Divesting Defendants may have to impede or prevent Acquiring Defendant from doing so. Where Divesting Defendants do not have the right to assign or transfer such rights, Divesting Defendants will cooperate with Acquiring Defendant in its attempt to obtain the rights.
- 5. Divesting Defendants shall Decommission unnecessary Cell Sites promptly. Divesting Defendants will vacate a Decommissioned Cell Site as soon as reasonably possible after the site is no longer in use on any of the Divesting Defendants' networks. As soon as reasonably possible after making Decommissioned Cell Sites available to the Acquiring Defendant, Divesting Defendants shall also make any Decommissioned transport-related equipment (including microwave backhaul gear and network switches) on such cell sites available for purchase by the Acquiring Defendant. If the Monitoring Trustee determines that Divesting Defendants have not complied with this Paragraph, the Monitoring Trustee may recommend and the United States may impose a fine of up to \$50,000 per Cell Site per week for which Acquiring Defendant exercised its option to acquire such Cell Site or transport-related equipment for any violation.
- 6. Subject to the terms and conditions of the applicable lease or easement for such Cell Site, Divesting Defendants shall provide Acquiring Defendant reasonable

access to inspect Decommissioned Cell Sites prior to the deadline for Acquiring Defendant to exercise its option on the Decommissioned Cell Sites.

D. **Decommissioned Retail Locations**

- 1. Divesting Defendants shall make all assignable or transferrable Retail Locations Decommissioned by Divesting Defendants within five (5) years of the closing of the divestiture of the Prepaid Assets, which will not be fewer than four hundred (400) Retail Locations, available to Acquiring Defendant immediately after such Decommissioning.
- 2. Divesting Defendants shall notify Acquiring Defendant of Retail Locations that Divesting Defendants plan to Decommission as soon as the locations are identified.
- 3. Divesting Defendants shall waive or terminate any rights they have to impede or prevent Acquiring Defendant from using the Retail Locations.
- 4. Subject to the terms and conditions of the applicable lease for such Retail Location, Divesting Defendants shall provide Acquiring Defendant reasonable access to inspect Decommissioned Retail Locations prior to the deadline for Acquiring Defendant to exercise its option on the Decommissioned Retail Locations.
- E. Unless the United States otherwise consents in writing or the Acquiring

 Defendant declines its option to purchase certain Decommissioned Cell Sites or

 Decommissioned Retail Locations, the divestitures pursuant to this Final Judgment will include the entire Divestiture Assets. The divestitures will be accomplished in such a way as to satisfy the United States, in its sole discretion, that the Divestiture Assets can and will be used by Acquiring Defendant as part of a viable, ongoing operation relating to the provision of retail

mobile wireless service. The divestitures will be accomplished so as to satisfy the United States, in its sole discretion, that none of the terms of any agreement between Acquiring Defendant and Divesting Defendants gives the Divesting Defendants the ability unreasonably to raise the Acquiring Defendant's costs, to lower the Acquiring Defendant's efficiency, or otherwise to interfere with the ability of the Acquiring Defendant to compete.

- F. Acquiring Defendant shall use the Divestiture Assets to offer retail mobile wireless services, including offering nationwide postpaid retail mobile wireless service within one (1) year of the closing of the sale of the Prepaid Assets.
- G. Divesting Defendants shall not take any action that will impede in any way the permitting, operation, or divestiture of the Divestiture Assets.
- H. Divesting Defendants shall warrant to Acquiring Defendant (1) that there are no material defects known to the Divesting Defendants in the environmental, zoning, or other permits pertaining to the operation of the Divestiture Assets, (2) that following the sale of the Divestiture Assets, Divesting Defendants will not undertake, directly or indirectly, any challenges to the environmental, zoning, or other permits relating to the operation of the Divestiture Assets in a manner adverse to the Acquiring Defendant, and (3) that the Divestiture Assets will be capable of full operation on the date of transfer. For purposes of this Paragraph, the Divestiture Assets shall not include any Decommissioned Cell Sites or Decommissioned Retail Locations as to which the Acquiring Defendant declined its option to acquire the assets.
- I. For a period of up to one (1) year following the divestiture closing, if the Acquiring Defendant determines that any assets not included in the Divestiture Assets were previously used by the divested business and are reasonably necessary for the continued

competitiveness of the Divestiture Assets, it shall notify the United States, the Plaintiff States, and the Divesting Defendants in writing that it requires such assets. Provided, however, that such assets shall not include any tangible or intangible wireless network or spectrum assets (except as provided herein), or any tangible or intangible IT assets or software licenses used by the remaining Sprint business. The United States, in its sole discretion, after consultation with the affected Plaintiff States, taking into account Acquiring Defendant's assets and business, shall determine whether any of the assets identified should be divested to Acquiring Defendant. If the United States determines that such assets should be divested, Divesting Defendants and Acquiring Defendant will negotiate an agreement within thirty (30) calendar days providing for the divestiture of such assets in a period to be determined by the United States in consultation with the affected Plaintiff States and Divesting Defendants and Acquiring Defendant.

V. <u>600 MHz SPECTRUM DEPLOYMENT</u>

A. Acquiring Defendant and Divesting Defendants agree to negotiate in good faith to reach an agreement for Divesting Defendants to lease some or all of Acquiring Defendant's 600 MHz Spectrum Licenses for deployment to retail consumers by Divesting Defendants.

Defendants shall report to the Monitoring Trustee within ninety (90) days after the filing of this Final Judgment regarding the status of these negotiations. If, at the end of one hundred and eighty (180) days, Defendants have not reached an agreement to lease some or all of Acquiring Defendant's 600 MHz Spectrum Licenses for deployment by Divesting Defendants and use by retail consumers, the Monitoring Trustee shall report to the United States, which may then resolve any dispute at the United States' sole discretion, provided such resolution shall be based on commercially reasonable and mutually beneficial terms for both parties, recognizing that the

lease(s) must be for a sufficient period of time for Divesting Defendants to make adequate commercial use of the 600 MHz Spectrum Licenses.

VI. FULL MOBILE VIRTUAL NETWORK OPERATOR

- A. Divesting Defendants and Acquiring Defendant shall enter into a Full MVNO Agreement for a term of no fewer than seven (7) years. The terms and conditions of the Acquiring Defendant's use of Divesting Defendants' wireless networks pursuant to any Full MVNO Agreement shall be commercially reasonable and must be acceptable to the United States, in its sole discretion, after consultation with the affected Plaintiff States.
- B. In carrying out its obligations under any Full MVNO Agreement, Divesting Defendants:
 - 1. shall not reject any of Acquiring Defendant's lawful traffic, unless authorized to do so by any Full MVNO Agreement and accepted by the United States, in its sole discretion, after consultation with the affected Plaintiff States;
 - 2. shall not unreasonably discriminate against Acquiring Defendant or Acquiring Defendant's subscribers, including by blocking, throttling, or otherwise deprioritizing the Acquiring Defendant's customers differently than Divesting Defendants' own similarly situated customers, unless authorized to do so by any Full MVNO Agreement;
 - 3. shall use reasonable best efforts to provide Acquiring Defendant all operational support required for Acquiring Defendant's customers (including, but not limited to, customers of the Prepaid Assets) to be able to use the Divesting Defendants' wireless networks;

- 4. shall not unreasonably refuse to allow any device used by Acquiring Defendant's customers to access the Divesting Defendants' wireless networks, or otherwise unreasonably refuse to approve or support any such devices, and shall approve such devices for use upon request as soon as reasonably practicable, and shall use commercially reasonable efforts to provide technical support or other assistance to the Acquiring Defendant as requested to facilitate approval of any devices for use on Divesting Defendants' wireless networks;
- 5. shall configure its wireless network as necessary to enable the provision of handover mobility for the Acquiring Defendant's customers in the boundary areas between the Acquiring Defendant's network, built out in contiguous coverage areas (e.g., city-wide coverage), and the Divesting Defendants' wireless networks; and
- 6. shall not otherwise unreasonably delay, impede, or frustrate Acquiring
 Defendant's ability to use any Full MVNO Agreement and the Divesting Defendants'
 networks to become a nationwide facilities-based retail mobile wireless services provider.

VII. MOBILE VIRTUAL NETWORK OPERATOR COMPETITION

A. Divesting Defendants shall abide by all terms of their existing MVNO agreements. Divesting Defendants shall agree to extend existing MVNO agreements on their existing terms (other than any "most favored nation" provisions) until the expiration of this Final Judgment unless the Divesting Defendants demonstrate to the Monitoring Trustee that doing so will result in a material adverse effect, other than as a result of competition, on the Divesting Defendants' ongoing business. For the avoidance of doubt, Divesting Defendants are not required to extend any MVNO agreements beyond the expiration of this Final Judgment or any

existing infrastructure-based MVNO agreement that includes a reciprocal facility sharing arrangement unless it includes a mutually beneficial reciprocal facility sharing arrangement for the duration of the MVNO agreement. Any disputes arising from the negotiation of an agreement pursuant to this Paragraph shall be resolved by the United States in its sole discretion.

- B. Divesting Defendants and Acquiring Defendant agree to support eSIM technology on smartphones, including working with handset equipment manufacturers to support eSIM-capable phones to the extent such phones are technically capable of operating on Divesting Defendants or Acquiring Defendant's wireless networks.
- C. Divesting Defendants and Acquiring Defendant shall not discriminate against devices for the reason that the device uses remote SIM provisioning and eSIM technology to connect to the Defendants' wireless networks. Examples of discrimination would include, but are not limited to, refusing to sell a device because it contains or uses an eSIM, and refusing to certify for network access a device because it uses an eSIM, but discrimination would not include the application of the Defendant's generally applicable device-locking policies to devices sold or leased by Defendant, provided that the locking policy is consistent with Paragraph VII(F), below.
- D. Divesting Defendants and Acquiring Defendant shall not discriminate against devices for the reason that the device allows multiple active profiles or for the reason that the device allows automatic switching between those profiles. Examples of discrimination would include, but are not limited to, refusing to sell a device because it has these functions, and refusing to certify for network access a device because it has these functions. For avoidance of doubt, nothing contained in this provision will prohibit Defendants from exercising discretion to

determine whether a device or technology will harm or impede the operation of their respective wireless networks.

- E. Divesting Defendants and Acquiring Defendant shall make their network plans available to consumers who use on-screen selection software or applications from devices capable of being remotely provisioned on the same terms as offered to other consumers in that geographic area. This provision will apply to any device that is the same make and model as any device Defendants sell or otherwise certify for network access.
- F. Divesting Defendants and Acquiring Defendant agree to abide by the following unlocking principles for all methods of locking (including any limitation on the use of an eSIM to switch between profiles) for any postpaid or prepaid mobile wireless device that they lock to their network: (i) Divesting Defendants and Acquiring Defendant will post on their respective websites their clear, concise, and readily accessible policies on postpaid and prepaid mobile device unlocking; (ii) Divesting Defendants and Acquiring Defendant will unlock mobile wireless devices for their customers and former customers in good standing and individual owners of eligible devices after the fulfillment of the applicable postpaid service contract, device financing plan, or payment of applicable early termination fee; (iii) Divesting Defendants and Acquiring Defendant will unlock prepaid mobile wireless devices no later than one (1) year after initial activation, consistent with reasonable time, payment, or usage requirements; and (iv) Divesting Defendants and Acquiring Defendant will automatically unlock devices remotely within two (2) business days of devices becoming eligible for unlocking, and without additional fee, provided, however, that if not technically possible to automatically unlock devices remotely,

Divesting Defendants and Acquiring Defendant shall instead provide immediate notice to consumers that the devices are eligible to be unlocked.

VIII. FACILITIES-BASED EXPANSION AND ENTRY

- A. Divesting Defendants shall comply with all network build commitments made to the FCC related to the merger of T-Mobile and Sprint or the divestiture to Acquiring Defendant as of the date of entry of this Final Judgment, subject to verification by the FCC. Acquiring Defendant shall comply with the June 14, 2023 AWS-4, 700 MHz, H Block, and Nationwide 5G Broadband network build commitments made to the FCC as of the date of entry of this Final Judgment, subject to verification by the FCC. Defendants shall provide to the United States and the Plaintiff States copies of any reports or submissions to the FCC that are associated with any FCC order(s) within three (3) business days of submission to the FCC.
- B. Divesting Defendants shall not interfere with Acquiring Defendant's efforts to deploy a nationwide facilities-based mobile wireless network, or to operate that network.

 Acquiring Defendant shall use its best efforts to serve subscribers over its facilities-based wireless network rather than over Divesting Defendants' wireless networks.
- C. On the first day of the first fiscal quarter following the entry of this Final Judgment and every one hundred and eighty (180) days thereafter, Acquiring Defendant shall submit to the United States and the Plaintiff States an update on the status of its wireless network deployment. This update will include a description of Acquiring Defendant's deployment efforts since Acquiring Defendant's last report, including (a) the number of towers and small cells deployed by Acquiring Defendant; (b) the spectrum bands over which Acquiring Defendant has deployed equipment; (c) Acquiring Defendant's progress in obtaining subscriber devices that

operate on each of its licensed spectrum bands; (d) the percentage of the population of the United States covered by Acquiring Defendant's wireless network; (e) the number of mobile wireless subscribers served by Acquiring Defendant; (f) the amount of traffic transmitted to and from these subscribers over Acquiring Defendant's facilities-based wireless network; (g) the amount of traffic transmitted to and from these subscribers over Divesting Defendants' network pursuant to a Full MVNO Agreement; and (h) any efforts by Divesting Defendants to interfere with Acquiring Defendant's efforts to deploy and operate its facilities-based wireless network.

IX. <u>FINANCING</u>

Divesting Defendants and Parent Defendants shall not finance any part of any purchase made pursuant to this Final Judgment, unless the United States approves such financing in its sole discretion.

X. STIPULATION AND ORDER

Until the divestitures required by this Final Judgment have been accomplished, Divesting Defendants shall take all steps necessary to comply with the Stipulation and Order entered by the Court. Defendants shall take no action that would jeopardize the divestiture ordered by the Court.

XI. <u>AFFIDAVITS</u>

A. Within twenty (20) calendar days of the filing of the Complaint in this matter,
Divesting Defendants shall deliver to the United States and the Plaintiff States an affidavit that
describes in reasonable detail all actions Divesting Defendants have taken and all steps Divesting
Defendants have implemented on an ongoing basis to comply with Section X of this Final
Judgment. Divesting Defendants shall deliver to the United States and the Plaintiff States an

affidavit describing any changes to the efforts and actions outlined in Divesting Defendants' earlier affidavits filed pursuant to this Section within fifteen (15) calendar days after the change is implemented.

B. Divesting Defendants shall keep all records of all efforts made to preserve and divest the Divestiture Assets until one (1) year after such divestiture has been completed.

XII. APPOINTMENT OF MONITORING TRUSTEE

- A. Upon application of the United States, after consultation with the Plaintiff States, the Court shall appoint a Monitoring Trustee selected by the United States and approved by the Court.
- B. The Monitoring Trustee shall have the power and authority to monitor

 Defendants' compliance with the terms of this Final Judgment and the Stipulation and Order
 entered by the Court, and shall have such other powers as the Court deems appropriate. The

 Monitoring Trustee shall be required to investigate and report on the Defendants' compliance
 with this Final Judgment and the Stipulation and Order, and the Defendants' progress toward
 effectuating the purposes of this Final Judgment, including but not limited to: Divesting

 Defendants' sale of the Divestiture Assets, Divesting Defendants' compliance with its
 requirements to make Cell Sites and Retail Locations available to Acquiring Defendant, and

 Acquiring Defendant's progress toward using the Divestiture Assets and other company assets to
 operate a retail mobile wireless network.
- C. Subject to Paragraph XII(E) of this Final Judgment, the Monitoring Trustee may hire at the cost and expense of Divesting Defendants any agents, investment bankers, attorneys, accountants, or consultants, who will be solely accountable to the Monitoring Trustee,

reasonably necessary in the Monitoring Trustee's judgment. Any such agents or consultants shall serve on such terms and conditions as the United States approves, including confidentiality requirements and conflict of interest certifications.

- D. Defendants shall not object to actions taken by the Monitoring Trustee in fulfillment of the Monitoring Trustee's responsibilities under any Order of the Court on any ground other than the Monitoring Trustee's malfeasance. Any such objections by Defendants must be conveyed in writing to the United States and the Monitoring Trustee within ten (10) calendar days after the action taken by the Monitoring Trustee giving rise to Defendants' objection.
- E. The Monitoring Trustee shall serve at the cost and expense of Divesting Defendants pursuant to a written agreement with Divesting Defendants and on such terms and conditions as the United States approves, including confidentiality requirements and conflict of interest certifications. The compensation of the Monitoring Trustee and any agents or consultants retained by the Monitoring Trustee shall be on reasonable and customary terms commensurate with the individuals' experience and responsibilities. If the Monitoring Trustee and Divesting Defendants are unable to reach agreement on the Monitoring Trustee's or any agents' or consultants' compensation or other terms and conditions of engagement within fourteen (14) calendar days of the appointment of the Monitoring Trustee, the United States may, in its sole discretion, take appropriate action, including making a recommendation to the Court. The Monitoring Trustee shall, within three (3) business days of hiring any agents or consultants, provide written notice of such hiring and the rate of compensation to Divesting Defendants and the United States.

- F. The Monitoring Trustee shall have no responsibility or obligation for the operation of Defendants' businesses.
- G. Defendants shall use their best efforts to assist the Monitoring Trustee in monitoring Defendants' compliance with their individual obligations under this Final Judgment and under the Stipulation and Order. The Monitoring Trustee and any agents or consultants retained by the Monitoring Trustee shall have full and complete access to the personnel, books, records, and facilities relating to compliance with this Final Judgment, subject to reasonable protection for trade secrets; other confidential research, development, or commercial information; or any applicable privileges. Defendants shall take no action to interfere with or to impede the Monitoring Trustee's accomplishment of its responsibilities.
- H. After its appointment, the Monitoring Trustee shall file reports monthly, or more frequently as needed, with the United States setting forth Defendants' efforts to comply with Defendants' obligations under this Final Judgment and under the Stipulation and Order. To the extent such reports contain information that the Monitoring Trustee deems confidential, such reports will not be filed in the public docket of the Court.
- I. The Monitoring Trustee shall serve until the divestiture of all the Divestiture Assets is finalized pursuant to this Final Judgment, until the buildout requirements are complete pursuant to Section VIII of this Final Judgment, until any Full MVNO Agreement expires or otherwise terminates, or until the term of any transition services agreement pursuant to Paragraph IV(A)(4) of this Final Judgment has expired, whichever is later.

J. If the United States determines that the Monitoring Trustee has ceased to act or failed to act diligently or in a reasonably cost-effective manner, it may recommend that the Court appoint a substitute Monitoring Trustee.

XIII. <u>FIREWALL</u>

- A. During the term of this Final Judgment, the Divesting Defendants and Acquiring Defendant shall implement and maintain reasonable procedures to prevent competitively sensitive information from being disclosed by or through implementation and execution of the obligations in this agreement or any associated agreements to components or individuals within the respective companies involved in the marketing, distribution, or sale of competing products.
- B. Divesting Defendants and Acquiring Defendant each shall, within thirty (30) business days of the entry of the Stipulation and Order, submit to the United States, the Plaintiff States, and the Monitoring Trustee a document setting forth in detail the procedures implemented to effect compliance with this Section. Upon receipt of the document, the United States shall inform Divesting Defendants and Acquiring Defendant within thirty (30) business days whether, in its sole discretion, it approves of or rejects each party's compliance plan. In the event that Divesting Defendants' or Acquiring Defendant's compliance plan is rejected, the United States shall provide Divesting Defendants or Acquiring Defendant, as applicable, the reasons for the rejection. Divesting Defendants or Acquiring Defendant, as applicable, shall be given the opportunity to submit, within ten (10) business days of receiving a notice of rejection, a revised compliance plan. If Divesting Defendants or Acquiring Defendant cannot agree with the United States on a compliance plan, the United States shall have the right to request that this Court rule

on whether Divesting Defendants' or Acquiring Defendant's proposed compliance plan fulfills the requirements of this Section.

- C. Divesting Defendants and Acquiring Defendant shall:
- 1. furnish a copy of this Final Judgment and related Competitive Impact

 Statement within sixty (60) calendar days of entry of the Stipulation and Order to (a) each
 officer, director, and any other employee that will receive competitively sensitive
 information; and (b) each officer, director, and any other employee that is involved in (i)
 any contacts with the other companies that are parties to any transition services
 agreement contemplated by this Final Judgment, or (ii) making decisions under any
 transition services agreement entered into pursuant to this Final Judgment;
- 2. furnish a copy of this Final Judgment and related Competitive Impact Statement to any successor to a person designated in Paragraph XIII(C)(1) upon assuming that position;
- 3. annually brief each person designated in Paragraph XIII(C)(1) and Paragraph XIII(C)(2) on the meaning and requirements of this Final Judgment and the antitrust laws; and
- 4. obtain from each person designated in Paragraph XI(C)(1) and Paragraph XI(C)(2), within thirty (30) calendar days of that person's receipt of the Final Judgment, a certification that he or she (a) has read and, to the best of his or her ability, understands and agrees to abide by the terms of this Final Judgment; (b) is not aware of any violation of the Final Judgment that has not been reported to the company; and (c) understands that any person's failure to comply with this Final Judgment may result in an enforcement

action for contempt of court against each Defendant or any person who violates this Final Judgment.

XIV. <u>COMPLIANCE INSPECTION</u>

- A. For the purposes of determining or securing compliance with this Final Judgment, or of any related orders such as any Stipulation and Order, or of determining whether the Final Judgment should be modified or vacated, and subject to any legally-recognized privilege, from time to time authorized representatives of the United States, including agents and consultants retained by the United States, shall, upon written request of an authorized representative of the Assistant Attorney General in charge of the Antitrust Division, and on reasonable notice to Defendants, be permitted:
 - 1. access during Defendants' office hours to inspect and copy, or at the option of the United States, to require Defendants to provide electronic copies of all books, ledgers, accounts, records, data, and documents in the possession, custody, or control of Defendants, relating to any matters contained in this Final Judgment; and
 - 2. to interview, either informally or on the record, Defendants' officers, employees, or agents, who may have their individual counsel present, regarding such matters. The interviews will be subject to the reasonable convenience of the interviewee and without restraint or interference by Defendants.
- B. Upon the written request of an authorized representative of the Assistant Attorney General in charge of the Antitrust Division, Defendants shall submit written reports or response to written interrogatories, under oath if requested, relating to any of the matters contained in this Final Judgment as may be requested.

- C. No information or documents obtained by the means provided in this Section will be divulged by the United States to any person other than an authorized representative of the executive branch of the United States, except in the course of legal proceedings to which the United States is a party (including grand jury proceedings), for the purpose of securing compliance with this Final Judgment, or as otherwise required by law.
- D. If at the time that Defendants furnish information or documents to the United States, Defendants represent and identify in writing the material in any such information or documents to which a claim of protection may be asserted under Rule 26(c)(1)(G) of the Federal Rules of Civil Procedure, and Defendants mark each pertinent page of such material, "Subject to claim of protection under Rule 26(c)(1)(G) of the Federal Rules of Civil Procedure," then the United States shall give Defendants ten (10) calendar days' notice prior to divulging such material in any legal proceeding (other than a grand jury proceeding).

XV. NO REACQUISITION OR SALE TO COMPETITOR

- A. Divesting Defendants and Parent Defendants shall not reacquire any part of the Divestiture Assets during the term of this Final Judgment.
- B. Divesting Defendants and Parent Defendants shall not acquire any other assets that are substantially similar to the Divestiture Assets from the Acquiring Defendant during the terms of this Final Judgment.
- C. Acquiring Defendant shall not sell, lease, or otherwise provide the right to use the Divestiture Assets (including, but not limited to, selling wholesale wireless network capacity) to any national facilities-based mobile wireless provider during the term of this Final Judgment, except for a roaming arrangement, without prior approval of the United States; provided,

however, that following the divestiture of the 800 MHz Spectrum Licenses, the Divesting Defendants will be permitted to lease back from the Acquiring Defendant up to 4 MHz of spectrum as needed for up to two (2) years following the divestiture of the 800 MHz Spectrum Licenses.

XVI. <u>NOTIFICATIONS</u>

- A. Acquiring Defendant shall notify the United States at least thirty (30) calendar days prior to any change in the corporation(s) that may affect compliance obligations arising under this Final Judgment, including, but not limited to: a dissolution, assignment, sale, merger, or other action that would result in the emergence of a successor corporation; the creation or dissolution of a subsidiary, parent, or affiliate that engages in any acts or practices subject to this Final Judgment; the proposed filing of a bankruptcy petition; or a change in the corporate name or address. Provided, however, that, with respect to any proposed change in the corporation(s) about which Acquiring Defendant learns fewer than thirty (30) calendar days prior to the date such action is to take place, Acquiring Defendant shall notify the United States as soon as is practicable after obtaining such knowledge.
- B. For transactions that are not subject to the reporting and waiting period requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, 15 U.S.C. § 18a (the "HSR Act"), Divesting Defendants shall not, without providing advanced notification to the United States, directly or indirectly acquire a financial interest, including through securities, loan, equity, or management interest, in any company that competes for the provision of mobile wireless retail services. Acquiring Defendant shall not sell any of the

Divestiture Assets or any currently held substantially similar assets, directly or indirectly, without providing advance notification to the United States.

- C. Such notification will be provided to the United States in the same format as, and per the instructions relating to, the Notification and Report Form set forth in the Appendix to Part 803 of Title 16 of the Code of Federal Regulations as amended. Notification will be provided at least thirty (30) calendar days prior to acquiring any such interest, and will include, beyond what may be required by the applicable instructions, the names of the principal representatives of the parties to the agreement who negotiated the agreement, and any management or strategic plans discussing the proposed transaction. If within thirty (30) calendar days after notification, the United States makes a written request for additional information, Defendants shall not consummate the proposed transaction or agreement until thirty (30) calendar days after submitting and certifying, in the manner described in Part 803 of Title 16 of the Code of Federal Regulations as amended, the truth, correctness, and completeness of all such additional information. Early termination of the waiting periods in this paragraph may be requested and, where appropriate, granted in the same manner as is applicable under the requirements and provisions of the HSR Act and rules promulgated thereunder. This Section will be broadly construed and any ambiguity or uncertainty regarding the filing of notice under this Section will be resolved in favor of filing notice. Defendants may, however, provide informal notice and request that the United States waive the requirement of formal notice for any transaction.
- D. Defendants represent and warrant to the United States that they have disclosed all agreements between Acquiring Defendant and either Divesting Defendants or Parent Defendants

related to the settlement of this action and their obligations and commitments put forth in this

Final Judgment. Defendants will provide thirty (30) days written notice to the United States of
any intent to enter into or execute any amendment, supplement, or modification to any of the
agreements between Divesting Defendants or Parent Defendants and Acquiring Defendant.

Notwithstanding any provision to the contrary in the agreements between Divesting Defendants
or Parent Defendants and Acquiring Defendant, Divesting Defendants or Parent Defendants may
not amend, supplement, terminate, or modify any of the agreements or any portion thereof
without obtaining the consent of the United States in its sole discretion. The United States will
not withhold consent to amendment, supplementation, modification, or termination of any of the
agreements or portion thereof if Divesting Defendants demonstrate to the United States, in its
sole discretion, that a refusal to amend, supplement, modify, or terminate the agreements would
prevent Divesting Defendants from meeting any build out requirements imposed by the FCC.

XVII. RETENTION OF JURISDICTION

The Court retains jurisdiction to enable any party to this Final Judgment to apply to the Court at any time for further orders and directions as may be necessary or appropriate to carry out or construe this Final Judgment, to modify any of its provisions, to enforce compliance, and to punish violations of its provisions.

XVIII. <u>ENFORCEMENT OF FINAL JUDGMENT</u>

A. The United States retains and reserves all rights to enforce the provisions of this Final Judgment, including the right to seek an order of contempt from the Court. Defendants agree that in any civil contempt action, any motion to show cause, or any similar action brought by the United States regarding an alleged violation of this Final Judgment, the United States may

establish a violation of the decree and the appropriateness of any remedy therefore by a preponderance of the evidence, and Defendants waive any argument that a different standard of proof should apply.

- B. The Final Judgment should be interpreted to give full effect to the procompetitive purposes of the antitrust laws and to restore all competition harmed by the challenged conduct. Defendants agree that they may be held in contempt of, and that the Court may enforce, any provision of this Final Judgment that, as interpreted by the Court in light of these procompetitive principles and applying ordinary tools of interpretation, is stated specifically and in reasonable detail, whether or not it is clear and unambiguous on its face. In any such interpretation, the terms of this Final Judgment should not be construed against either party as the drafter.
- C. In any enforcement proceeding in which the Court finds that Defendants have violated this Final Judgment, the United States may apply to the Court for a one-time extension of this Final Judgment, together with such other relief as may be appropriate. In connection with any successful effort by the United States to enforce this Final Judgment against a Defendant, whether litigated or resolved prior to litigation, that Defendant agrees to reimburse the United States for the fees and expenses of its attorneys, as well as any other costs including experts' fees, incurred in connection with that enforcement effort, including in the investigation of the potential violation.
- D. For a period of four (4) years after the expiration or termination of the Final Judgment pursuant to Section XIX, if the United States has evidence that a Defendant violated this Final Judgment before it expired or was terminated, the United States may file an action against that Defendant in this Court requiring that the Court order (i) Defendant to comply with

the terms of this Final Judgment for an additional term of at least four (4) years following the filing of the enforcement action under this Section, (ii) any appropriate contempt remedies, (iii) any additional relief needed to ensure that Defendant complies with the terms of the Final Judgment, and (iv) fees or expenses as called for in Paragraph XVIII(C).

XIX. EXPIRATION OF FINAL JUDGMENT

Unless the Court grants an extension, this Final Judgment expires seven (7) years from the date of its entry, except that after five (5) years from the date of its entry, this Final Judgment may be terminated upon notice by the United States to the Court and Defendants that the divestitures, buildouts and other requirements have been completed and that the continuation of the Final Judgment no longer is necessary or in the public interest.

XX. PUBLIC INTEREST DETERMINATION

Entry of this Final Judgment is in the public interest. The parties have complied with the requirements of the Antitrust Procedures and Penalties Act, 15 U.S.C. § 16, including making copies available to the public of this Final Judgment, the Competitive Impact Statement, any comments thereon, and the United States' responses to comments. Based upon the record before the Court, which includes the Competitive Impact Statement and any comments and responses to comments filed with the Court, entry of this Final Judgment is in the public interest.

comments filed with the Court, entry of this Final Judgment is in the public interest.
Date:
[Court approval subject to procedures of Antitrust Procedures and Penalties Act, 15 U.S.C. § 16]

United States District Judge

DISH Network/ParkerB.Com L.L.C. 600 MHz Licenses (sorted by call sign)

Call Sign	PEA Number	<u>Municipality</u>	<u>State</u>
WQZM232	PEA 026	Las Vegas	NV
WQZM233	PEA 026	Las Vegas	NV
WQZM234	PEA 362	Payette	ID
WQZM235	PEA 148	Bellingham	WA
WQZM236	PEA 195	Lewiston	ID
WQZM237	PEA 237	Hinesville	GA
WQZM238	PEA 215	Hickory	NC
WQZM239	PEA 410	Valentine	NE
WQZM240	PEA 254	Merrill	WI
WQZM241	PEA 185	Marquette	MI
WQZM242	PEA 137	Eau Claire	WI
WQZM243	PEA 074	Chattanooga	TN
WQZM244	PEA 009	Miami	FL
WQZM245	PEA 009	Miami	FL
WQZM246	PEA 009	Miami	FL
WQZM247	PEA 335	Natchitoches	LA
WQZM248	PEA 404	Kanab	UT
WQZM249	PEA 285	Gallup	NM
WQZM250	PEA 392	Maryville	МО
WQZM251	PEA 412	Puerto Rico	PR
WQZM252	PEA 412	Puerto Rico	PR
WQZM253	PEA 038	Milwaukee	WI
WQZM254	PEA 386	Barnwell	SC
WQZM255	PEA 029	Jacksonville	FL
WQZM256	PEA 382	Riverton	WY
WQZM257	PEA 343	Pecos	TX
WQZM258	PEA 261	Fargo	ND
WQZM259	PEA 226	Lima	ОН
WQZM260	PEA 336	Grand Forks	ND
WQZM261	PEA 390	Snyder	TX
WQZM262	PEA 408	Ballinger	TX
WQZM263	PEA 363	Big Spring	TX
WQZM264	PEA 402	Brady	TX
WQZM265	PEA 288	Abilene	TX
WQZM266	PEA 320	San Angelo	TX
WQZM267	PEA 247	Nampa	ID
WQZM268	PEA 156	Boise City	ID
WQZM269	PEA 046	Little Rock	AR
WQZM270	PEA 046	Little Rock	AR

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM271	PEA 297	Pendleton	OR
WQZM272	PEA 206	Wenatchee	WA
WQZM273	PEA 119	Yakima	WA
WQZM274	PEA 107	Bangor	ME
WQZM275	PEA 127	Evansville	IN
WQZM276	PEA 323	Socorro	NM
WQZM277	PEA 263	Santa Fe	NM
WQZM278	PEA 345	Newberry	SC
WQZM279	PEA 327	Orangeburg	SC
WQZM280	PEA 284	Greenwood	SC
WQZM281	PEA 332	Bennettsville	SC
WQZM282	PEA 188	Jamestown	NY
WQZM283	PEA 138	Burlington	VT
WQZM284	PEA 319	Albany	GA
WQZM285	PEA 371	Wytheville	VA
WQZM286	PEA 230	Lumberton	NC
WQZM287	PEA 291	Rockingham	NC
WQZM288	PEA 309	Elizabeth City	NC
WQZM289	PEA 228	Roanoke	VA
WQZM290	PEA 131	Sanford	NC
WQZM291	PEA 169	Goldsboro	NC
WQZM292	PEA 146	Wilmington	NC
WQZM293	PEA 305	Altus	OK
WQZM294	PEA 302	Enid	OK
WQZM295	PEA 251	Salina	KS
WQZM296	PEA 277	Hutchinson	KS
WQZM297	PEA 070	Eugene	OR
WQZM298	PEA 403	Lewistown	MT
WQZM299	PEA 334	Pampa	TX
WQZM300	PEA 411	Van Horn	TX
WQZM301	PEA 048	Harrisburg	PA
WQZM302	PEA 084	Mobile	AL
WQZM303	PEA 027	Salt Lake City	UT
WQZM304	PEA 027	Salt Lake City	UT
WQZM305	PEA 011	Atlanta	GA
WQZM306	PEA 011	Atlanta	GA TV
WQZM307	PEA 132	Corpus Christi	TX
WQZM308	PEA 409	Haskell	TX
WQZM309	PEA 400	Muleshoe	TX
WQZM310	PEA 401	Floydada	TX

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM311	PEA 376	Hereford	TX
WQZM312	PEA 355	Casper	WY
WQZM313	PEA 346	Franklin	NC
WQZM314	PEA 233	Shelby	NC
WQZM315	PEA 208	Salisbury	NC
WQZM316	PEA 207	Brunswick	GA
WQZM317	PEA 008	Dallas	TX
WQZM318	PEA 008	Dallas	TX
WQZM319	PEA 004	San Francisco	CA
WQZM320	PEA 004	San Francisco	CA
WQZM321	PEA 004	San Francisco	CA
WQZM322	PEA 010	Houston	TX
WQZM323	PEA 416	Gulf of Mexico	N/A
WQZM324	PEA 416	Gulf of Mexico	N/A
WQZM325	PEA 396	Winterset	IA
WQZM326	PEA 370	Washington	IA
WQZM327	PEA 265	Winona	MN
WQZM328	PEA 354	New London	WI
WQZM329	PEA 253	Baraboo	WI
WQZM330	PEA 269	Racine	WI
WQZM331	PEA 268	Clinton	IA
WQZM332	PEA 218	Wausau	WI
WQZM333	PEA 294	Waterloo	IA
WQZM334	PEA 267	Sheboygan	WI
WQZM335	PEA 252	Sioux City	IA
WQZM336	PEA 209	Green Bay	WI
WQZM337	PEA 176	Ames	IA
WQZM338	PEA 163	Davenport	IA
WQZM339	PEA 225	La Crosse	WI
WQZM340	PEA 223	Dubuque	IA
WQZM341	PEA 179	Burlington	IA
WQZM342	PEA 155	Appleton	WI
WQZM343	PEA 182	Cedar Rapids	IA
WQZM344	PEA 122	Madison	WI
WQZM345	PEA 219	Mason City	IA
WQZM346	PEA 153	Fond du Lac	WI
WQZM347	PEA 159	Valdosta	GA
WQZM348	PEA 197	Wheeling	WV
WQZM349	PEA 121	Altoona	PA
WQZM350	PEA 194	State College	PA

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM351	PEA 387	Wahpeton	ND
WQZM352	PEA 270	Ottawa	IL
WQZM353	PEA 118	Richmond	IN
WQZM354	PEA 143	Keene	NH
WQZM355	PEA 407	Salmon	ID
WQZM356	PEA 324	Honesdale	PA
WQZM357	PEA 136	Williamsport	PA
WQZM358	PEA 241	Dublin	GA
WQZM359	PEA 298	Fairbanks	AK
WQZM360	PEA 298	Fairbanks	AK
WQZM361	PEA 298	Fairbanks	AK
WQZM362	PEA 264	Kodiak	AK
WQZM363	PEA 264	Kodiak	AK
WQZM364	PEA 264	Kodiak	AK
WQZM365	PEA 406	Anamosa	IA
WQZM366	PEA 322	Minot	ND
WQZM367	PEA 318	Thief River Falls	MN
WQZM368	PEA 274	Twin Falls	ID
WQZM369	PEA 187	Pocatello	ID
WQZM370	PEA 279	Logan	UT
WQZM371	PEA 158	Helena	MT
WQZM372	PEA 068	Grand Rapids	MI
WQZM373	PEA 056	Kalamazoo	MI
WQZM374	PEA 061	Toledo	ОН
WQZM375	PEA 315	Sheridan	WY
WQZM376	PEA 348	Aberdeen	SD
WQZM377	PEA 129	Springfield	IL
WQZM378	PEA 256	Lynchburg	VA
WQZM379	PEA 256	Lynchburg	VA
WQZM380	PEA 378	Waynesboro	GA
WQZM381	PEA 378	Waynesboro	GA
WQZM382	PEA 147	Salisbury	MD
WQZM383	PEA 147	Salisbury	MD
WQZM384	PEA 260	Alpena	MI
WQZM385	PEA 203	Traverse City	MI
WQZM386	PEA 338	Durango	CO
WQZM387	PEA 178	Sedalia	MO
WQZM388	PEA 178	Sedalia	MO
WQZM389	PEA 101	Wichita	KS
WQZM390	PEA 073	El Paso	TX

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM391	PEA 031	Indianapolis	IN
WQZM392	PEA 031	Indianapolis	IN
WQZM393	PEA 022	Sacramento	CA
WQZM394	PEA 022	Sacramento	CA
WQZM395	PEA 398	South Sioux City	NE
WQZM396	PEA 075	Albuquerque	NM
WQZM397	PEA 414	US Virgin Islands	USVI
WQZM398	PEA 001	New York	NY
WQZM399	PEA 001	New York	NY
WQZM400	PEA 001	New York	NY
WQZM401	PEA 001	New York	NY
WQZM402	PEA 357	Espanola	NM
WQZM403	PEA 180	Flagstaff	AZ
WQZM404	PEA 308	Americus	GA
WQZM405	PEA 262	Hilton Head Island	SC
WQZM406	PEA 128	Macon	GA
WQZM407	PEA 151	Winston-Salem	NC
WQZM408	PEA 384	Manchester	IA
WQZM409	PEA 042	Honolulu	HI
WQZM410	PEA 042	Honolulu	HI
WQZM411	PEA 032	Nashville	TN
WQZM412	PEA 032	Nashville	TN
WQZM413	PEA 340	Clovis	NM
WQZM414	PEA 259	Roswell	NM
WQZM415	PEA 211	Ardmore	OK
WQZM416	PEA 266	Lenoir	NC
WQZM417	PEA 018	San Diego	CA
WQZM418	PEA 018	San Diego	CA
WQZM419	PEA 286	Sioux Falls	SD
WQZM420	PEA 289	Price	UT
WQZM421	PEA 047	Brownsville	TX
WQZM422	PEA 053	Tucson	AZ
WQZM423	PEA 017	Minneapolis-St. Paul	MN
WQZM424	PEA 017	Minneapolis-St. Paul	MN
WQZM425	PEA 016	Seattle	WA
WQZM426	PEA 016	Seattle	WA
WQZM427	PEA 006	Philadelphia Philadelphia	PA
WQZM428 WQZM429	PEA 006	Baltimore-Washington	PA MD/DC
WQZM430	PEA 005 PEA 005	Baltimore-Washington	MD/DC MD/DC
WQZIVI430	PEA 003	baltimore-washington	MD/DC

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM431	PEA 280	Garden City	KS
WQZM432	PEA 339	Scottsbluff	NE
WQZM433	PEA 331	Plainview	TX
WQZM434	PEA 276	Rapid City	SD
WQZM435	PEA 383	Creston	IA
WQZM436	PEA 036	New Orleans	LA
WQZM437	PEA 040	Birmingham	AL
WQZM438	PEA 040	Birmingham	AL
WQZM439	PEA 248	Sumter	SC
WQZM440	PEA 134	Newark	ОН
WQZM441	PEA 141	Brainerd	MN
WQZM442	PEA 051	Louisville	KY
WQZM443	PEA 051	Louisville	KY
WQZM444	PEA 342	Mitchell	SD
WQZM445	PEA 044	Rochester	NY
WQZM446	PEA 044	Rochester	NY
WQZM447	PEA 060	Manchester	NH
WQZM448	PEA 060	Manchester	NH
WQZM449	PEA 069	Springfield	MA
WQZM450	PEA 391	Ontario	OR
WQZM451	PEA 379	Sault Ste. Marie	MI
WQZM452	PEA 380	Escanaba	MI
WQZM453	PEA 395	Jamestown	ND
WQZM454	PEA 072	Tallahassee	FL
WQZM455	PEA 067	Sarasota	FL
WQZM456	PEA 065	Cape Coral	FL
WQZM457	PEA 002	Los Angeles	CA
WQZM458	PEA 002	Los Angeles	CA
WQZM459		Cullman	AL
WQZM460	PEA 170	Dothan	AL
WQZM461	PEA 139	Hot Springs	AR
WQZM462	PEA 034	Fresno	CA
WQZM463	PEA 034	Fresno	CA
WQZM464	PEA 023	Pittsburgh	PA
WQZM465	PEA 023	Pittsburgh	PA
WQZM466	PEA 019	Portland	OR
WQZM467	PEA 019	Portland	OR
WQZM468	PEA 013	Orlando	FL
WQZM469	PEA 013	Orlando	FL
WQZM470	PEA 204	Owensboro	KY

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM471	PEA 234	Lexington	NC
WQZM472	PEA 003	Chicago	IL
WQZM473	PEA 292	Pueblo	СО
WQZM474	PEA 304	Mount Airy	NC
WQZM475	PEA 076	Reno	NV
WQZM476	PEA 105	Augusta	GA
WQZM477	PEA 117	La Grange	GA
WQZM478	PEA 071	Knoxville	TN
WQZM479	PEA 394	Martin	SD
WQZM480	PEA 351	Dickinson	ND
WQZM481	PEA 162	Elizabeth town	KY
WQZM482	PEA 081	Saginaw	MI
WQZM483	PEA 081	Saginaw	MI
WQZM484	PEA 326	Fergus Falls	MN
WQZM485	PEA 112	Bowling Green	KY
WQZM486	PEA 077	Portland	ME
WQZM487	PEA 287	Kenosha	WI
WQZM488	PEA 224	De Kalb	IL
WQZM489	PEA 186	Rock Hill	SC
WQZM490	PEA 193	Saint Joseph	МО
WQZM491	PEA 193	Saint Joseph	МО
WQZM492	PEA 250	Las Cruces	NM
WQZM493	PEA 066	Lansing	MI
WQZM494	PEA 066	Lansing	MI
WQZM495	PEA 333	Sidney	ОН
WQZM496	PEA 321	Batesville	IN
WQZM497	PEA 123	Mansfield	ОН
WQZM498	PEA 368	Concordia	KS
WQZM499		Pottsville	PA
WQZM500	PEA 290	Watertown	SD
WQZM501	PEA 290	Watertown	SD
WQZM502	PEA 037	Columbus	OH
WQZM503	PEA 037	Columbus	OH
WQZM504	PEA 361	Richfield	UT
WQZM505	PEA 303	Great Falls	MT
WQZM506	PEA 140	Fredericksburg	VA
WQZM507	PEA 140	Fredericksburg	VA
WQZM508	PEA 063	Tulsa	OK
WQZM509	PEA 413	Guam Grand Innation	Guam
WQZM510	PEA 102	Grand Junction	СО

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM511	PEA 316	Rock Springs	WY
WQZM512	PEA 366	Pullman	WA
WQZM513	PEA 366	Pullman	WA
WQZM514	PEA 353	Watseka	IL
WQZM515	PEA 113	Erie	PA
WQZM516	PEA 092	Decatur	IL
WQZM517	PEA 083	Fort Wayne	IN
WQZM518	PEA 064	South Bend	IN
WQZM519	PEA 058	Bloomington	IN
WQZM520	PEA 089	Columbia	SC
WQZM521	PEA 015	Phoenix	AZ
WQZM522	PEA 015	Phoenix	AZ
WQZM523	PEA 347	New Roads	LA
WQZM524	PEA 350	Forrest City	AR
WQZM525	PEA 293	Lawrenceburg	TN
WQZM526	PEA 310	Farmington	MO
WQZM527	PEA 196	Cape Girardeau	MO
WQZM528	PEA 145	Columbia	TN
WQZM529	PEA 174	Springfield	MO
WQZM530	PEA 161	Carbondale	IL
WQZM531	PEA 125	Alton	IL
WQZM532	PEA 273	Bloomington	IL
WQZM533	PEA 329	Kingsville	TX
WQZM534	PEA 385	Hannibal	МО
WQZM535	PEA 255	Greenville	MS
WQZM536	PEA 149	Biloxi	MS
WQZM537	PEA 175	Southaven	MS
WQZM538	PEA 030	Kansas City	MO
WQZM539		Kansas City	МО
WQZM540	PEA 020	Denver	СО
WQZM541	PEA 020	Denver	CO
WQZM542	PEA 012	Detroit	MI
WQZM543	PEA 012	Detroit	MI
WQZM544	PEA 393	Macon	MO
WQZM545	PEA 367	Moberly	MO
WQZM546	PEA 098	Johnson City	TN
WQZM547	PEA 055	Huntsville	AL
WQZM548 WQZM549	PEA 399	Lampasas	TX NM
WQZM550	PEA 375 PEA 352	Deming Gonzales	TX
WUZIVISSU	PEA 332	Gonzales	IX

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM551	PEA 358	Marble Falls	TX
WQZM552	PEA 337	Mineral Wells	TX
WQZM553	PEA 314	Jacksonville	TX
WQZM554	PEA 313	Lockhart	TX
WQZM555	PEA 275	Corsicana	TX
WQZM556	PEA 272	Brownwood	TX
WQZM557	PEA 221	Laredo	TX
WQZM558	PEA 201	Eagle Pass	TX
WQZM559	PEA 160	Victoria	TX
WQZM560	PEA 126	Casa Grande	AZ
WQZM561	PEA 133	Nacogdoches	TX
WQZM562	PEA 152	Tyler	TX
WQZM563	PEA 144	Paris	TX
WQZM564	PEA 096	Richmond	KY
WQZM565	PEA 021	Tampa	FL
WQZM566	PEA 021	Tampa	FL
WQZM567	PEA 110	Jackson	TN
WQZM568	PEA 243	Paducah	KY
WQZM569	PEA 078	Greensboro	NC
WQZM570	PEA 085	Charleston	SC
WQZM571	PEA 045	Raleigh	NC
WQZM572	PEA 093	Lafayette	LA
WQZM573	PEA 111	Fayetteville	AR
WQZM574	PEA 086	Frankfort	KY
WQZM575	PEA 082	Baton Rouge	LA
WQZM576	PEA 091	Colorado Springs	СО
WQZM577	PEA 090	Jackson	MS
WQZM578	PEA 397	Aliceville	AL
WQZM579		Aliceville	AL
WQZM580	PEA 108	Des Moines	IA
WQZM581	PEA 239	Kannapolis	NC
WQZM582	PEA 049	Albany	NY
WQZM583	PEA 041	Syracuse	NY
WQZM584	PEA 271	Elmira	NY
WQZM585	PEA 271	Elmira	NY
WQZM586	PEA 094	Waco	TX
WQZM587	PEA 330	Olney Florence	IL SC
WQZM588 WQZM589	PEA 238		SC SC
WQZM590	PEA 154 PEA 389	Myrtle Beach McCook	NE NE
WQZIVIS90	FEM 309	IVICCOOK	INE

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM591	PEA 171	Fort Smith	AR
WQZM592	PEA 062	Dayton	ОН
WQZM593	PEA 282	Galesburg	IL
WQZM594	PEA 168	Peoria	IL
WQZM595	PEA 198	Jonesboro	AR
WQZM596	PEA 216	Joplin	MO
WQZM597	PEA 232	Topeka	KS
WQZM598	PEA 164	Montgomery	AL
WQZM599	PEA 245	West Plains	MO
WQZM600	PEA 299	Kirksville	MO
WQZM601	PEA 183	Columbia	MO
WQZM602	PEA 150	Rolla	MO
WQZM603	PEA 222	Morristown	TN
WQZM604	PEA 244	Manhattan	KS
WQZM605	PEA 079	Hattiesburg	MS
WQZM606	PEA 099	Tupelo	MS
WQZM607	PEA 415	American Samoa	N/A
WQZM608	PEA 415	American Samoa	N/A
WQZM609	PEA 312	Farmington	NM
WQZM610	PEA 033	Virginia Beach	VA
WQZM611	PEA 033	Virginia Beach	VA
WQZM612	PEA 039	Oklahoma City	OK
WQZM613	PEA 039	Oklahoma City	OK
WQZM614	PEA 025	Cincinnati	OH
WQZM615	PEA 025	Cincinnati	OH
WQZM616	PEA 035	Austin	TX
WQZM617	PEA 035	Austin	TX
WQZM618	PEA 014	Cleveland	OH
WQZM619 WQZM620		Cleveland	OH
	PEA 142	Merced	CA
WQZM621 WQZM622	PEA 142	Merced Yuma	CA AZ
WQZM623	PEA 157 PEA 157	Yuma	AZ
WQZM624	PEA 137	Frederick	MD
WQZM625	PEA 088	Frederick	MD
WQZM626	PEA 028	San Antonio	TX
WQZM627	PEA 028	San Antonio	TX
WQZM628	PEA 181	Texarkana	TX
WQZM629	PEA 050	Greenville	SC
WQZM630	PEA 043	Charlotte	NC
TT QLIVIUSU	LACOTO	Charlotto	110

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM631	PEA 007	Boston	MA
WQZM632	PEA 007	Boston	MA
WQZM633	PEA 325	Bismarck	ND
WQZM634	PEA 388	Atlantic	IA
WQZM635	PEA 374	North Platte	NE
WQZM636	PEA 295	Stillwater	OK
WQZM637	PEA 306	Wichita Falls	TX
WQZM638	PEA 231	Fremont	NE
WQZM639	PEA 365	Vernon	TX
WQZM640	PEA 236	Grand Island	NE
WQZM641	PEA 281	Muskogee	OK
WQZM642	PEA 214	Lincoln	NE
WQZM643	PEA 278	Bartlesville	OK
WQZM644	PEA 114	Morgantown	WV
WQZM645	PEA 116	Rockford	IL
WQZM646	PEA 080	Omaha	NE
WQZM647	PEA 057	Richmond	VA
WQZM648	PEA 199	Dalton	GA
WQZM649	PEA 165	Rome	GA
WQZM650	PEA 054	Buffalo	NY
WQZM651	PEA 200	Danville	VA
WQZM652	PEA 240	Charlottesville	VA
WQZM653	PEA 167	Harrisonburg	VA
WQZM654	PEA 349	Marion	NC
WQZM655	PEA 205	Douglas City	CA
WQZM656	PEA 213	Bend	OR
WQZM657	PEA 364	Butte	MT
WQZM658	PEA 373	Walla Walla	WA
WQZM659		Jackson	WY
WQZM660	PEA 190	Bozeman	MT
WQZM661	PEA 369	Red Oak	IA
WQZM662	PEA 172	Duluth	MN
WQZM663	PEA 172	Duluth	MN
WQZM664	PEA 307	Yankton	SD
WQZM665	PEA 106	Zanesville	OH
WQZM666	PEA 097	Mankato	MN
WQZM667	PEA 052	Charleston	WV
WQZM668	PEA 059	Memphis	TN
WQZM669	PEA 377	Demopolis	AL
WQZM670	PEA 344	Clanton	AL

Call Sign	PEA Number	Municipality	<u>State</u>
WQZM671	PEA 300	Selma	AL
WQZM672	PEA 311	Trinidad	СО
WQZM673	PEA 372	Colby	KS
WQZM674	PEA 359	Sterling	СО
WQZM675	PEA 115	Asheville	NC
WQZM676	PEA 360	Juneau	AK
WQZM677	PEA 360	Juneau	AK
WQZM678	PEA 212	Anchorage	AK
WQZM679	PEA 212	Anchorage	AK
WQZM680	PEA 341	Alamogordo	NM
WQZM681	PEA 130	Spokane	WA
WQZM682	PEA 087	Pensacola	FL
WQZM683	PEA 166	Redding	CA
WQZM684	PEA 124	Olympia	WA
WQZM685	PEA 328	Winslow	AZ
WQZM686	PEA 109	Rocky Mount	NC
WQZM687	PEA 100	Greenville	NC
WQZM688	PEA 103	Winchester	VA
WQZM689	PEA 301	Rochester	MN
WQZM690	PEA 301	Rochester	MN
WQZM691	PEA 381	Del Rio	TX
WQZM692	PEA 095	Bluefield	WV
WQZM693	PEA 191	Petersburg	VA
WQZM694	PEA 177	Savannah	GA
WQZM695	PEA 024	Saint Louis	МО
WQZM696	PEA 024	Saint Louis	МО
WQZM697	PEA 184	Ruston	LA
WQZM698	PEA 246	Auburn	AL
WQZM699	PEA 192	Fayetteville	NC
WQZM700	PEA 173	Blacksburg	VA
WQZM701	PEA 202	Columbus	GA
WQZM702	PEA 249	Bryan	TX
WQZM703	PEA 356	Colville	WA
WQZM704	PEA 229	Saint George	UT
WQZM705	PEA 257	Cheyenne	WY
WQZM706	PEA 217	Lubbock	TX
WQZM707	PEA 189	Alexandria	LA
WQZM708	PEA 242	Lake Charles	LA
WQZM709	PEA 235	Amarillo	TX
WQZM710	PEA 220	Odessa	TX

Case 1:19-cv-02232 Document 2-2 Filed 07/26/19 Page 49 of 64

Call Sign	PEA Number	<u>Municipality</u>	<u>State</u>
WQZM711	PEA 120	Shreveport	LA
WQZM712	PEA 135	Beaumont	TX
WQZM713	PEA 317	Beatrice	NE
WQZM714	PEA 283	Plattsburgh	NY
WQZM715	PEA 227	Watertown	NY
WQZM716	PEA 210	Binghamton	NY
WQZM717	PEA 104	Fort Collins	СО

800 MHz Economic Area License Holdings at end of Rebanding

		Radio					
		Service		Channel		Expiration	
Call Sign	Licensee	Code	EA Market	Block	Market Name	Date	Frequency Range
WPLM546	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA001	В	Bangor, ME	6/17/2028 817 - 818 / 862	2 - 863
WPLM547	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA001	Χ	Bangor, ME	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM549	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA002	В	Portland, ME	6/17/2028 817 - 818 / 862	2 - 863
WPLM550	ATLANTIC, INC. NEXTEL COMMUNICATIONS OF THE MID	YH	BEA002	X	Portland, ME	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM552	ATLANTIC, INC. NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA003	В	Boston-Worcester-Lawrence-Lowe	6/17/2028 817 - 818 / 862	2 - 863
WPQT200	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA003	В	Boston-Worcester-Lawrence-Lowe	6/17/2028 817 - 818 / 862	2 - 863
WPLM553	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA003	X	Boston-Worcester-Lawrence-Lowe	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPQT201	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA003	X	Boston-Worcester-Lawrence-Lowe	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM555	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA004	В	Burlington, VT-NY	6/17/2028 817 - 818 / 862	2 - 863
WPLM556	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA004	X	Burlington, VT-NY	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPOI277	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA005	В	Albany-Schenectady-Troy, NY	6/17/2028 817 - 818 / 862	2 - 863
WPOI278	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA005	X	Albany-Schenectady-Troy, NY	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPOI280	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA006	В	Syracuse, NY-PA	6/17/2028 817 - 818 / 862	2 - 863
WPOI281	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA006	X	Syracuse, NY-PA	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM564	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA007	В	Rochester, NY-PA	6/17/2028 817 - 818 / 862	2 - 863
WPLM565	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA007	X	Rochester, NY-PA	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM567	•	YC	BEA008	В	Buffalo-Niagara Falls, NY-PA	6/17/2028 817 - 818 / 862	2 - 863
WPLM568	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YH	BEA008	X	Buffalo-Niagara Falls, NY-PA	6/17/2028 818 - 821 / 863	8 - 866 / 821 - 824 / 866 - 869
WPLM570	NEXTEL WEST CORP.	YC	BEA009	В	State College, PA	6/17/2028 817 - 818 / 862	· - 863
WPLM570	NEXTEL WEST CORP.	YH	BEA009	X	State College, PA		3 - 866 / 821 - 824 / 866 - 869
WPLM573	NEXTEL OF NEW YORK, INC.	YC	BEA010	В	New York-No. New JerLong Isl	6/17/2028 817 - 818 / 862	
WPOI316	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC, INC.	YC	BEA010	В	New York-No. New JerLong Isl	6/17/2028 817 - 818 / 862	
WPQS997	NEXTEL WEST CORP.	YC	BEA010	В	New York-No. New JerLong Isl	6/17/2028 817 - 818 / 862	2 - 863

WPVP595	NEXTEL WEST CORP.	YC	BEA010	В	New York-No. New JerLong Isl	6/17/2028 817 - 818 / 862 - 863
WPLM574	NEXTEL OF NEW YORK, INC.	YH	BEA010	Χ	New York-No. New JerLong Isl	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI317	NEXTEL COMMUNICATIONS OF THE MID	YH	BEA010	Χ	New York-No. New JerLong Isl	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					
WPQS998	NEXTEL WEST CORP.	YH	BEA010	Χ	New York-No. New JerLong Isl	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM576	NEXTEL WEST CORP.	YC	BEA011	В	Harrisburg-Lebanon-Carlisle, P	6/17/2028 817 - 818 / 862 - 863
WPOI283	NEXTEL WEST CORP.	YC	BEA011	В	Harrisburg-Lebanon-Carlisle, P	6/17/2028 817 - 818 / 862 - 863
WPLM577	NEXTEL WEST CORP.	YH	BEA011	Χ	Harrisburg-Lebanon-Carlisle, P	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI284	NEXTEL WEST CORP.	YH	BEA011	Χ	Harrisburg-Lebanon-Carlisle, P	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM579	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA012	В	Philadelphia-Wilmington-Atl. C	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.					
WPOI319	NEXTEL WEST CORP.	YC	BEA012	В	Philadelphia-Wilmington-Atl. C	6/17/2028 817 - 818 / 862 - 863
WPLM580	NEXTEL COMMUNICATIONS OF THE MID	YH	BEA012	Χ	Philadelphia-Wilmington-Atl. C	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					
WPOI320	NEXTEL WEST CORP.	YH	BEA012	Χ	Philadelphia-Wilmington-Atl. C	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM582	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA013	В	Washington-Baltimore, DC-MD-VA	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.			_		-1-1
WPOI322	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA013	В	Washington-Baltimore, DC-MD-VA	6/17/2028 817 - 818 / 862 - 863
MADOTAGA	ATLANTIC, INC.	\ <u>'</u>	DE 4 0 4 2		W. I B.W. BCMBWA	C 147 12020 047 040 10C2 0C2
WPQT203	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA013	В	Washington-Baltimore, DC-MD-VA	6/17/2028 817 - 818 / 862 - 863
WPLM583	ATLANTIC, INC. NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA013	Х	Washington Politimore DC MD VA	6/17/2020 010 021 / 062 066 / 021 024 / 066 060
WPLIVIDOS	ATLANTIC, INC.	ĭП	DEAU13	^	Washington-Baltimore, DC-MD-VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI323	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA013	Х	Washington-Baltimore, DC-MD-VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WI OIJZJ	ATLANTIC, INC.		DEAOIS	^	washington baltimore, be wib va	0/17/2020 010 021/ 003 000/ 021 024/ 000 003
WPQT204	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA013	Х	Washington-Baltimore, DC-MD-VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
4.20	ATLANTIC, INC.		22/1020		Trasmingeon Salemiore, Se ins tri	0,17,2020 020 022, 000 000, 022 021, 000 000
WPLM585	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA014	В	Salisbury, MD-DE-VA	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.				,	
WQHV920	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA014	В	Salisbury, MD-DE-VA	6/17/2028 817.0125 - 817.1125 / 862.0125 - 862.1125
	ATLANTIC, INC.					
WPLM586	NEXTEL COMMUNICATIONS OF THE MID	YH	BEA014	Χ	Salisbury, MD-DE-VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					
WPLM588	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA015	В	Richmond-Petersburg, VA	6/17/2028 817.4 - 818 / 862.4 - 863
	ATLANTIC, INC.					
WPOI413	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA015	В	Richmond-Petersburg, VA	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.					
WQHX606	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA015	В	Richmond-Petersburg, VA	6/17/2028 817 - 817.2 / 862 - 862.2
	ATLANTIC, INC.					
WQHX885	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA015	В	Richmond-Petersburg, VA	6/17/2028 817.2 - 817.4 / 862.2 - 862.4
\.\.\D\.\\AFOO	ATLANTIC, INC.	V (1)	DE 4045	v	D' L D L L MA	C/47/2020 040 024 / 052 055 / 024 024 / 055 050
WPLM589	NEXTEL COMMUNICATIONS OF THE MID	YH	BEA015	X	Richmond-Petersburg, VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
M/DOI/11/	ATLANTIC, INC. NEXTEL COMMUNICATIONS OF THE MID	YH	DEAO1E	Х	Richmond Retershurg VA	6/17/2020 010 021/062 066/021 024/066 060
WPOI414	ATLANTIC, INC.	тĦ	BEA015	^	Richmond-Petersburg, VA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM591	NEXTEL WEST CORP.	YC	BEA016	В	Staunton, VA-WV	6/17/2028 817 - 818 / 862 - 863
AA L FIAIDAT	WEATER WEST COM .	10	PLYOTO	ט	Stauritori, VA-VV V	0/11/2020 011 - 010 / 002 - 003

WPOI325	NEXTEL WEST CORP.	YC	BEA016	В	Staunton, VA-WV	6/17/2028 817 - 818 / 862 - 863
WPLM592	NEXTEL WEST CORP.	ΥH	BEA016	Χ	Staunton, VA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI326	NEXTEL WEST CORP.	ΥH	BEA016	Χ	Staunton, VA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM594	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA017	В	Roanoke, VA-NC-WV	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.					
WPQT206	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA017	В	Roanoke, VA-NC-WV	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.					
WQHV921	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA017	В	Roanoke, VA-NC-WV	6/17/2028 817.0125 - 817.1125 / 862.0125 - 862.1125
	ATLANTIC, INC.					
WPLM595	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA017	Χ	Roanoke, VA-NC-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					
WPQT235	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA017	Χ	Roanoke, VA-NC-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					
WPLM597	Nextel South Corp.	YC	BEA018	В	Greensboro-Winston-Salem-High	6/17/2028 817 - 818 / 862 - 863
WPOI328	Nextel South Corp.	YC	BEA018	В	Greensboro-Winston-Salem-High	6/17/2028 817 - 818 / 862 - 863
WPVQ580	Nextel South Corp.	YC	BEA018	В	Greensboro-Winston-Salem-High	6/17/2028 817 - 818 / 862 - 863
WPVV599	Nextel South Corp.	YC	BEA018	В	Greensboro-Winston-Salem-High	6/17/2028 817 - 818 / 862 - 863
WPLM598	Nextel South Corp.	ΥH	BEA018	Χ	Greensboro-Winston-Salem-High	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI329	Nextel South Corp.	ΥH	BEA018	Χ	Greensboro-Winston-Salem-High	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPVQ581	Nextel South Corp.	ΥH	BEA018	Χ	Greensboro-Winston-Salem-High	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPVV600	Nextel South Corp.	ΥH	BEA018	Χ	Greensboro-Winston-Salem-High	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM600	Nextel South Corp.	YC	BEA019	В	Raleigh-Durham-Chapel Hill, NC	6/17/2028 817 - 818 / 862 - 863
WQHV922	Nextel South Corp.	YC	BEA019	В	Raleigh-Durham-Chapel Hill, NC	6/17/2028 817.6375 - 817.7375 / 862.6375 - 862.7375
WQHX597	Nextel South Corp.	YC	BEA019	В	Raleigh-Durham-Chapel Hill, NC	6/17/2028 817.0125 - 817.4875 / 862.0125 - 862.4875
WQHX600	Nextel South Corp.	YC	BEA019	В	Raleigh-Durham-Chapel Hill, NC	6/17/2028 817.5125 - 817.6125 / 862.5125 - 862.6125
WPLM601	Nextel South Corp.	ΥH	BEA019	Χ	Raleigh-Durham-Chapel Hill, NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM603	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA020	В	Norfolk-Virginia Beach-Newport	6/17/2028 817 - 818 / 862 - 863
	ATLANTIC, INC.				Ŭ ,	, ,
WQHX607	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA020	В	Norfolk-Virginia Beach-Newport	6/17/2028 817.0125 - 817.3625 / 862.0125 - 862.3625
	ATLANTIC, INC.				Ŭ ,	, ,
WQHX886	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA020	В	Norfolk-Virginia Beach-Newport	6/17/2028 817.3875 - 817.7375 / 862.3875 - 862.7375
•	ATLANTIC, INC.				0 1 11 1	., ,
WPLM604	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA020	Χ	Norfolk-Virginia Beach-Newport	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	ATLANTIC, INC.					, , , , , , , , , , , , , , , , , , , ,
WPLM606	Nextel South Corp.	YC	BEA021	В	Greenville, NC	6/17/2028 817 - 818 / 862 - 863
WQHV924	Nextel South Corp.	YC	BEA021	В	Greenville, NC	6/17/2028 817.6375 - 817.7375 / 862.6375 - 862.7375
WQHX595	Nextel South Corp.	YC	BEA021	В	Greenville, NC	6/17/2028 817.2625 - 817.4875 / 862.2625 - 862.4875
WQHX598	Nextel South Corp.	YC	BEA021	В	Greenville, NC	6/17/2028 817.0125 - 817.2375 / 862.0125 - 862.2375
WQHX602	Nextel South Corp.	YC	BEA021	В	Greenville, NC	6/17/2028 817.5125 - 817.6125 / 862.5125 - 862.6125
WPLM607	Nextel South Corp.	YH	BEA021	X	Greenville, NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM609	Nextel South Corp.	YC	BEA022	В	Fayetteville, NC	6/17/2028 817 - 818 / 862 - 863
WQHV925	Nextel South Corp.	YC	BEA022	В	Fayetteville, NC	6/17/2028 817.3875 - 817.4875 / 862.3875 - 862.4875
WQHX603	Nextel South Corp.	YC	BEA022	В	Fayetteville, NC	6/17/2028 817.2625 - 817.3625 / 862.2625 - 862.3625
WQHX887	Nextel South Corp.	YC	BEA022	В	Fayetteville, NC	6/17/2028 817.0125 - 817.2375 / 862.0125 - 862.2375
WPLM610	Nextel South Corp.	YH	BEA022	X	Fayetteville, NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM612	Nextel South Corp.	YC	BEA023	В	Charlotte-Gastonia-Rock Hill,	6/17/2028 817 - 818 / 862 - 863
			52,1025	5	Charlette dustonia Rock Hilly	5, 2., 2020 31, 320, 302 303

WQHV928	Nextel South Corp.	YC	BEA023	В	Charlotte-Gastonia-Rock Hill,	6/17/2028 817.1375 - 817.2375 / 862.1375 - 862.2375
WQHX605	Nextel South Corp.	YC	BEA023	В	Charlotte-Gastonia-Rock Hill,	6/17/2028 817.0125 - 817.1125 / 862.0125 - 862.1125
WPLM613	Nextel South Corp.	ΥH	BEA023	Χ	Charlotte-Gastonia-Rock Hill,	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM615	Nextel South Corp.	YC	BEA024	В	Columbia, SC	6/17/2028 817 - 818 / 862 - 863
WQHV929	Nextel South Corp.	YC	BEA024	В	Columbia, SC	6/17/2028 817.0125 - 817.1125 / 862.0125 - 862.1125
WPLM616	Nextel South Corp.	ΥH	BEA024	Χ	Columbia, SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM618	Nextel South Corp.	YC	BEA025	В	Wilmington, NC-SC	6/17/2028 817 - 818 / 862 - 863
WQHV923	Nextel South Corp.	YC	BEA025	В	Wilmington, NC-SC	6/17/2028 817.5875 - 817.6875 / 862.5875 - 862.6875
WQHX596	Nextel South Corp.	YC	BEA025	В	Wilmington, NC-SC	6/17/2028 817.2375 - 817.4375 / 862.2375 - 862.4375
WQHX601	Nextel South Corp.	YC	BEA025	В	Wilmington, NC-SC	6/17/2028 817.4625 - 817.5625 / 862.4625 - 862.5625
WQHX888	Nextel South Corp.	YC	BEA025	В	Wilmington, NC-SC	6/17/2028 817.0125 - 817.2125 / 862.0125 - 862.2125
WPLM619	Nextel South Corp.	ΥH	BEA025	Χ	Wilmington, NC-SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM621	Nextel South Corp.	YC	BEA026	В	Charleston-North Charleston, S	6/17/2028 817 - 818 / 862 - 863
WQHV926	Nextel South Corp.	YC	BEA026	В	Charleston-North Charleston, S	6/17/2028 817.1375 - 817.2375 / 862.1375 - 862.2375
WQHX604	Nextel South Corp.	YC	BEA026	В	Charleston-North Charleston, S	6/17/2028 817.0125 - 817.1125 / 862.0125 - 862.1125
WPLM622	Nextel South Corp.	ΥH	BEA026	Χ	Charleston-North Charleston, S	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM624	Nextel South Corp.	YC	BEA027	В	Augusta-Aiken, GA-SC	6/17/2028 817.4 - 818 / 862.4 - 863
WPOK440	Nextel South Corp.	YC	BEA027	В	Augusta-Aiken, GA-SC	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM625	Nextel South Corp.	ΥH	BEA027	Χ	Augusta-Aiken, GA-SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK441	Nextel South Corp.	ΥH	BEA027	Χ	Augusta-Aiken, GA-SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM627	Nextel South Corp.	YC	BEA028	В	Savannah, GA-SC	6/17/2028 817.4 - 818 / 862.4 - 863
WPOK444	Nextel South Corp.	YC	BEA028	В	Savannah, GA-SC	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM628	Nextel South Corp.	ΥH	BEA028	Χ	Savannah, GA-SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK443	Nextel South Corp.	ΥH	BEA028	Χ	Savannah, GA-SC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM630	Nextel South Corp.	YC	BEA029	В	Jacksonville, FL-GA	6/17/2028 817 - 818 / 862 - 863
WPOI331	Nextel South Corp.	YC	BEA029	В	Jacksonville, FL-GA	6/17/2028 817 - 818 / 862 - 863
WPLM631	Nextel South Corp.	ΥH	BEA029	Χ	Jacksonville, FL-GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI332	Nextel South Corp.	ΥH	BEA029	Χ	Jacksonville, FL-GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM633	Nextel South Corp.	YC	BEA030	В	Orlando, FL	6/17/2028 817 - 818 / 862 - 863
WPLM634	Nextel South Corp.	ΥH	BEA030	Χ	Orlando, FL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM636	Nextel South Corp.	YC	BEA031	В	Miami-Fort Lauderdale, FL	6/17/2028 817 - 818 / 862 - 863
WPLM637	Nextel South Corp.	ΥH	BEA031	Χ	Miami-Fort Lauderdale, FL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM639	Nextel South Corp.	YC	BEA032	В	Fort Myers-Cape Coral, FL	6/17/2028 817 - 818 / 862 - 863
WPLM640	Nextel South Corp.	ΥH	BEA032	Χ	Fort Myers-Cape Coral, FL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM642	Nextel South Corp.	YC	BEA033	В	Sarasota-Bradenton, FL	6/17/2028 817 - 818 / 862 - 863
WPLM643	Nextel South Corp.	ΥH	BEA033	Χ	Sarasota-Bradenton, FL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM645	Nextel South Corp.	YC	BEA034	В	Tampa-St. Petersburg-Clearwate	6/17/2028 817 - 818 / 862 - 863
WPLM646	Nextel South Corp.	ΥH	BEA034	Χ	Tampa-St. Petersburg-Clearwate	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM647	Nextel South Corp.	YC	BEA035	В	Tallahassee, FL-GA	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM648	Nextel South Corp.	ΥH	BEA035	Χ	Tallahassee, FL-GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR938	Nextel South Corp.	ΥH	BEA035	Χ	Tallahassee, FL-GA	3/21/2021 817.4 - 818 / 862.4 - 863
WPOI289	Nextel South Corp.	YC	BEA036	В	Dothan, AL-FL-GA	6/17/2028 817.4 - 818 / 862.4 - 863
WPOI290	Nextel South Corp.	ΥH	BEA036	Χ	Dothan, AL-FL-GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR936	Nextel South Corp.	ΥH	BEA036	Χ	Dothan, AL-FL-GA	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM652	Nextel South Corp.	YC	BEA037	В	Albany, GA	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM653	Nextel South Corp.	ΥH	BEA037	Χ	Albany, GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPOK449	Nextel South Corp.	YC	BEA038	В	Macon, GA	6/17/2028 817.4 - 818 / 862.4 - 863
WPOK450	Nextel South Corp.	YH	BEA038	Χ	Macon, GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK451	Nextel South Corp.	YC	BEA039	В	Columbus, GA-AL	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM658	Nextel South Corp.	ΥH	BEA039	Χ	Columbus, GA-AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI376	Nextel South Corp.	ΥH	BEA039	Χ	Columbus, GA-AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK452	Nextel South Corp.	YH	BEA039	Χ	Columbus, GA-AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR937	Nextel South Corp.	ΥH	BEA039	Χ	Columbus, GA-AL	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM660	Nextel South Corp.	YC	BEA040	В	Atlanta, GA-AL-NC	6/17/2028 817.4 - 818 / 862.4 - 863
WPOK454	Nextel South Corp.	YC	BEA040	В	Atlanta, GA-AL-NC	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM661	Nextel South Corp.	ΥH	BEA040	Χ	Atlanta, GA-AL-NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI335	Nextel South Corp.	ΥH	BEA040	Χ	Atlanta, GA-AL-NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK455	Nextel South Corp.	ΥH	BEA040	Χ	Atlanta, GA-AL-NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR939	Nextel South Corp.	ΥH	BEA040	Χ	Atlanta, GA-AL-NC	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM663	Nextel South Corp.	YC	BEA041	В	Greenville-Spartanburg-Anderso	6/17/2028 817 - 818 / 862 - 863
WPLM664	Nextel South Corp.	ΥH	BEA041	Χ	Greenville-Spartanburg-Anderso	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM666	Nextel South Corp.	YC	BEA042	В	Asheville, NC	6/17/2028 817 - 818 / 862 - 863
WPLM667	Nextel South Corp.	ΥH	BEA042	Χ	Asheville, NC	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM669	Nextel South Corp.	YC	BEA043	В	Chattanooga, TN-GA	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM670	Nextel South Corp.	ΥH	BEA043	Χ	Chattanooga, TN-GA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM672	Nextel South Corp.	YC	BEA044	В	Knoxville, TN	6/17/2028 817 - 818 / 862 - 863
WPOI337	Nextel South Corp.	YC	BEA044	В	Knoxville, TN	6/17/2028 817 - 818 / 862 - 863
WPLM673	Nextel South Corp.	ΥH	BEA044	Χ	Knoxville, TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI338	Nextel South Corp.	ΥH	BEA044	Χ	Knoxville, TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM675	Nextel South Corp.	YC	BEA045	В	Johnson City-Kingsport-Bristol	6/17/2028 817 - 818 / 862 - 863
WPLM676	Nextel South Corp.	ΥH	BEA045	Χ	Johnson City-Kingsport-Bristol	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM678	Nextel South Corp.	YC	BEA046	В	Hickory-Morganton, NC-TN	6/17/2028 817 - 818 / 862 - 863
WPOI340	Nextel South Corp.	YC	BEA046	В	Hickory-Morganton, NC-TN	6/17/2028 817 - 818 / 862 - 863
WPLM679	Nextel South Corp.	ΥH	BEA046	Χ	Hickory-Morganton, NC-TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI341	Nextel South Corp.	ΥH	BEA046	Χ	Hickory-Morganton, NC-TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM681	NEXTEL WEST CORP.	YC	BEA047	В	Lexington, KY-TN-VA-WV	6/17/2028 817 - 818 / 862 - 863
WPOI378	NEXTEL WEST CORP.	YC	BEA047	В	Lexington, KY-TN-VA-WV	6/17/2028 817 - 818 / 862 - 863
WPLM682	NEXTEL WEST CORP.	ΥH	BEA047	Χ	Lexington, KY-TN-VA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI379	NEXTEL WEST CORP.	ΥH	BEA047	Χ	Lexington, KY-TN-VA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM684	NEXTEL WEST CORP.	YC	BEA048	В	Charleston, WV-KY-OH	6/17/2028 817 - 818 / 862 - 863
WPLM685	NEXTEL WEST CORP.	ΥH	BEA048	Χ	Charleston, WV-KY-OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM687	NEXTEL WEST CORP.	YC	BEA049	В	Cincinnati-Hamilton, OH-KY-IN	6/17/2028 817 - 818 / 862 - 863
WPQT236	NEXTEL WEST CORP.	YC	BEA049	В	Cincinnati-Hamilton, OH-KY-IN	6/17/2028 817 - 818 / 862 - 863
WPLM688	NEXTEL WEST CORP.	ΥH	BEA049	Χ	Cincinnati-Hamilton, OH-KY-IN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT237	NEXTEL WEST CORP.	ΥH	BEA049	Χ	Cincinnati-Hamilton, OH-KY-IN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM690	NEXTEL WEST CORP.	YC	BEA050	В	Dayton-Springfield, OH	6/17/2028 817 - 818 / 862 - 863
WPLM691	NEXTEL WEST CORP.	ΥH	BEA050	Χ	Dayton-Springfield, OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM693	NEXTEL WEST CORP.	YC	BEA051	В	Columbus, OH	6/17/2028 817 - 818 / 862 - 863
WPQS976	NEXTEL WEST CORP.	YC	BEA051	В	Columbus, OH	6/17/2028 817 - 818 / 862 - 863
WPLM694	NEXTEL WEST CORP.	ΥH	BEA051	Χ	Columbus, OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQS977	NEXTEL WEST CORP.	ΥH	BEA051	Χ	Columbus, OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM696	NEXTEL WEST CORP.	YC	BEA052	В	Wheeling, WV-OH	6/17/2028 817 - 818 / 862 - 863
-		-			<i>5</i> ,	• • • • • • • • • • • • • • • • • • • •

WPILMOP NEXTEL WEST CORP YH BEADS2 X Wheeling, WV-OH 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS3 B PILSBurgh, PA-WW 6,17/2028 818-821/863-866, 821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS3 B PILSBurgh, PA-WW 6,17/2028 817-818/862-863 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBurgh, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBurgh, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBurgh, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBURGH, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBURGH, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS3 X PILSBURGH, PA-WW 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YH BEADS4 X ENGINE WEST CORP YH BEADS4 X ENGINE WEST CORP YH BEADS5 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS5 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AKTON, OH-PA 6,17/2028 818-821/863-866/821-824/866-869 WPILMOP NEXTEL WEST CORP YC BEADS6 X CUEWIANG-AK	WPQS979	NEXTEL WEST CORP.	YC	BEA052	В	Wheeling, WV-OH	6/17/2028 817 - 818 / 862 - 863
WPLM998 NEXTEL WEST CORP.	WPLM697	NEXTEL WEST CORP.	ΥH	BEA052	Χ	Wheeling, WV-OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLDAY0 WPLTEL WEST CORP.	WPQS980	NEXTEL WEST CORP.	ΥH	BEA052	Χ	Wheeling, WV-OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLIM700 NEXTE WEST CORP.	WPLM699	NEXTEL WEST CORP.	YC	BEA053	В	Pittsburgh, PA-WV	6/17/2028 817 - 818 / 862 - 863
WPC-1982 NEXTEL WEST CORP. H BEAGS W PILSUNGTh, PA-WV G1/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869	WPQS982	NEXTEL WEST CORP.	YC	BEA053	В	Pittsburgh, PA-WV	6/17/2028 817 - 818 / 862 - 863
WPLIM70 NEXTEL COMMUNICATIONS OF THE MID VC BEA054 S Frie, PA 6/17/2028 817 - 818 / 862 - 863	WPLM700	NEXTEL WEST CORP.	ΥH	BEA053	Χ	Pittsburgh, PA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
ATLANTEL INC.	WPQS983	NEXTEL WEST CORP.	ΥH	BEA053	Χ	Pittsburgh, PA-WV	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLIM703 NEXTEL COMMUNICATIONS OF THE MID YH BEA054 X Erie, PA 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XTLANTIC, INC. WPLIM703 XTREE COMMUNICATIONS OF THE MID YH BEA055 X Erie, PA 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XTLANTIC, INC. WPLIM705 XTREE COMMUNICATIONS OF THE MID YH BEA055 X Erie, PA 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XTLANTIC, INC. WPLIM705 XTLEE COMMUNICATIONS OF THE MID YH BEA055 X Erie, PA 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA055 X Cleveland-Akron, OH-PA 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA056 X Toledo, OH 6/17/2028 817 - 818 / 862 - 863 XPLEE WEST CORP. YC BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 817 - 818 / 862 - 863 XPLEE WEST CORP. YC BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 817 - 818 / 862 - 863 XPLEE WEST CORP. YC BEA058 X Northern Michigan, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA059 X ORTHER WEST CORP. YC BEA059 X ORTHER WEST CORP. YC BEA059 X ORTHER WEST CORP. YC BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA069 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 XPLEE WEST CORP. YC BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 8	WPLM702	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA054	В	Erie, PA	6/17/2028 817 - 818 / 862 - 863
ATLANTIC, INC.		ATLANTIC, INC.					
MYPLM703 MEXTEL COMMUNICATIONS OF THE MID YH	WPOI381	NEXTEL COMMUNICATIONS OF THE MID	YC	BEA054	В	Erie, PA	6/17/2028 817 - 818 / 862 - 863
ATLANTIC, INC. WPOINS 2 NEXTEL COMMUNICATIONS OF THE MID DY ATLANTIC, INC. WPLAY705 NEXTEL WEST CORP. YC BEA055 B Cleveland-Akron, OH-PA 6/17/2028 817-818/862-863 WPLAY706 NEXTEL WEST CORP. YC BEA056 B Toledo, OH 6/17/2028 817-818/862-863 WPLAY708 NEXTEL WEST CORP. YC BEA056 B Toledo, OH 6/17/2028 817-818/862-863 WPLAY708 NEXTEL WEST CORP. YC BEA056 B Toledo, OH 6/17/2028 817-818/862-863 WPLAY709 NEXTEL WEST CORP. YC BEA057 B Detroit Ann Arbor-Flint, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY711 NEXTEL WEST CORP. YC BEA057 B Detroit Ann Arbor-Flint, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY712 NEXTEL WEST CORP. YC BEA057 B Detroit Ann Arbor-Flint, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY713 NEXTEL WEST CORP. YC BEA058 B Northern Michigan, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY714 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY715 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY718 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY718 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY71 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY72 NEXTEL WEST CORP. YC BEA069 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY72 NEXTEL WEST CORP. YC BEA069 B Green Bay, WI-MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY72 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY2 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY2 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY2 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY2 NEXTEL WEST CORP. YC BEA063 B MINWALWE-Racine, WI 6/17/2028 818-821/863-866/821-824/866-869 WPLAY2 NEXT		ATLANTIC, INC.					
NEXTEL COMMUNICATIONS OF THE MID	WPLM703	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA054	Χ	Erie, PA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
ATLANTIC, INC.		ATLANTIC, INC.					
WPLM705 NEXTEL WEST CORP.	WPOI382	NEXTEL COMMUNICATIONS OF THE MID	ΥH	BEA054	Χ	Erie, PA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM706 NEXTEL WEST CORP. YH BEA055 X Cleveland-Akron, OH-PA 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM706 NEXTEL WEST CORP. YH BEA056 X Toledo, OH 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM711 NEXTEL WEST CORP. YH BEA057 B Detroit-Ann Arbor-Flint, MI 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM712 NEXTEL WEST CORP. YH BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM714 NEXTEL WEST CORP. YH BEA058 X Northern Michigan, MI 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM715 NEXTEL WEST CORP. YH BEA058 X Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 WPLM715 NEXTEL WEST CORP. YH BEA058 X Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 WPLM715 NEXTEL WEST CORP. YH BEA058 X Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 WPLM715 NEXTEL WEST CORP. YH BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 817 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM728 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM728 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Hollan		ATLANTIC, INC.					
WPLM708 NEXTEL WEST CORP.	WPLM705	NEXTEL WEST CORP.	YC	BEA055	В	Cleveland-Akron, OH-PA	6/17/2028 817 - 818 / 862 - 863
WPLM709 NEXTEL WEST CORP.	WPLM706	NEXTEL WEST CORP.	ΥH	BEA055	Χ	Cleveland-Akron, OH-PA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM711 NEXTEL WEST CORP. YC BEA057 B Detroit-Ann Arbor-Flint, MI 6/17/2028 817 - 818 / 862 - 863 WPLM712 NEXTEL WEST CORP. YH BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 813 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM714 NEXTEL WEST CORP. YC BEA058 X Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 WPLM717 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPVP602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YC BEA059 X Green Bay, WI-MI 6/17/2028 817 - 818 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM7210 NEXTEL WEST CORP. YC BEA059 X	WPLM708	NEXTEL WEST CORP.	YC	BEA056	В	Toledo, OH	6/17/2028 817 - 818 / 862 - 863
WPLM712 NEXTEL WEST CORP. YH BEA057 X Detroit-Ann Arbor-Flint, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM714 NEXTEL WEST CORP. YC BEA058 B Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 WPLM717 NEXTEL WEST CORP. YC BEA059 B Green Bay, Wi-MI 6/17/2028 817 - 818 / 862 - 863 WPUM717 NEXTEL WEST CORP. YC BEA059 B Green Bay, Wi-MI 6/17/2028 817 - 818 / 862 - 863 WPUM384 NEXTEL WEST CORP. YC BEA059 B Green Bay, Wi-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, Wi-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, Wi-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, Wi-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, W	WPLM709	NEXTEL WEST CORP.	ΥH	BEA056	Χ	Toledo, OH	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM714 NEXTEL WEST CORP. YC BEA058 B Northern Michigan, MI 6/17/2028 817 - 818 / 862 - 863 86 / 821 - 824 / 866 - 869 WPLM715 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 866 / 821 - 824 / 866 - 869 WPD0334 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPVP602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YC BEA059 X Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA069 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. <td>WPLM711</td> <td>NEXTEL WEST CORP.</td> <td>YC</td> <td>BEA057</td> <td>В</td> <td>Detroit-Ann Arbor-Flint, MI</td> <td>6/17/2028 817 - 818 / 862 - 863</td>	WPLM711	NEXTEL WEST CORP.	YC	BEA057	В	Detroit-Ann Arbor-Flint, MI	6/17/2028 817 - 818 / 862 - 863
WPLM715 NEXTEL WEST CORP. YH BEA058 X Northern Michigan, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM717 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPV9602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPV9602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. YC BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM723 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM724 NEXTEL WEST CORP. YC BEA061 X Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM724 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM725 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM730 NEXTEL WEST CORP. YC BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM730 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-	WPLM712	NEXTEL WEST CORP.	ΥH	BEA057	Χ	Detroit-Ann Arbor-Flint, MI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM717 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPO0384 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPVP602 NEXTEL WEST CORP. YC BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM718 NEXTEL WEST CORP. YH BEA0599 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YH BEA061 B Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM723 NEXTEL WEST CORP. YH BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 <td>WPLM714</td> <td>NEXTEL WEST CORP.</td> <td>YC</td> <td>BEA058</td> <td>В</td> <td>Northern Michigan, MI</td> <td>6/17/2028 817 - 818 / 862 - 863</td>	WPLM714	NEXTEL WEST CORP.	YC	BEA058	В	Northern Michigan, MI	6/17/2028 817 - 818 / 862 - 863
WPU602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPVF602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. YC BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM723 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM724 NEXTEL WEST CORP. YC BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM731 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 866 WPO1344 NEXTEL WEST CORP. YC BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 866 WPO1347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 866 WPO1340 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 866 WPLM735 NEXTEL WEST CORP. YH BEA065 X El	WPLM715	NEXTEL WEST CORP.	ΥH	BEA058	Χ	Northern Michigan, MI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPVP602 NEXTEL WEST CORP. YC BEA059 B Green Bay, WI-MI 6/17/2028 817 - 818 / 862 - 863 WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YH BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM726 NEXTEL WEST CORP. YH BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM730 NEXTEL WEST CORP. YC BEA063 B Milwaukee-R	WPLM717	NEXTEL WEST CORP.	YC	BEA059	В	Green Bay, WI-MI	6/17/2028 817 - 818 / 862 - 863
WPLM718 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPO1385 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YC BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM721 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM724 NEXTEL WEST CORP. YC BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST	WPOI384	NEXTEL WEST CORP.	YC	BEA059	В	Green Bay, WI-MI	6/17/2028 817 - 818 / 862 - 863
WPO1385 NEXTEL WEST CORP. YH BEA059 X Green Bay, WI-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM720 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM724 NEXTEL WEST CORP. YH BEA061 X Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM726 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwau	WPVP602	NEXTEL WEST CORP.	YC	BEA059		Green Bay, WI-MI	6/17/2028 817 - 818 / 862 - 863
WPLM720 NEXTEL WEST CORP. YC BEA060 B Appleton-Oshkosh-Neenah, WI 6/17/2028 817 - 818 / 862 - 863 WPLM721 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM723 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YH BEA063 X	WPLM718	NEXTEL WEST CORP.	ΥH	BEA059	Χ	Green Bay, WI-MI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM721 NEXTEL WEST CORP. YH BEA060 X Appleton-Oshkosh-Neenah, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM723 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM724 NEXTEL WEST CORP. YH BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 866 / 821 - 824 / 866 - 869 WPLM727 NEXTEL WEST CORP. YC BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732	WPOI385	NEXTEL WEST CORP.	ΥH	BEA059	Χ	Green Bay, WI-MI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM723 NEXTEL WEST CORP. YC BEA061 B Traverse City, MI 6/17/2028 817 - 818 / 862 - 863 WPLM724 NEXTEL WEST CORP. YH BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPOI344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B	WPLM720	NEXTEL WEST CORP.	YC	BEA060	В	Appleton-Oshkosh-Neenah, WI	6/17/2028 817 - 818 / 862 - 863
WPLM724 NEXTEL WEST CORP. YH BEA061 X Traverse City, MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPO1343 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPUM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPUM732 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH	WPLM721	NEXTEL WEST CORP.	ΥH	BEA060	Χ	Appleton-Oshkosh-Neenah, WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM726 NEXTEL WEST CORP. YC BEA062 B Grand Rapids-Muskegon-Holland, 6/17/2028 817 - 818 / 862 - 863 WPLM727 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPD1344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPD1344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM735 NEXTEL WEST CORP. YH <td< td=""><td>WPLM723</td><td>NEXTEL WEST CORP.</td><td>YC</td><td>BEA061</td><td>В</td><td>Traverse City, MI</td><td>6/17/2028 817 - 818 / 862 - 863</td></td<>	WPLM723	NEXTEL WEST CORP.	YC	BEA061	В	Traverse City, MI	6/17/2028 817 - 818 / 862 - 863
WPLM727 NEXTEL WEST CORP. YH BEA062 X Grand Rapids-Muskegon-Holland, 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPOI343 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL	WPLM724	NEXTEL WEST CORP.	ΥH	BEA061	Χ	Traverse City, MI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM729 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPOI343 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPOI346 NEXTEL WEST CORP. YH BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM736 NEXTEL WEST CORP.	WPLM726	NEXTEL WEST CORP.	YC	BEA062	В	Grand Rapids-Muskegon-Holland,	6/17/2028 817 - 818 / 862 - 863
WPOI343 NEXTEL WEST CORP. YC BEA063 B Milwaukee-Racine, WI 6/17/2028 817 - 818 / 862 - 863 WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM736 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM738 NEXTEL WEST CORP.	WPLM727	NEXTEL WEST CORP.		BEA062		Grand Rapids-Muskegon-Holland,	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM730 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPO1344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM736 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863	WPLM729	NEXTEL WEST CORP.	YC	BEA063	В	Milwaukee-Racine, WI	6/17/2028 817 - 818 / 862 - 863
WPOI344 NEXTEL WEST CORP. YH BEA063 X Milwaukee-Racine, WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPOI346 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863	WPOI343	NEXTEL WEST CORP.	YC	BEA063		Milwaukee-Racine, WI	6/17/2028 817 - 818 / 862 - 863
WPLM732 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPOI346 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YH BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863	WPLM730	NEXTEL WEST CORP.	ΥH	BEA063	Χ	Milwaukee-Racine, WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI346 NEXTEL WEST CORP. YC BEA064 B Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 817 - 818 / 862 - 863 WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863	WPOI344	NEXTEL WEST CORP.		BEA063	Χ	Milwaukee-Racine, WI	
WPLM733 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863	WPLM732	NEXTEL WEST CORP.	YC	BEA064		Chicago-Gary-Kenosha, IL-IN-WI	6/17/2028 817 - 818 / 862 - 863
WPOI347 NEXTEL WEST CORP. YH BEA064 X Chicago-Gary-Kenosha, IL-IN-WI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863							·
WPLM735 NEXTEL WEST CORP. YC BEA065 B Elkhart-Goshen, IN-MI 6/17/2028 817 - 818 / 862 - 863 WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863						Chicago-Gary-Kenosha, IL-IN-WI	
WPLM736 NEXTEL WEST CORP. YH BEA065 X Elkhart-Goshen, IN-MI 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869 WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863						Chicago-Gary-Kenosha, IL-IN-WI	
WPLM738 NEXTEL WEST CORP. YC BEA066 B Fort Wayne, IN 6/17/2028 817 - 818 / 862 - 863							
\cdot							
WPLM739 NEXTEL WEST CORP. YH BEA066 X Fort Wayne, IN 6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869						•	·
	WPLM739	NEXTEL WEST CORP.	ΥH	BEA066	Χ	Fort Wayne, IN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPLM741	NEXTEL WEST CORP.	YC	BEA067	В	Indianapolis, IN-IL	6/17/2028 817 - 818 / 862 - 863
WPOI349	NEXTEL WEST CORP.	YC	BEA067	В	Indianapolis, IN-IL	6/17/2028 817 - 818 / 862 - 863
WPOK457	NEXTEL WEST CORP.	YC	BEA067	В	Indianapolis, IN-IL	6/17/2028 817 - 818 / 862 - 863
WPLM742	NEXTEL WEST CORP.	ΥH	BEA067	Χ	Indianapolis, IN-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI350	NEXTEL WEST CORP.	ΥH	BEA067	Χ	Indianapolis, IN-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK458	NEXTEL WEST CORP.	ΥH	BEA067	Χ	Indianapolis, IN-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI292	NEXTEL WEST CORP.	YC	BEA068	В	Champaign-Urbana, IL	6/17/2028 817 - 818 / 862 - 863
WPOI293	NEXTEL WEST CORP.	ΥH	BEA068	Χ	Champaign-Urbana, IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM747	NEXTEL WEST CORP.	YC	BEA069	В	Evansville-Henderson, IN-KY-IL	6/17/2028 817 - 818 / 862 - 863
WPLM748	NEXTEL WEST CORP.	ΥH	BEA069	Χ	Evansville-Henderson, IN-KY-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM750	NEXTEL WEST CORP.	YC	BEA070	В	Louisville, KY-IN	6/17/2028 817 - 818 / 862 - 863
WPOI295	NEXTEL WEST CORP.	YC	BEA070	В	Louisville, KY-IN	6/17/2028 817 - 818 / 862 - 863
WPLM751	NEXTEL WEST CORP.	ΥH	BEA070	Χ	Louisville, KY-IN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI296	NEXTEL WEST CORP.	ΥH	BEA070	Χ	Louisville, KY-IN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI355	NEXTEL WEST CORP.	YC	BEA071	В	Nashville, TN-KY	6/17/2028 817 - 818 / 862 - 863
WPSN389	Nextel South Corp.	YC	BEA071	В	Nashville, TN-KY	6/17/2028 817 - 818 / 862 - 863
WPOI356	NEXTEL WEST CORP.	ΥH	BEA071	Χ	Nashville, TN-KY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPSN390	Nextel South Corp.	ΥH	BEA071	Χ	Nashville, TN-KY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM756	NEXTEL WEST CORP.	YC	BEA072	В	Paducah, KY-IL	6/17/2028 817 - 818 / 862 - 863
WPLM757	NEXTEL WEST CORP.	YH	BEA072	Χ	Paducah, KY-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM759	Nextel South Corp.	YC	BEA073	В	Memphis, TN-AR-MS-KY	6/17/2028 817 - 818 / 862 - 863
WPOK460	Nextel South Corp.	YC	BEA073	В	Memphis, TN-AR-MS-KY	6/17/2028 817 - 818 / 862 - 863
WPLM760	Nextel South Corp.	ΥH	BEA073	Χ	Memphis, TN-AR-MS-KY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK461	Nextel South Corp.	ΥH	BEA073	Χ	Memphis, TN-AR-MS-KY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM761	Nextel South Corp.	YC	BEA074	В	Huntsville, AL-TN	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM762	Nextel South Corp.	ΥH	BEA074	Χ	Huntsville, AL-TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR940	Nextel South Corp.	YH	BEA074	Χ	Huntsville, AL-TN	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM763	Nextel South Corp.	YC	BEA075	В	Tupelo, MS-AL-TN	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM764	Nextel South Corp.	ΥH	BEA075	Χ	Tupelo, MS-AL-TN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR941	Nextel South Corp.	YH	BEA075	Χ	Tupelo, MS-AL-TN	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM765	Nextel South Corp.	YC	BEA076	В	Greenville, MS	6/17/2028 817 - 818 / 862 - 863
WPOI386	Nextel South Corp.	YC	BEA076	В	Greenville, MS	6/17/2028 817 - 818 / 862 - 863
WPLM766	Nextel South Corp.	YH	BEA076	Χ	Greenville, MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI387	Nextel South Corp.	YH	BEA076	Χ	Greenville, MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM768	Nextel South Corp.	YC	BEA077	В	Jackson, MS-AL-LA	6/17/2028 817.4 - 818 / 862.4 - 863
WPOI298	Nextel South Corp.	YC	BEA077	В	Jackson, MS-AL-LA	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM769	Nextel South Corp.	ΥH	BEA077	Χ	Jackson, MS-AL-LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI299	Nextel South Corp.	ΥH	BEA077	Χ	Jackson, MS-AL-LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR942	Nextel South Corp.	ΥH	BEA077	Χ	Jackson, MS-AL-LA	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM771	Nextel South Corp.	YH	BEA078	Χ	Birmingham, AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI358	Nextel South Corp.	YH	BEA078	Χ	Birmingham, AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR931	Nextel South Corp.	ΥH	BEA078	Χ	Birmingham, AL	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM773	Nextel South Corp.	YH	BEA079	Χ	Montgomery, AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR932	Nextel South Corp.	YH	BEA079	Χ	Montgomery, AL	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM775	Nextel South Corp.	YH	BEA080	Χ	Mobile, AL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR933	Nextel South Corp.	YH	BEA080	Χ	Mobile, AL	3/21/2021 817.4 - 818 / 862.4 - 863

WPLM777	Nextel South Corp.	YH	BEA081	Χ	Pensacola, FL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR934	Nextel South Corp.	ΥH	BEA081	Χ	Pensacola, FL	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM779	Nextel South Corp.	ΥH	BEA082	Χ	Biloxi-Gulfport-Pascagoula, MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WQVR935	Nextel South Corp.	ΥH	BEA082	Χ	Biloxi-Gulfport-Pascagoula, MS	3/21/2021 817.4 - 818 / 862.4 - 863
WPLM781	Nextel South Corp.	YC	BEA083	В	New Orleans, LA-MS	6/17/2028 817 - 818 / 862 - 863
WQVR930	Nextel South Corp.	YC	BEA083	В	New Orleans, LA-MS	6/17/2028 817.4 - 818 / 862.4 - 863
WPLM782	Nextel South Corp.	ΥH	BEA083	Χ	New Orleans, LA-MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI361	Nextel South Corp.	ΥH	BEA083	Χ	New Orleans, LA-MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM784	Nextel South Corp.	YC	BEA084	В	Baton Rouge, LA-MS	6/17/2028 817 - 818 / 862 - 863
WPLM785	Nextel South Corp.	ΥH	BEA084	Χ	Baton Rouge, LA-MS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM787	Nextel South Corp.	YC	BEA085	В	Lafayette, LA	6/17/2028 817 - 818 / 862 - 863
WPOI301	Nextel South Corp.	YC	BEA085	В	Lafayette, LA	6/17/2028 817 - 818 / 862 - 863
WPLM788	Nextel South Corp.	ΥH	BEA085	Χ	Lafayette, LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI302	Nextel South Corp.	ΥH	BEA085	Χ	Lafayette, LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM790	Nextel South Corp.	YC	BEA086	В	Lake Charles, LA	6/17/2028 817 - 818 / 862 - 863
WPLM791	Nextel South Corp.	YH	BEA086	Χ	Lake Charles, LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM793	NEXTEL WEST CORP.	YC	BEA087	В	Beaumont-Port Arthur, TX	6/17/2028 817 - 818 / 862 - 863
WPLM794	NEXTEL WEST CORP.	ΥH	BEA087	Χ	Beaumont-Port Arthur, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI304	Nextel South Corp.	YC	BEA088	В	Shreveport-Bossier City, LA-AR	6/17/2028 817 - 818 / 862 - 863
WPOI305	Nextel South Corp.	ΥH	BEA088	Χ	Shreveport-Bossier City, LA-AR	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM799	Nextel South Corp.	YC	BEA089	В	Monroe, LA	6/17/2028 817 - 818 / 862 - 863
WPLM800	Nextel South Corp.	YH	BEA089	Χ	Monroe, LA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH201	Nextel South Corp.	YC	BEA090	В	Little Rock-North Little Rock,	6/17/2028 817 - 818 / 862 - 863
WPOH202	Nextel South Corp.	ΥH	BEA090	Χ	Little Rock-North Little Rock,	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH203	NEXTEL WEST CORP.	YC	BEA091	В	Fort Smith, AR-OK	6/17/2028 817 - 818 / 862 - 863
WPOH204	NEXTEL WEST CORP.	YH	BEA091	Χ	Fort Smith, AR-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH205	Nextel South Corp.	YC	BEA092	В	Fayetteville-Springdale-Rogers	6/17/2028 817 - 818 / 862 - 863
WPOK462	NEXTEL WEST CORP.	YC	BEA092	В	Fayetteville-Springdale-Rogers	6/17/2028 817 - 818 / 862 - 863
WPOH206	Nextel South Corp.	YH	BEA092	Χ	Fayetteville-Springdale-Rogers	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK463	NEXTEL WEST CORP.	YH	BEA092	Χ	Fayetteville-Springdale-Rogers	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH208	NEXTEL WEST CORP.	YC	BEA093	В	Joplin, MO-KS-OK	6/17/2028 817 - 818 / 862 - 863
WPOH209	NEXTEL WEST CORP.	YH	BEA093	Χ	Joplin, MO-KS-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH211	NEXTEL WEST CORP.	YC	BEA094	В	Springfield, MO	6/17/2028 817 - 818 / 862 - 863
WPOH212	NEXTEL WEST CORP.	YH	BEA094	Χ	Springfield, MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH213	Nextel South Corp.	YC	BEA095	В	Jonesboro, AR-MO	6/17/2028 817 - 818 / 862 - 863
WPOK464	Nextel South Corp.	YC	BEA095	В	Jonesboro, AR-MO	6/17/2028 817 - 818 / 862 - 863
WPOH214	Nextel South Corp.	YH	BEA095	Χ	Jonesboro, AR-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK465	Nextel South Corp.	YH	BEA095	Χ	Jonesboro, AR-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH216	NEXTEL WEST CORP.	YC	BEA096	В	St. Louis, MO-IL	6/17/2028 817 - 818 / 862 - 863
WPOK467	NEXTEL WEST CORP.	YC	BEA096	В	St. Louis, MO-IL	6/17/2028 817 - 818 / 862 - 863
WPOH217	NEXTEL WEST CORP.	YH	BEA096	Χ	St. Louis, MO-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK468	NEXTEL WEST CORP.	YH	BEA096	Χ	St. Louis, MO-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH219	NEXTEL WEST CORP.	YC	BEA097	В	Springfield, IL-MO	6/17/2028 817 - 818 / 862 - 863
WPOI389	NEXTEL WEST CORP.	YC	BEA097	В	Springfield, IL-MO	6/17/2028 817 - 818 / 862 - 863
WPOH220	NEXTEL WEST CORP.	YH	BEA097	Χ	Springfield, IL-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI390	NEXTEL WEST CORP.	YH	BEA097	Χ	Springfield, IL-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPOH222	NEXTEL WEST CORP.	YC	BEA098	В	Columbia, MO	6/17/2028 817 - 818 / 862 - 863
WPOH223	NEXTEL WEST CORP.	ΥH	BEA098	Χ	Columbia, MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH225	NEXTEL WEST CORP.	YC	BEA099	В	Kansas City, MO-KS	6/17/2028 817 - 818 / 862 - 863
WPOH226	NEXTEL WEST CORP.	ΥH	BEA099	Χ	Kansas City, MO-KS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH228	NEXTEL WEST CORP.	YC	BEA100	В	Des Moines, IA-IL-MO	6/17/2028 817 - 818 / 862 - 863
WPOI392	NEXTEL WEST CORP.	YC	BEA100	В	Des Moines, IA-IL-MO	6/17/2028 817 - 818 / 862 - 863
WQPK770	NEXTEL WEST CORP.	YC	BEA100	В	Des Moines, IA-IL-MO	6/17/2028 817 - 818 / 862 - 863
WPOH229	NEXTEL WEST CORP.	ΥH	BEA100	Χ	Des Moines, IA-IL-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI393	NEXTEL WEST CORP.	ΥH	BEA100	Χ	Des Moines, IA-IL-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH230	NEXTEL WEST CORP.	YC	BEA101	В	Peoria-Pekin, IL	6/17/2028 817 - 818 / 862 - 863
WPOH231	NEXTEL WEST CORP.	ΥH	BEA101	Χ	Peoria-Pekin, IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH233	NEXTEL WEST CORP.	YC	BEA102	В	Davenport-Moline-Rock Island,	6/17/2028 817 - 818 / 862 - 863
WQPK773	NEXTEL WEST CORP.	YC	BEA102	В	Davenport-Moline-Rock Island,	6/17/2028 817 - 818 / 862 - 863
WPOH234	NEXTEL WEST CORP.	ΥH	BEA102	Χ	Davenport-Moline-Rock Island,	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH236	NEXTEL WEST CORP.	YC	BEA103	В	Cedar Rapids, IA	6/17/2028 817 - 818 / 862 - 863
WPOH237	NEXTEL WEST CORP.	ΥH	BEA103	Χ	Cedar Rapids, IA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH239	NEXTEL WEST CORP.	YC	BEA104	В	Madison, WI-IA-IL	6/17/2028 817 - 818 / 862 - 863
WPOI395	NEXTEL WEST CORP.	YC	BEA104	В	Madison, WI-IA-IL	6/17/2028 817 - 818 / 862 - 863
WQPK776	NEXTEL WEST CORP.	YC	BEA104	В	Madison, WI-IA-IL	6/17/2028 817 - 818 / 862 - 863
WQPK821	NEXTEL WEST CORP.	YC	BEA104	В	Madison, WI-IA-IL	6/17/2028 817 - 818 / 862 - 863
WPOH240	NEXTEL WEST CORP.	ΥH	BEA104	Χ	Madison, WI-IA-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI396	NEXTEL WEST CORP.	ΥH	BEA104	Χ	Madison, WI-IA-IL	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH242	NEXTEL WEST CORP.	YC	BEA105	В	La Crosse, WI-MN	6/17/2028 817 - 818 / 862 - 863
WPOH243	NEXTEL WEST CORP.	ΥH	BEA105	Χ	La Crosse, WI-MN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH245	NEXTEL WEST CORP.	YC	BEA106	В	Rochester, MN-IA-WI	6/17/2028 817 - 818 / 862 - 863
WPOI398	NEXTEL WEST CORP.	YC	BEA106	В	Rochester, MN-IA-WI	6/17/2028 817 - 818 / 862 - 863
WPQT224	NEXTEL WEST CORP.	YC	BEA106	В	Rochester, MN-IA-WI	6/17/2028 817 - 818 / 862 - 863
WPOH246	NEXTEL WEST CORP.	ΥH	BEA106	Χ	Rochester, MN-IA-WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI399	NEXTEL WEST CORP.	ΥH	BEA106	Χ	Rochester, MN-IA-WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT225	NEXTEL WEST CORP.	ΥH	BEA106	Χ	Rochester, MN-IA-WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH248	NEXTEL WEST CORP.	YC	BEA107	В	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 817 - 818 / 862 - 863
WPOI363	NEXTEL WEST CORP.	YC	BEA107	В	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 817 - 818 / 862 - 863
WPQT227	NEXTEL WEST CORP.	YC	BEA107	В	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 817 - 818 / 862 - 863
WPOH249	NEXTEL WEST CORP.	ΥH	BEA107	Χ	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI364	NEXTEL WEST CORP.	ΥH	BEA107	Χ	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT228	NEXTEL WEST CORP.	ΥH	BEA107	Χ	Minneapolis-St. Paul, MN-WI-IA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI401	NEXTEL WEST CORP.	YC	BEA108	В	Wausau, WI	6/17/2028 817 - 818 / 862 - 863
WPOI402	NEXTEL WEST CORP.	ΥH	BEA108	Χ	Wausau, WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI404	NEXTEL WEST CORP.	YC	BEA109	В	Duluth-Superior, MN-WI	6/17/2028 817 - 818 / 862 - 863
WPOI405	NEXTEL WEST CORP.	ΥH	BEA109	Χ	Duluth-Superior, MN-WI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH256	NEXTEL WEST CORP.	YC	BEA110	В	Grand Forks, ND-MN	6/17/2028 817 - 818 / 862 - 863
WPOH257	NEXTEL WEST CORP.	ΥH	BEA110	Χ	Grand Forks, ND-MN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH258	NEXTEL WEST CORP.	YC	BEA111	В	Minot, ND	6/17/2028 817 - 818 / 862 - 863
WPOH259	NEXTEL WEST CORP.	ΥH	BEA111	Χ	Minot, ND	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH260	NEXTEL WEST CORP.	YC	BEA112	В	Bismarck, ND-MT-SD	6/17/2028 817 - 818 / 862 - 863
WPOH261	NEXTEL WEST CORP.	ΥH	BEA112	Χ	Bismarck, ND-MT-SD	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPOH262	NEXTEL WEST CORP.	YC	BEA113	В	Fargo-Moorhead, ND-MN	6/17/2028 817 - 818 / 862 - 863
WPOH263	NEXTEL WEST CORP.	ΥH	BEA113	Χ	Fargo-Moorhead, ND-MN	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH264	NEXTEL WEST CORP.	YC	BEA114	В	Aberdeen, SD	6/17/2028 817 - 818 / 862 - 863
WPOH265	NEXTEL WEST CORP.	ΥH	BEA114	Χ	Aberdeen, SD	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH267	NEXTEL WEST CORP.	YC	BEA115	В	Rapid City, SD-MT-ND-NE	6/17/2028 817 - 818 / 862 - 863
WPOH268	NEXTEL WEST CORP.	ΥH	BEA115	Χ	Rapid City, SD-MT-ND-NE	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH270	NEXTEL WEST CORP.	YC	BEA116	В	Sioux Falls, SD-IA-MN-NE	6/17/2028 817 - 818 / 862 - 863
WPOH271	NEXTEL WEST CORP.	YH	BEA116	Χ	Sioux Falls, SD-IA-MN-NE	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH273	NEXTEL WEST CORP.	YC	BEA117	В	Sioux City, IA-NE-SD	6/17/2028 817 - 818 / 862 - 863
WPOH274	NEXTEL WEST CORP.	ΥH	BEA117	Χ	Sioux City, IA-NE-SD	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI407	NEXTEL WEST CORP.	YC	BEA118	В	Omaha, NE-IA-MO	6/17/2028 817 - 818 / 862 - 863
WPOI408	NEXTEL WEST CORP.	ΥH	BEA118	Χ	Omaha, NE-IA-MO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI307	NEXTEL WEST CORP.	YC	BEA119	В	Lincoln, NE	6/17/2028 817 - 818 / 862 - 863
WPOH280	NEXTEL WEST CORP.	ΥH	BEA119	Χ	Lincoln, NE	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH282	NEXTEL WEST CORP.	YC	BEA120	В	Grand Island, NE	6/17/2028 817 - 818 / 862 - 863
WPOH283	NEXTEL WEST CORP.	YH	BEA120	Χ	Grand Island, NE	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH284	NEXTEL WEST CORP.	YC	BEA121	В	North Platte, NE-CO	6/17/2028 817 - 818 / 862 - 863
WPQS994	NEXTEL WEST CORP.	YC	BEA121	В	North Platte, NE-CO	6/17/2028 817 - 818 / 862 - 863
WPOH285	NEXTEL WEST CORP.	YH	BEA121	Χ	North Platte, NE-CO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQS995	NEXTEL WEST CORP.	YH	BEA121	Χ	North Platte, NE-CO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH287	NEXTEL WEST CORP.	YC	BEA122	В	Wichita, KS-OK	6/17/2028 817 - 818 / 862 - 863
WPQT233	NEXTEL WEST CORP.	YC	BEA122	В	Wichita, KS-OK	6/17/2028 817 - 818 / 862 - 863
WPOH288	NEXTEL WEST CORP.	YH	BEA122	Χ	Wichita, KS-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT234	NEXTEL WEST CORP.	YH	BEA122	Χ	Wichita, KS-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH290	NEXTEL WEST CORP.	YC	BEA123	В	Topeka, KS	6/17/2028 817 - 818 / 862 - 863
WPOH291	NEXTEL WEST CORP.	YH	BEA123	Χ	Topeka, KS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH293	NEXTEL WEST CORP.	YC	BEA124	В	Tulsa, OK-KS	6/17/2028 817 - 818 / 862 - 863
WPOH294	NEXTEL WEST CORP.	YH	BEA124	Χ	Tulsa, OK-KS	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH296	NEXTEL WEST CORP.	YC	BEA125	В	Oklahoma City, OK	6/17/2028 817 - 818 / 862 - 863
WPOH297	NEXTEL WEST CORP.	YH	BEA125	Χ	Oklahoma City, OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH298	NEXTEL WEST CORP.	YC	BEA126	В	Western Oklahoma, OK	6/17/2028 817 - 818 / 862 - 863
WPOH299	NEXTEL WEST CORP.	YH	BEA126	Χ	Western Oklahoma, OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH301	NEXTEL WEST CORP.	YC	BEA127	В	Dallas-Fort Worth, TX-AR-OK	6/17/2028 817 - 818 / 862 - 863
WPOI410	NEXTEL WEST CORP.	YC	BEA127	В	Dallas-Fort Worth, TX-AR-OK	6/17/2028 817 - 818 / 862 - 863
WPOH302	NEXTEL WEST CORP.	YH	BEA127	Χ	Dallas-Fort Worth, TX-AR-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI411	NEXTEL WEST CORP.	YH	BEA127	Χ	Dallas-Fort Worth, TX-AR-OK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK472	NEXTEL WEST CORP.	YC	BEA128	В	Abilene, TX	6/17/2028 817 - 818 / 862 - 863
WPOK473	NEXTEL WEST CORP.	YC	BEA128	Χ	Abilene, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH305	NEXTEL WEST CORP.	YC	BEA129	В	San Angelo, TX	6/17/2028 817 - 818 / 862 - 863
WPOH306	NEXTEL WEST CORP.	YC	BEA129	Χ	San Angelo, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH308	NEXTEL WEST CORP.	YC	BEA130	В	Austin-San Marcos, TX	6/17/2028 817 - 818 / 862 - 863
WPOI366	NEXTEL WEST CORP.	YC	BEA130	В	Austin-San Marcos, TX	6/17/2028 817 - 818 / 862 - 863
WPQT222	NEXTEL WEST CORP.	YC	BEA130	В	Austin-San Marcos, TX	6/17/2028 817 - 818 / 862 - 863
WPOH309	NEXTEL WEST CORP.	YH	BEA130	Χ	Austin-San Marcos, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI367	NEXTEL WEST CORP.	YH	BEA130	Χ	Austin-San Marcos, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT238	NEXTEL WEST CORP.	YH	BEA130	Χ	Austin-San Marcos, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPOH311	NEXTEL WEST CORP.	YC	BEA131	В	Houston-Galveston-Brazoria, TX	6/17/2028 817 - 818 / 862 - 863
WPOI369	NEXTEL WEST CORP.	YC	BEA131	В	Houston-Galveston-Brazoria, TX	6/17/2028 817 - 818 / 862 - 863
WPOH312	NEXTEL WEST CORP.	ΥH	BEA131	Χ	Houston-Galveston-Brazoria, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI370	NEXTEL WEST CORP.	ΥH	BEA131	Χ	Houston-Galveston-Brazoria, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH314	NEXTEL WEST CORP.	YC	BEA132	В	Corpus Christi, TX	6/17/2028 817 - 818 / 862 - 863
WPOI310	NEXTEL WEST CORP.	YC	BEA132	В	Corpus Christi, TX	6/17/2028 817 - 818 / 862 - 863
WPOH315	NEXTEL WEST CORP.	YC	BEA132	Χ	Corpus Christi, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI311	NEXTEL WEST CORP.	YC	BEA132	Χ	Corpus Christi, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH317	NEXTEL WEST CORP.	YC	BEA133	В	McAllen-Edinburg-Mission, TX	6/17/2028 817 - 818 / 862 - 863
WPOH318	NEXTEL WEST CORP.	YC	BEA133	Χ	McAllen-Edinburg-Mission, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH320	NEXTEL WEST CORP.	YC	BEA134	В	San Antonio, TX	6/17/2028 817 - 818 / 862 - 863
WPOK475	NEXTEL WEST CORP.	YC	BEA134	В	San Antonio, TX	6/17/2028 817 - 818 / 862 - 863
WPOH321	NEXTEL WEST CORP.	YC	BEA134	Χ	San Antonio, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOK476	NEXTEL WEST CORP.	YC	BEA134	Χ	San Antonio, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH323	NEXTEL WEST CORP.	YC	BEA135	В	Odessa-Midland, TX	6/17/2028 817 - 818 / 862 - 863
WPOH324	NEXTEL WEST CORP.	YC	BEA135	Χ	Odessa-Midland, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH325	NEXTEL WEST CORP.	YC	BEA136	В	Hobbs, NM-TX	6/17/2028 817 - 818 / 862 - 863
WPQS989	NEXTEL WEST CORP.	YC	BEA136	В	Hobbs, NM-TX	6/17/2028 817 - 818 / 862 - 863
WPOH326	NEXTEL WEST CORP.	ΥH	BEA136	Χ	Hobbs, NM-TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQS990	NEXTEL WEST CORP.	ΥH	BEA136	Χ	Hobbs, NM-TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH328	NEXTEL WEST CORP.	YC	BEA137	В	Lubbock, TX	6/17/2028 817 - 818 / 862 - 863
WPOH329	NEXTEL WEST CORP.	ΥH	BEA137	Χ	Lubbock, TX	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH330	NEXTEL WEST CORP.	YC	BEA138	В	Amarillo, TX-NM	6/17/2028 817 - 818 / 862 - 863
WPQT230	NEXTEL WEST CORP.	YC	BEA138	В	Amarillo, TX-NM	6/17/2028 817 - 818 / 862 - 863
WPOH331	NEXTEL WEST CORP.	ΥH	BEA138	Χ	Amarillo, TX-NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT231	NEXTEL WEST CORP.	ΥH	BEA138	Χ	Amarillo, TX-NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH333	NEXTEL WEST CORP.	YC	BEA139	В	Santa Fe, NM	6/17/2028 817 - 818 / 862 - 863
WPOH334	NEXTEL WEST CORP.	ΥH	BEA139	Χ	Santa Fe, NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH336	NEXTEL WEST CORP.	YC	BEA140	В	Pueblo, CO-NM	6/17/2028 817 - 818 / 862 - 863
WPOH337	NEXTEL WEST CORP.	ΥH	BEA140	Χ	Pueblo, CO-NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH339	NEXTEL WEST CORP.	YC	BEA141	В	Denver-Boulder-Greeley, CO-KS-	6/17/2028 817 - 818 / 862 - 863
WPQS992	NEXTEL WEST CORP.	YC	BEA141	В	Denver-Boulder-Greeley, CO-KS-	6/17/2028 817 - 818 / 862 - 863
WPOH340	NEXTEL WEST CORP.	ΥH	BEA141	Χ	Denver-Boulder-Greeley, CO-KS-	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQS993	NEXTEL WEST CORP.	ΥH	BEA141	Χ	Denver-Boulder-Greeley, CO-KS-	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH341	NEXTEL WEST CORP.	YC	BEA142	В	Scottsbluff, NE-WY	6/17/2028 817 - 818 / 862 - 863
WPOH342	NEXTEL WEST CORP.	ΥH	BEA142	Χ	Scottsbluff, NE-WY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH343	NEXTEL WEST CORP.	YC	BEA143	В	Casper, WY-ID-UT	6/17/2028 817 - 818 / 862 - 863
WPOH344	NEXTEL WEST CORP.	ΥH	BEA143	Χ	Casper, WY-ID-UT	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH345	NEXTEL WEST CORP.	YC	BEA144	В	Billings, MT-WY	6/17/2028 817 - 818 / 862 - 863
WPOH346	NEXTEL WEST CORP.	ΥH	BEA144	Χ	Billings, MT-WY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH347	NEXTEL WEST CORP.	YC	BEA145	В	Great Falls, MT	6/17/2028 817 - 818 / 862 - 863
WPOH348	NEXTEL WEST CORP.	ΥH	BEA145	Χ	Great Falls, MT	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH349	NEXTEL WEST CORP.	YC	BEA146	В	Missoula, MT	6/17/2028 817 - 818 / 862 - 863
WPOH350	NEXTEL WEST CORP.	ΥH	BEA146	Χ	Missoula, MT	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH352	NEXTEL WEST CORP.	YC	BEA147	В	Spokane, WA-ID	6/17/2028 817 - 818 / 862 - 863
WPOH353	NEXTEL WEST CORP.	ΥH	BEA147	Χ	Spokane, WA-ID	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869

WPOI371	NEXTEL WEST CORP.	YC	BEA148	В	Idaho Falls, ID-WY	6/17/2028 817 - 818 / 862 - 863
WPOI372	NEXTEL WEST CORP.	YH	BEA148	X	Idaho Falls, ID-WY	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH356	NEXTEL WEST CORP.	YC	BEA149	В	Twin Falls, ID	6/17/2028 817 - 818 / 862 - 863
WPOI312	NEXTEL WEST CORP.	YC	BEA149	В	Twin Falls, ID	6/17/2028 817 - 818 / 862 - 863
	NEXTEL WEST CORP.	YH	BEA149	X	Twin Falls, ID	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI313	NEXTEL WEST CORP.	YH	BEA149	X	Twin Falls, ID	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	NEXTEL WEST CORP.	YC	BEA150	В	Boise City, ID-OR	6/17/2028 817 - 818 / 862 - 863
WPOH358	NEXTEL WEST CORP.	YH	BEA150	X	Boise City, ID-OR	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOI314	NEXTEL WEST CORP.	YH	BEA150	X	Boise City, ID-OR	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPQT207	NEXTEL WEST CORP.	YH	BEA150	X	Boise City, ID-OR	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH359	NEXTEL WEST CORP.	YC	BEA151	В	Reno, NV-CA	6/17/2028 817 - 818 / 862 - 863
	NEXTEL WEST CORP.	YH	BEA151	X	Reno, NV-CA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH362	NEXTEL WEST CORP.	YC	BEA152	В	Salt Lake City-Ogden, UT-ID	6/17/2028 817 - 818 / 862 - 863
WPOH363	NEXTEL WEST CORP.	YH	BEA152	X	Salt Lake City-Ogden, UT-ID	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	NEXTEL WEST CORP.	YC	BEA153	В	Las Vegas, NV-AZ-UT	6/17/2028 817 - 818 / 862 - 863
WQPK846	NEXTEL WEST CORP.	YC	BEA153	В	Las Vegas, NV-AZ-UT	6/17/2028 817 - 818 / 862 - 863
	NEXTEL WEST CORP.	YH	BEA153	X	Las Vegas, NV-AZ-UT	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	NEXTEL WEST CORP.	YC	BEA154	В	Flagstaff, AZ-UT	6/17/2028 817 - 818 / 862 - 863
	NEXTEL WEST CORP.	ΥH	BEA154	Χ	Flagstaff, AZ-UT	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
	NEXTEL WEST CORP.	YC	BEA155	В	Farmington, NM-CO	6/17/2028 817 - 818 / 862 - 863
	NEXTEL WEST CORP.	ΥH	BEA155	Χ	Farmington, NM-CO	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH372	NEXTEL WEST CORP.	YC	BEA156	В	Albuquerque, NM-AZ	6/17/2028 817 - 818 / 862 - 863
WPOH373	NEXTEL WEST CORP.	ΥH	BEA156	Χ	Albuquerque, NM-AZ	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH375	NEXTEL WEST CORP.	YC	BEA157	В	El Paso, TX-NM	6/17/2028 817 - 818 / 862 - 863
WPOH376	NEXTEL WEST CORP.	YC	BEA157	Χ	El Paso, TX-NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH378	NEXTEL WEST CORP.	YC	BEA158	В	Phoenix-Mesa, AZ-NM	6/17/2028 817 - 818 / 862 - 863
WPOH379	NEXTEL WEST CORP.	YC	BEA158	Χ	Phoenix-Mesa, AZ-NM	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH381	NEXTEL WEST CORP.	YC	BEA159	В	Tucson, AZ	6/17/2028 817 - 818 / 862 - 863
WPOH382	NEXTEL WEST CORP.	YC	BEA159	Χ	Tucson, AZ	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH384	NEXTEL WEST CORP.	YC	BEA160	В	Los Angeles-Riverside-Orange C	6/17/2028 817 - 818 / 862 - 863
WPOH385	NEXTEL WEST CORP.	YC	BEA160	Χ	Los Angeles-Riverside-Orange C	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH387	NEXTEL WEST CORP.	YC	BEA161	В	San Diego, CA	6/17/2028 817 - 818 / 862 - 863
WPOH388	NEXTEL WEST CORP.	YC	BEA161	Χ	San Diego, CA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH390	NEXTEL WEST CORP.	YC	BEA162	В	Fresno, CA	6/17/2028 817 - 818 / 862 - 863
WPOH391	NEXTEL WEST CORP.	ΥH	BEA162	Χ	Fresno, CA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH393	NEXTEL WEST CORP.	YC	BEA163	В	San Francisco-Oakland-San Jose	6/17/2028 817 - 818 / 862 - 863
WPOH394	NEXTEL WEST CORP.	ΥH	BEA163	Χ	San Francisco-Oakland-San Jose	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH396	NEXTEL WEST CORP.	YC	BEA164	В	Sacramento-Yolo, CA	6/17/2028 817 - 818 / 862 - 863
WPOH397	NEXTEL WEST CORP.	ΥH	BEA164	Χ	Sacramento-Yolo, CA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH398	NEXTEL WEST CORP.	YC	BEA165	В	Redding, CA-OR	6/17/2028 817 - 818 / 862 - 863
WPOH399	NEXTEL WEST CORP.	ΥH	BEA165	Χ	Redding, CA-OR	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH400	NEXTEL WEST CORP.	YC	BEA166	В	Eugene-Springfield, OR-CA	6/17/2028 817 - 818 / 862 - 863
WPOH401	NEXTEL WEST CORP.	ΥH	BEA166	Χ	Eugene-Springfield, OR-CA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH403	NEXTEL WEST CORP.	YC	BEA167	В	Portland-Salem, OR-WA	6/17/2028 817 - 818 / 862 - 863
WPOH404	NEXTEL WEST CORP.	YH	BEA167	X	Portland-Salem, OR-WA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH433	NEXTEL WEST CORP.	YC	BEA168	В	Pendleton, OR-WA	6/17/2028 817 - 818 / 862 - 863

Case 1:19-cv-02232 Document 2-2 Filed 07/26/19 Page 62 of 64

WPOH405	NEXTEL WEST CORP.	ΥH	BEA168	Χ	Pendleton, OR-WA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH406	NEXTEL WEST CORP.	YC	BEA169	В	Richland-Kennewick-Pasco, WA	6/17/2028 817 - 818 / 862 - 863
WPOH407	NEXTEL WEST CORP.	ΥH	BEA169	Χ	Richland-Kennewick-Pasco, WA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH409	NEXTEL WEST CORP.	YC	BEA170	В	Seattle-Tacoma-Bremerton, WA	6/17/2028 817 - 818 / 862 - 863
WPOH410	NEXTEL WEST CORP.	ΥH	BEA170	Χ	Seattle-Tacoma-Bremerton, WA	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPOH412	NEXTEL WEST CORP.	YC	BEA171	В	Anchorage, AK	6/17/2028 817 - 818 / 862 - 863
WPOH413	NEXTEL WEST CORP.	YH	BEA171	Χ	Anchorage, AK	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPLM204	NEXTEL WEST CORP.	YC	BEA172	В	Honolulu, HI	3/10/2028 817 - 818 / 862 - 863
WPOH414	NEXTEL WEST CORP.	YH	BEA172	Χ	Honolulu, HI	6/17/2028 818 - 821 / 863 - 866 / 821 - 824 / 866 - 869
WPRQ760	Nextel of Puerto Rico, Inc.	YC	BEA174	Χ	Puerto Rico & Virgin Isl.	12/20/2020 - / - / 821.575 - 824 / 866.575 - 869

Number of

800 MHz Site-based License Holdings next to Canadian Border area designed to use contours to preserve capability inside Canadian Border area and preclude expansion from existing licensees inside and outside the Canadian Border area

						Discreet	
		Radio				Frequencies	
		Service				Above 862.0	Expiration
Call Sign	Licensee	Code	Transmitter City	Transmitter County	State	MHz	Date
WQXZ925	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC,	YM	SOUTH PORTLAND	CUMBERLAND	ME	160	7/26/2026
11/01/7020	INC.	\/h.4	DDECDEN	A 41 (S) ((A) (S) (A) 4	011	160	7/26/2026
WQXZ929	NEXTEL WEST CORP.	YM	DRESDEN	MUSKINGUM	OH	160	7/26/2026
WQXZ930	NEXTEL WEST CORP.	YM	UPPER ARLINGTON	FRANKLIN	OH	160	7/26/2026
WQYF361	NEXTEL WEST CORP.	YM	ORISKANY	ONEIDA	NY	160	9/8/2026
WQYG555	NEXTEL WEST CORP.	YM	KALAMA	COWLITZ	WA	160	9/21/2026
WQYG556	NEXTEL WEST CORP.	YM	OUTLOOK	YAKIMA	WA	160	9/21/2026
WQYG557	NEXTEL WEST CORP.	YM	EPHRATA	GRANT	WA	160	9/21/2026
WQYT997	NEXTEL WEST CORP.	YM	LEONIDAS	ST. JOSEPH	MI	160	1/19/2027
WQYU236	NEXTEL WEST CORP.	YM	FOWLER	CLINTON	MI	160	1/20/2027
WQYV719	NEXTEL WEST CORP.	YM	COTTON	ST. LOUIS	MN	160	2/6/2027
WQYV720	NEXTEL WEST CORP.	YM	HIBBING	ST. LOUIS	MN	160	2/6/2027
WQYV722	NEXTEL WEST CORP.	YM	BEMIDJI	BELTRAMI	MN	160	2/6/2027
WQZS657	NEXTEL WEST CORP.	YM	CAMPBELL	STEUBEN	NY	160	7/24/2027
WQZS659	NEXTEL WEST CORP.	YM	SPENCERVILLE	ALLEN	IN	160	7/24/2027
WQZS660	NEXTEL WEST CORP.	YM	JACKSON CENTER	SHELBY	ОН	160	7/24/2027
WQZS661	NEXTEL WEST CORP.	YM	EAST PALESTINE	COLUMBIANA	ОН	160	7/24/2027
WQZS662	NEXTEL WEST CORP.	YM	UHRICHSVILLE	TUSCARAWAS	ОН	160	7/24/2027
WRAA475	NEXTEL WEST CORP.	YM	CONVOY	VAN WERT	ОН	160	9/22/2027
WRAN278	NEXTEL WEST CORP.	YM	TRAVERSE CITY	GRAND TRAVERSE	MI	160	1/11/2028
WRBR754	NEXTEL WEST CORP.	YM	HOOD RIVER	HOOD RIVER	OR	160	5/31/2028
WRBV827	NEXTEL WEST CORP.	YM	HIGGINGS LAKE	ROSCOMMON	MI	160	7/3/2028
WRBV828	NEXTEL WEST CORP.	YM	GREAT FALLS	CASCADE	MT	160	7/3/2028
WRBV829	NEXTEL WEST CORP.	YM	DEER PARK	SPOKANE	WA	160	7/3/2028
WRBV830	NEXTEL WEST CORP.	YM	MT. PLEASANT	ISABELLA	MI	160	7/3/2028
WRBV831	NEXTEL WEST CORP.	YM	COLEMAN	MIDLAND	MI	160	7/3/2028
WRBV855	NEXTEL WEST CORP.	YM	CORTLAND	CORTLAND	NY	160	7/3/2028
WRBV857	NEXTEL WEST CORP.	YM	HOOD RIVER	HOOD RIVER	OR	160	7/3/2028
WRBV861	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC,	YM	SOUTH ROYALTON	WINDSOR	VT	160	7/3/2028
	INC.						
WRBV862	NEXTEL WEST CORP.	YM	GLOVERSVILLE	FULTON	NY	160	7/3/2028
WRBV864	NEXTEL WEST CORP.	YM	RIDGEWAY	ELK	PA	160	7/3/2028
WRBV870	NEXTEL WEST CORP.	YM	RIDGEWAY	ELK	PA	160	7/3/2028
WRBV873	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC,	YM	BIDDEFORD	YORK	ME	160	7/3/2028
	INC.						

Case 1:19-cv-02232 Document 2-2 Filed 07/26/19 Page 64 of 64

WRBV875	NEXTEL WEST CORP.	YM	NEW HAVEN	ADDISON	VT	160	7/3/2028
WRBV876	NEXTEL WEST CORP.	YM	SCHROON LAKE	ESSEX	NY	160	7/3/2028
WRBX774	NEXTEL WEST CORP.	YM	ATHOL	BONNER	ID	160	7/19/2028
WRBX833	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC,	YM	ELLSWORTH	HANCOCK	ME	155	7/19/2028
	INC.						
WRCA788	NEXTEL WEST CORP.	YM	HARRISON TOWNSHIP	LICKING	OH	160	8/13/2028
WRCA793	NEXTEL WEST CORP.	YM	MALVERN	CARROLL	OH	160	8/13/2028
WRCA794	NEXTEL COMMUNICATIONS OF THE MID ATLANTIC,	YM	CLINTONVILLE	VENANGO	PA	160	8/13/2028
	INC.						
WRCL354	NEXTEL WEST CORP.	YM	LAKE ODESSA	BARRY	MI	160	10/30/2028

EXHIBIT 2

STPULATION & ORDER

UNITED STATES DISTRICT COURT FOR THE DISTRICT OF COLUMBIA

UNITED STATES OF AMERICA et al.,

Plaintiffs,

Case No.

v.

DEUTSCHE TELEKOM AG, T-MOBILE US, INC., SOFTBANK GROUP CORP., and SPRINT CORPORATION,

Filed:

Defendants.

STIPULATION AND ORDER

It is hereby stipulated and agreed by and between the undersigned parties, subject to approval and entry by the Court, that:

I. <u>DEFINITIONS</u>

As used in this Stipulation and Order:

- A. "Assurance Wireless" means the prepaid wireless business conducted by Virgin Mobile under the Assurance Lifeline brand.
- B. "Cell Site" means any wireless communications towers, rooftops, water towers, or other wireless communications facilities owned or leased by T-Mobile or Sprint and the physical location and wireless equipment thereto.
- C. "Cell Site and Retail Assets" means all Cell Sites and Retail Locations that

 T-Mobile and Sprint Decommission within five (5) years of the divestiture of the Prepaid Assets.
 - D. "Decommissioned" or "Decommissioning" means, with respect to a Cell Site,

when the Cell Site is no longer transmitting on T-Mobile's or Sprint's network. With respect to Retail Locations, Decommissioned or Decommissioning means when T-Mobile or Sprint cease customer service operations.

- E. "Deutsche Telekom" means Deutsche Telekom AG, a German corporation headquartered in Bonn, Germany, that is the controlling shareholder of T-Mobile; its successors and assigns; and its parents, subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- F. "DISH" means DISH Network Corporation, a Nevada corporation with its headquarters in Englewood, Colorado; its successors and assigns; and its parents, subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- G. "Divestiture Assets" means the Prepaid Assets, the Spectrum Assets, and the Cell Site and Retail Assets.
- H. "Prepaid Assets" means all tangible and intangible assets primarily used by the Boost Mobile, Sprint-branded prepaid, and Virgin Mobile businesses today, including but not limited to Boost and Virgin Mobile Retail Locations, licenses, personnel, facilities, data, and intellectual property, as well as all relationships and/or contracts with prepaid customers served by Sprint, Boost Mobile, and Virgin Mobile. Prepaid Assets do not include the Assurance Wireless business and the prepaid wireless customers of Shenandoah Telecommunications Company and Swiftel Communications, Inc.
- I. "Prepaid Assets Personnel" means all employees whose jobs currently focus on the support of the Prepaid Assets, or whose jobs have previously focused on supporting the

Prepaid Assets at any time between January 1, 2016 and the date on which the Prepaid Assets are divested to DISH. Prepaid Assets Personnel shall include no fewer than 400 current employees of Sprint, which shall include employees involved in sales management, marketing management, distribution support, sales support, and finance.

- J. "Retail Locations" means any retail locations owned or operated by T-Mobile or Sprint and from which either T-Mobile or Sprint sells mobile wireless services under any of their affiliated brands, including Sprint, Boost Mobile, Virgin Mobile, T-Mobile, Metro by T-Mobile, and MetroPCS.
- K. "SoftBank" means SoftBank Group Corp., a Japanese corporation and controlling shareholder of Sprint; its successors and assigns; and its parents, subsidiaries, divisions, groups, affiliates, partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- L. "Spectrum Assets" means all of Sprint's 800 MHz spectrum holdings as listed and described in Attachment A to the proposed Final Judgment.
- M. "Sprint" means Defendant Sprint Corporation, a Delaware corporation with its headquarters in Overland Park, Kansas; its successors and assigns; and its subsidiaries, divisions, groups, affiliates (other than SoftBank), partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.
- N. "T-Mobile" means Defendant T-Mobile US, Inc., a Delaware corporation with its headquarters in Bellevue, Washington; its successors and assigns; and its subsidiaries, divisions, groups, affiliates (other than Deutsche Telekom), partnerships, and joint ventures, and their directors, officers, managers, agents, and employees.

II. OBJECTIVES

- A. The proposed Final Judgment filed in this case is meant to ensure (i) T-Mobile's and Sprint's prompt divestiture of the Prepaid Assets, and (ii) the ability of DISH to acquire the Spectrum Assets and exercise its option to Cell Site and Retail Assets for the purpose of establishing a viable competitor in the provision of retail mobile wireless service in order to address the effects that the United States alleges would likely result from the merger of T-Mobile and Sprint. Under the proposed Final Judgment, DISH would acquire the Prepaid Assets and Spectrum Assets from Sprint or T-Mobile and have the ability to exercise its option to acquire the Cell Site and Retail Assets.
- B. The United States has determined, for the purposes of this settlement, that DISH is an acceptable purchaser of the Divestiture Assets. Unless the United States otherwise consents in writing or DISH declines its option to purchase certain Decommissioned Cell Sites or Decommissioned Retail Locations, the divestitures pursuant to the Final Judgment will include the entire Divestiture Assets. If approved by the Court, the proposed Final Judgment would resolve the United States' claims in this antitrust lawsuit, which sought to enjoin the merger of Sprint and T-Mobile.
- C. Central to the relief offered by the proposed Final Judgment is the requirement that DISH undertake certain actions to effectuate the complete transfer of the Divestiture Assets to achieve the competitive objectives of the proposed Final Judgment. Among other actions, DISH must: (i) offer nationwide postpaid retail mobile wireless service to American consumers within one (1) year of the closing of the sale of the Prepaid Assets; (ii) comply with network build commitments; (iii) submit to the Department of Justice regular periodic updates on the

status of its network deployment; and (iv) implement strict firewall procedures to prevent the exchange of competitively sensitive information through the ongoing relationships between DISH and T-Mobile or Sprint necessitated by the divestitures. This relief cannot be effectively accomplished without DISH's participation, and DISH has agreed to be bound as a partydefendant by the Final Judgment. DISH shares an interest in the resolution of this litigation and shares common issues of law and fact with Sprint and T-Mobile such that joinder is proper under Rule 20(a) of the Federal Rules of Civil Procedure and under Section 15 of the Clayton Act, 15 U.S.C. § 25. See United States v. Bayer AG, No. 1:18-cv-01241, Doc. No. 11 (D.D.C. May 29, 2018) (entering stipulation adding divestiture buyer as a defendant); United States v. Pabst Brewing Co., 183 F. Supp. 220, 221 (E.D. Wis. 1960) (holding that "[i]n a proceeding under § 7 of the Clayton Act, the court has authority to grant relief not only against parties who are found to have violated that section, but also against other parties if such relief is necessary to eliminate the effects of an acquisition offensive to the statute"); see also United States v. Phillips Petroleum Co., 367 F. Supp. 1226, 1261-62 (C.D. Cal. 1973) (denying dismissal to a defendant as a properly joined party "against whom relief may be granted").

D. This Stipulation and Order ensures that the relief afforded in the proposed Final Judgment will be effective by: (i) adding DISH as a Defendant in this action for purposes of settlement and for entry and enforcement of the proposed Final Judgment; (ii) ensuring, prior to the completion of all of the proposed divestitures, that the Divestiture Assets remain economically viable and ongoing business concerns; and (iii) ensuring that T-Mobile, Sprint, and DISH will be bound by the terms of the proposed Final Judgment during the settlement approval

process that will occur under the Antitrust Procedures and Penalties Act ("APPA"), 15 U.S.C. § 16(b)-(h).

III. JURISDICTION AND VENUE

The Court has jurisdiction over the subject matter of this action and over each of the parties hereto with respect to this action. The Complaint states a claim upon which relief may be granted against T-Mobile, Sprint, Deutsche Telekom, and SoftBank under Section 7 of the Clayton Act, as amended, 15 U.S.C. § 18. Pursuant to Section V of this Stipulation and Order filed simultaneously with the proposed Final Judgment, DISH has consented to this Court's exercise of specific personal jurisdiction over it in this matter solely for the purposes of settlement and for the entry and enforcement of the proposed Final Judgment. Venue of this action is proper in the United States District Court for the District of Columbia. Defendants waive service of summons of the Complaint.

IV. COMPLIANCE WITH AND ENTRY OF FINAL JUDGMENT

A. The parties stipulate that a Final Judgment in the form attached hereto as Exhibit A may be filed with and entered by the Court, upon the motion of any party or upon the Court's own motion, at any time after compliance with the requirements of the APPA, and without further notice to any party or other proceedings, provided that the United States has not withdrawn its consent, which it may do at any time before the entry of the proposed Final Judgment by serving notice thereof on T-Mobile and Sprint and by filing that notice with the Court. T-Mobile and Sprint agree to arrange, at their expense, publication as quickly as possible of the newspaper notice required by the APPA, which shall be drafted by the United States in its sole discretion. The publication shall be arranged no later than three (3) business days after T-

Mobile and Sprint's receipt from the United States of the text of the notice and the identity of the newspapers within which the publication shall be made. T-Mobile and Sprint shall promptly send to the United States (i) confirmation that publication of the newspaper notices has been arranged, and (ii) certifications of publication prepared by the newspapers within which the notices were published.

- B. Defendants shall abide by and comply with the provisions of the proposed Final Judgment pending the Judgment's entry by the Court, or until expiration of time for all appeals of any Court ruling declining entry of the proposed Final Judgment, and shall, from the date of the signing of this Stipulation by the parties, comply with all the terms and provisions of the proposed Final Judgment. The United States shall have the full rights and enforcement powers in the proposed Final Judgment, including Sections XIV and XVIII, as though the same were in full force and effect as the Final order of the Court.
- C. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall not consummate the transaction sought to be enjoined by the Complaint herein before the Court has signed this Stipulation and Order.
- D. This Stipulation and Order shall apply with equal force and effect to any amended proposed Final Judgment agreed upon in writing by the parties and submitted to the Court.
- E. In the event (i) the United States has withdrawn its consent, as provided in Paragraph IV(A) above, or (ii) the proposed Final Judgment is not entered pursuant to this Stipulation and Order, the time has expired for all appeals of any Court ruling declining entry of the proposed Final Judgment, and the Court has not otherwise ordered continued compliance with the terms and provisions of the proposed Final Judgment, then the parties are released from

all further obligations under this Stipulation and Order, and the making of this Stipulation and Order shall be without prejudice to any party in this or any other proceeding.

F. Defendants represent that the divestitures and other relief ordered in the proposed Final Judgment can and will be completed, and that Defendants will not later raise any claim of mistake, hardship or difficulty of compliance as grounds for asking the Court to modify any of the provisions contained therein.

V. <u>JOINDER OF DISH AS A DEFENDANT</u>

It is hereby stipulated and agreed by and between Plaintiff and Defendants that, upon approval and entry by the Court, DISH be added as a Defendant in this action for purposes of settlement and for entry and enforcement of the proposed Final Judgment.

VI. PRESERVATION OF THE PREPAID ASSETS

Until the divestitures of the Prepaid Assets required by the proposed Final Judgment have been accomplished:

- A. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall preserve, maintain, and, pursuant to Paragraph VI(I) below, operate the Prepaid Assets as independent, ongoing, economically viable competitive businesses. Defendants shall not coordinate their production, marketing, or terms of sale of any products with those produced by or sold under any of the Prepaid Assets.
- B. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall take all steps necessary to ensure that (i) the Prepaid Assets will be maintained and operated as an ongoing, economically viable and active competitor in the retail mobile wireless service business; (ii) management of the Prepaid Assets will not be influenced by T-Mobile, Sprint, Deutsche Telekom, and SoftBank;

and (iii) competitively sensitive sales, marketing and pricing information, and decision-making concerning production, distribution, of sales of products by or under any of the Prepaid Assets will be kept separate and apart from T-Mobile's, Sprint's, Deutsche Telekom's, and SoftBank's other operations.

- C. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall use all reasonable efforts to maintain and increase the sales and revenues of the products and services sold using the Prepaid Assets and shall maintain at 2018 levels or previously approved levels for 2019, whichever are higher, all promotional, advertising, sales, technical assistance, marketing, and merchandising support for the Prepaid Assets. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall also ensure that all plans and efforts to improve current products sold by Sprint using the Prepaid Assets are continued.
- D. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall provide sufficient working capital and lines and sources of credit to continue to maintain the Prepaid Assets as economically viable and competitive, ongoing businesses, consistent with the requirements of Paragraph VI(A) and Paragraph VI(B) of this Stipulation and Order.
- E. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall take all steps necessary to ensure that the Prepaid Assets are fully maintained in operable condition at no less than current capacity and sales, and shall maintain and adhere to normal repair and maintenance schedules for the Prepaid Assets.
- F. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall not, except (i) as part of a divestiture approved by the United States in accordance with the terms of the proposed Final Judgment or (ii) in the ordinary course of business consistent with past practice (e.g., the

ordinary course sale of inventory), remove, sell, lease, assign, transfer, pledge, or otherwise dispose of any of the Prepaid Assets.

- G. To the extent consistent with past practices, T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall maintain, in accordance with sound accounting principles, accurate, and complete financial ledgers, books, and records that report on a periodic basis, such as the last business day of every month, the assets, liabilities, expenses, revenues, and income of the Prepaid Assets.
- H. Defendants shall take no action that would jeopardize, delay, or impede the sale of the Prepaid Assets.
- I. The Prepaid Assets Personnel shall not be transferred or reassigned to other areas within the company except for transfer bids initiated by Prepaid Assets Personnel pursuant to Sprint's regular, established job posting policy. T-Mobile and Sprint shall provide the United States with ten (10) business days' advance notice of such transfer.
- J. Subject to the approval of the United States, T-Mobile and Sprint shall appoint a person or persons to oversee the Prepaid Assets, and who will be responsible for T-Mobile and Sprint's compliance with this Section. This person or persons shall have complete managerial responsibility for the Prepaid Assets, subject to the provisions of the proposed Final Judgment, and shall make all business decisions relating to the operations of the Prepaid Assets, including all production, sale, pricing, and discounting decisions, independent of T-Mobile, Deutsche Telekom, Sprint, and SoftBank. In the event any such person or persons is unable to perform his or her duties, T-Mobile and Sprint shall appoint, subject to the approval of the United States, a replacement within ten (10) business days. Should T-Mobile and Sprint fail to appoint a

replacement acceptable to the United States within this time period, the United States shall appoint a replacement.

VII. PRESERVATION OF THE SPECTRUM ASSETS

Until the divestitures of the Spectrum Assets required by the proposed Final Judgment have been accomplished:

- A. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall preserve, maintain, and continue to operate the Spectrum Assets.
- B. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall not, except as part of a divestiture approved by the United States in accordance with the terms of the proposed Final Judgment, remove, sell, lease, assign, transfer, pledge, or otherwise dispose of any of the Spectrum Assets.
- C. Defendants shall take no action that would jeopardize, delay, or impede the transfer of the Spectrum Assets.

VIII. PRESERVATION OF THE CELL SITE AND RETAIL ASSETS

Until the divestitures of the Cell Site and Retail Assets required by the proposed Final Judgment have been accomplished:

- A. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall preserve, maintain, and continue to operate the Cell Site and Retail Assets.
- D. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall take all steps necessary to ensure that the Cell Site and Retail Assets are fully maintained in operable condition.
- E. T-Mobile, Sprint, Deutsche Telekom, and SoftBank shall not, except as part of a divestiture approved by the United States in accordance with the terms of the proposed Final

Judgment, remove, sell, lease, assign, transfer, pledge, or otherwise dispose of any of the Cell Site and Retail Assets.

G. Defendants shall take no action that would jeopardize, delay, or impede the sale of the Cell Site and Retail Assets.

IX. <u>DURATION OF OBLIGATIONS</u>

- A. The obligations of Defendants under Section VI, VII, and VIII of this Stipulation and Order shall remain in effect until (i) consummation of the divestitures required by the proposed Final Judgment or (ii) until further order of the Court. If the United States voluntarily dismisses the Complaint in this matter, Defendants are released from all further obligations under this Stipulation and Order.
- B. Notwithstanding anything herein to the contrary, SoftBank's obligations under Sections VI, VII, and VIII of this Stipulation and Order shall cease as of the completion of the merger of T-Mobile and Sprint, provided that, in no event shall SoftBank delay or impede Sprint, T-Mobile, or Deutsche Telekom from complying with their respective obligations under this Stipulation and Order.

Dated: 7/26/19

Respectfully submitted,

FOR PLAINTIFF

UNITED STATES OF AMERICA:

Frederick S. Young

U.S. Department of Justice

Antitrust Division

450 Fifth Street NW, Suite 7000

Washington, DC 20530

Tel.: (202) 514-5621 Fax: (202) 514-6381

Email: Federick. Young@usdoj.gov

FOR DEFENDANT

T-MOBILE / DEUTSCHE TELEKOM

Mark W. Nelson (D.C. Bar #442461)

Cleary Gottlieb Steen & Hamilton LLP

2112 Pennsylvania Avenue NW Washington, DC 20037

Tel.: (202) 974-1622

Fax: (202) 974-1999 Email: mnelson@cgsh.com

FOR DEFENDANT SOFTBANK:

David L. Meyer (D.C. Bar #414420)

Morrison & Foerster LLP

2000 Pennsylvania Avenue NW, Suite 6000

Washington, DC 20006-1888

Tel.: (202) 887-1888 Fax: (202) 887-0763 Email: dmeyer@mofo.com

FOR DEFENDANT SPRINT:

Steven C. Sunshine (D.C. Bar #450078)

Skadden, Arps, Slate, Meagher & Flom LLP

1440 New York Avenue NW

Washington, DC 20005 Tel.: (202) 371-7860

Fax: (202) 661-0560

Email: steve.sunshine@skadden.com

FOR DEFENDANT DISH:

The way

Steven L. Holley (*Pro Hac Vice* Pending) Bradley P. Smith (D.C. Bar #468060)

Sullivan & Cromwell LLP

125 Broad Street

New York, New York 10004

Tel.: (212) 558-4000 Fax: (212) 558-3588

Email: holleys@sullcrom.com; smithbr@sullcrom.com

Pantelis Michalopoulos (D.C. Bar #453179)

Madrager,

Andrew Golodny

(D.C. Bar #998151, Pro Hac Vice Pending)

Steptoe & Johnson LLP

1330 Connecticut Avenue NW

Washington, DC 20036 Tel.: (202) 429-6494

Fax: (202) 429-3902

Email: pmichalopoulos@steptoe.com;

agolodny@steptoe.com

ORD	<u>ER</u>	
IT IS SO ORDERED by the Court, this	, 2019.	
	United States District Judge	

EXHIBIT 3

T-MOBILE USA FORM 8-K

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): July 26, 2019



(Exact Name of Registrant as Specified in Charter)

DELAWARE

(State or other jurisdiction of incorporation)

1-33409

(Commission File Number) 20-0836269

(IRS Employer Identification No.)

12920 SE 38th Street
Bellevue, Washington
(Address of principal executive offices)

98006-1350 (Zip Code)

Registrant's telephone number, including area code: (425) 378-4000

(Former Name or Former Address, if Changed Since Last Report):

	<u>-</u>			
	ck the appropriate box below if the Form 8-K filing is intenowing provisions (see General Instructions A.2. below):	ded to simultaneously satisfy the fil	ing obligation of the Registrant under any of the	
\boxtimes	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Sec	urities registered pursuant to Section 12(b) of the Act:			
	Title of each class	Trading symbol	Name of each exchange on which registered	
C	ommon Stock, \$0.00001 par value per share	TMUS	The NASDAQ Stock Market LLC	
	cate by check mark whether the registrant is an emerging groter) or Rule 12b-2 of the Securities Exchange Act of 1934	* *	05 of the Securities Act of 1933 (§230.405 of this	
Eme	erging growth company			
	n emerging growth company, indicate by check mark if the seed financial accounting standards provided pursuant to Sec	_	extended transition period for complying with any new o	

Item 1.01 Entry into a Material Definitive Agreement.

Asset Purchase Agreement

On July 26, 2019, T-Mobile US, Inc. ("T-Mobile") entered into an Asset Purchase Agreement (the "Asset Purchase Agreement") with Sprint Corporation ("Sprint") and DISH Network Corporation ("DISH"). T-Mobile and Sprint are collectively referred to as the "Sellers." Pursuant to the Asset Purchase Agreement, upon the terms and subject to the conditions thereof, following the consummation of the Merger Transactions (as defined below), DISH will acquire Sprint's prepaid wireless business, currently operated under the Boost Mobile, Virgin Mobile and Sprint prepaid brands (excluding the Assurance brand Lifeline customers and the prepaid wireless customers of Shenandoah Telecommunications Company and Swiftel Communications, Inc.), including customer accounts, inventory, contracts, intellectual property and certain other specified assets (the "Prepaid Business"), and will assume certain related liabilities (the "Prepaid Transaction"). DISH will pay the Sellers \$1.4 billion for the Prepaid Business, subject to a working capital adjustment.

The Asset Purchase Agreement has been approved by the Boards of Directors of each of DISH, T-Mobile and Sprint. In addition, each of Deutsche Telekom AG ("Deutsche Telekom") and SoftBank Group Corp. ("SoftBank") have consented to the entry into the Asset Purchase Agreement.

The Sellers have made customary representations and warranties and have agreed to customary covenants regarding the operation of the Prepaid Business between the execution of the Asset Purchase Agreement and the closing of the Prepaid Transaction, including a covenant to conduct the Prepaid Business in the ordinary course during the period prior to closing, subject to certain exceptions.

The consummation of the Prepaid Transaction is subject to the consummation of the Merger Transactions and other customary closing conditions, including, among others: (i) the absence of a breach of each party's representations and warranties, subject to agreed materiality standards, (ii) the absence of a material adverse effect on the Prepaid Business, (iii) the performance of certain covenants in all material respects, and (iv) the ability of T-Mobile to activate customers of the Prepaid Business on the T-Mobile network pursuant to a Master Network Services Agreement between T-Mobile and DISH that will be executed upon the closing of the Prepaid Transaction.

The Asset Purchase Agreement contains certain termination rights for the Sellers and DISH. The Asset Purchase Agreement also provides that either party may specifically enforce the other party's obligations under the Asset Purchase Agreement. In addition to the foregoing termination rights, and subject to certain limitations, DISH may terminate the Asset Purchase Agreement if the Prepaid Transaction is not consummated by July 26, 2020 and the Sellers or DISH may terminate the Asset Purchase Agreement if the Prepaid Transaction is not consummated by the date that is 90 days following the consummation of the Merger Transactions.

In addition, the Asset Purchase Agreement requires the Sellers to indemnify DISH for certain damages arising out of breaches of representations or warranties of the Sellers in the Asset Purchase Agreement, certain failures to perform obligations under the Asset Purchase Agreement, excluded liabilities, and certain other matters.

At the closing of the Prepaid Transaction (which will not occur unless the Merger Transactions have been consummated), the Sellers and DISH will enter into (i) a License Purchase Agreement pursuant to which (a) the Sellers will sell certain 800 MHz spectrum licenses held by Sprint to DISH for a total of approximately \$3.6 billion in a transaction to be completed, subject to certain additional closing conditions, following an application

for FCC approval to be filed three years following the closing of the Merger Transactions and (b) the Sellers will have the option to lease back from DISH, as needed, a portion of the spectrum sold for an additional two years following the closing of the spectrum sale transaction, (ii) a Transition Services Agreement providing for the Sellers' provision of standard transition services to DISH in connection with the Prepaid Business for a period of up to three years following the closing of the Prepaid Business for a period of up to seven years following the closing of the Prepaid Transaction, and (iv) an Option to Acquire Tower and Retail Assets offering DISH the option to take on leases for certain decommissioned towers and retail locations from the Sellers, subject to obtaining all necessary third-party consents, for a period of up to five years following the closing of the Prepaid Transaction. If DISH breaches the License Purchase Agreement prior to the closing of such agreement or fails to deliver the purchase price following the satisfaction or waiver of all closing conditions, DISH's sole liability to T-Mobile will be to pay T-Mobile a fee of approximately \$72 million.

The foregoing description of the Asset Purchase Agreement is not complete and is qualified in its entirety by reference to a copy of the Asset Purchase Agreement, which is filed as Exhibit 2.1 hereto and is incorporated herein by reference.

Amendment to Business Combination Agreement

On July 26, 2019, in connection with the entry into the Asset Purchase Agreement, T-Mobile and the other parties to the Business Combination Agreement (as defined below) entered into Amendment No. 1 (the "Amendment") to the Business Combination Agreement (the "Business Combination Agreement"), dated as of April 29, 2018, by and among T-Mobile, Sprint, Huron Merger Sub LLC, Superior Merger Sub Corporation, Starburst I, Inc., Galaxy Investment Holdings, Inc., and for the limited purposes set forth therein, Deutsche Telekom, Deutsche Telekom Holding B.V. and SoftBank. As previously disclosed, the Business Combination Agreement provides for T-Mobile and Sprint to combine their respective businesses, on the terms and subject to the conditions set forth in the Business Combination Agreement (the "Merger Transactions").

The Amendment extends the Outside Date (as defined in the Business Combination Agreement) to November 1, 2019, or, if the Marketing Period (as defined in the Business Combination Agreement) has started and is in effect at such date, then January 2, 2020. The Amendment also provides that the closing of the Merger Transactions will occur on the first business day of the first month (other than the third month of any calendar quarter) where such first business day is at least three business days following the satisfaction or waiver of all of the conditions to the closing of the Merger Transactions, or, if the Marketing Period has not ended at the time of such satisfaction or waiver, the closing shall occur on the earlier of (a) any date during or after the Marketing Period specified by T-Mobile (subject to the consent of Sprint to the extent such date falls after the Outside Date) or (b) the first business day of the first month (other than the third month of any calendar quarter) where such first business day is at least three business days following the final day of the Marketing Period. The Amendment also modifies the Business Combination Agreement so as to limit the actions the parties may be required to undertake or agree to in order to obtain any remaining governmental consents or avoid an action or proceeding by any governmental entity in connection with the Merger Transactions, recognizing the substantial undertakings already agreed to by the parties, including the transactions contemplated by the Asset Purchase Agreement.

The foregoing description of the Amendment is not complete and is qualified in its entirety by reference to a copy of the Amendment, which is filed as Exhibit 2.2 hereto and is incorporated herein by reference.

The completion of the Merger Transactions remains subject to regulatory approvals and other closing conditions, pursuant to the Business Combination Agreement, as amended. T-Mobile and Sprint expect to receive final federal regulatory approval in the third quarter of 2019 and currently anticipate that the Merger Transactions will be permitted to close in the second half of the year.

Item 8.01 Other Events.

On July 26, 2019, the U.S. Department of Justice (the "DOJ") filed a complaint and a proposed final judgment (the "Proposed Consent Decree") agreed to by T-Mobile, Deutsche Telekom, Sprint, SoftBank, and DISH with the U.S. District Court for the District of Columbia. The Proposed Consent Decree would fully resolve the DOJ's investigation into the Merger Transactions and would require the parties to, among other things, carry out the divestitures to be made pursuant to the Asset Purchase Agreement described above upon the closing of the Merger Transactions. The Proposed Consent Decree is subject to judicial approval under the Antitrust Procedures and Penalties Act, 15 U.S.C. § 16, for the limited purpose of determining whether the Proposed Consent Decree is in the public interest.

Also, on July 26, 2019, T-Mobile and Sprint issued a press release announcing the agreement to divest the Prepaid Business and the DOJ's filing of the Proposed Consent Decree. A copy of the press release is filed as Exhibit 99.1 hereto and is incorporated herein by reference.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

Exhibit <u>No.</u>	Description
2.1	Asset Purchase Agreement, dated as of July 26, 2019, by and among T-Mobile US, Inc., Sprint Corporation and DISH Network Corporation.*
2.2	Amendment No. 1, dated as of July 26, 2019, to the Business Combination Agreement, dated as of April 29, 2018, by and among T-Mobile US, Inc., Huron Merger Sub LLC, Superior Merger Sub Corporation, Sprint Corporation, Starburst I, Inc., Galaxy Investment Holdings, Inc., and for the limited purposes set forth therein, Deutsche Telekom AG, Deutsche Telekom Holding B.V., and SoftBank Group Corp.*
99.1	Press Release dated July 26, 2019.

* This filing excludes certain schedules pursuant to Item 601(a)(5) of Regulation S-K, which the registrant agrees to furnish supplementally to the Securities and Exchange Commission upon request by the Commission.

Important Additional Information

In connection with the proposed transaction, T-Mobile US, Inc. ("T-Mobile") has filed a registration statement on Form S-4 (File No. 333-226435), which was declared effective by the U.S. Securities and Exchange Commission (the "SEC") on October 29, 2018, and which contains a joint consent solicitation statement of T-Mobile and Sprint Corporation ("Sprint"), that also constitutes a prospectus of T-Mobile (the "joint consent solicitation statement/prospectus"), and each party will file other documents regarding the proposed transaction with the SEC. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE JOINT CONSENT SOLICITATION STATEMENT/PROSPECTUS AND OTHER RELEVANT DOCUMENTS FILED WITH THE SEC WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. The documents filed by T-Mobile may be obtained free of charge at T-Mobile's website, at www.t-mobile.com, or at the SEC's website, at www.sec.gov, or from T-Mobile by requesting them by mail at T-Mobile US, Inc., Investor Relations, 1 Park Avenue, 14th Floor, New York, NY 10016, or by telephone at 212-358-3210. The documents filed by Sprint may be obtained free of charge at Sprint's website, at www.sprint.com, or at the SEC's website, at www.sec.gov, or from Sprint by requesting them by mail at Sprint Corporation, Shareholder Relations, 6200 Sprint Parkway, Mailstop KSOPHF0302-3B679, Overland Park, Kansas 66251, or by telephone at 913-794-1091.

No Offer or Solicitation

This communication shall not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended.

Cautionary Statement Regarding Forward-Looking Statements

This communication contains certain forward-looking statements concerning T-Mobile, Sprint and the proposed transaction between T-Mobile and Sprint. All statements other than statements of fact, including information concerning future results, are forward-looking statements. These forward-looking statements are generally identified by the words "anticipate," "believe," "estimate," "expect," "intend," "may," "could" or similar expressions. Such forward-looking statements include, but are not limited to, statements about the benefits of the proposed transaction, including anticipated future financial and operating results, synergies, accretion and growth rates, T-Mobile's, Sprint's and the combined company's plans, objectives, expectations and intentions, and the expected timing of completion of the proposed transaction. There are several factors which could cause actual plans and results to differ materially from those expressed or implied in forward-looking statements, Such factors include, but are not limited to, the failure to obtain, or delays in obtaining, required regulatory approvals, and the risk that such approvals may result in the imposition of conditions that could adversely affect the combined company or the expected benefits of the proposed transaction, or the failure to satisfy any of the other conditions to the proposed transaction on a timely basis or at all; the occurrence of events that may give rise to a right of one or both of the parties to terminate the business combination agreement; adverse effects on the market price of T-Mobile's or Sprint's common stock and on T-Mobile's or Sprint's operating results because of a failure to complete the proposed transaction in the anticipated timeframe or at all; inability to obtain the financing contemplated to be obtained in connection with the proposed transaction on the expected terms or timing or at all; the ability of T-Mobile, Sprint and the combined company to make payments on debt or to repay existing or future indebtedness when due or to comply with the covenants contained therein; adverse changes in the ratings of T-Mobile's or Sprint's debt securities or adverse conditions in the credit markets; negative effects of the announcement, pendency or consummation of the transaction on the market price of T-Mobile's or Sprint's common stock and on T-Mobile's or Sprint's operating results, including as a result of changes in key customer, supplier, employee or other business relationships; significant transaction costs, including financing costs, and unknown liabilities; failure to realize the expected benefits and synergies of the proposed transaction in the expected timeframes or at all; costs or difficulties related to the integration of Sprint's network and operations into T-Mobile; the risk of litigation or regulatory actions, including the antitrust litigation brought by the attorneys general of thirteen states and the District of Columbia; the inability of T-Mobile, Sprint or the combined company to retain and hire key personnel; the risk that certain contractual restrictions contained in the business combination agreement during the pendency of the proposed transaction could adversely affect T-Mobile's or Sprint's ability to pursue business opportunities or strategic transactions; effects of changes in the regulatory environment in which T-Mobile and Sprint operate; changes in global, political, economic, business, competitive and market conditions; changes in tax and other laws and regulations; and other risks and uncertainties detailed in the Form S-4, as well as in T-Mobile's Annual Report on Form 10-K for the fiscal year ended December 31, 2018 and in its subsequent reports on Form 10-Q, including in the sections thereof captioned "Risk Factors" and "Cautionary Statement Regarding Forward-Looking Statements," as well as in its subsequent reports on Form 8-K, all of which are filed with the SEC and available at www.sec.gov and www.t-mobile.com. Forward-looking statements are based on current expectations and assumptions, which are subject to risks and uncertainties that may cause actual results to differ materially from those expressed in or implied by such forward-looking statements. Given these risks and uncertainties, persons reading this communication are cautioned not to place undue reliance on such forward-looking statements. T-Mobile assumes no obligation to update or revise the information contained in this communication (whether as a result of new information, future events or otherwise), except as required by applicable law.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: July 26, 2019 T-MOBILE US, INC.

By: /s/ J. Braxton Carter

J. Braxton Carter

Executive Vice President and Chief Financial Officer

ASSET PURCHASE AGREEMENT

AMONG

T-MOBILE US, INC.

SPRINT CORPORATION

AND

DISH NETWORK CORPORATION

DATED AS OF July 26, 2019

CONTENTS

		Page
ARTICLE	E 1 DEFINITIONS	1
1.1	Certain Defined Terms	1
1.2	Table of Defined Terms	13
1.3	Principles of Interpretation	13
ARTICLE	E 2 PURCHASE AND SALE	14
2.1	Purchase and Sale	14
2.2	Excluded Assets	14
2.3	Assumption of the Liabilities	16
2.4	Purchase Price	18
2.5	Certain Adjustments to Purchase Price	18
2.6	Allocation of the Purchase Price	20
ARTICLE	E 3 REPRESENTATIONS AND WARRANTIES OF THE SELLERS	21
3.1	Organization	21
3.2	Authorization, Enforceability	21
3.3	No Approvals or Conflicts	21
3.4	Litigation	22
3.5	Compliance with Law; Permits	22
3.6	Financial Information	23
3.7	No Undisclosed Liabilities	24
3.8	Absence of Certain Changes	24
3.9	Tax Matters	24
3.10		25
3.11	Real Property	26
3.12	Intellectual Property	26
3.13	Data Protection	27
3.14		28
3.15		29
3.16	Principal Suppliers and Principal Distributors	29
3.17	Transactions with Affiliates	30
3.18	Brokers	30
3.19	No Other Representations and Warranties	30
ARTICLE	E 4 REPRESENTATIONS AND WARRANTIES OF THE BUYER	30
4.1	Organization	30
4.2	Authorization, Enforceability	30
4.3	No Approvals or Conflicts	31
4.4	Litigation; Regulatory Approvals	31
4.5	Financing	31
4.6	Brokers	31
4.7	Solvency	31
4.8	No Other Representations or Warranties	32

i

4.9	No Reliance	32
ARTICLE	E 5 PRE-CLOSING COVENANTS	32
5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12	Conduct of Business Prior to the Closing Regulatory Approvals Third Party Consents Shared Contracts Insurance Exclusivity Further Assurances Access and Information Financing; Financing Cooperation	32 34 37 38 38 39 39 40 41 41
5.12	Cooperation Regarding Onboarding	42
	E 6 EMPLOYEE MATTERS	42
6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11	Offer of Employment Seller Benefit Plans Benefit Levels Service Credit Terms of Coverage 401(k) Plan Benefits Liability Retention Incentives Transaction Bonuses WARN Act No Third Party Beneficiaries	42 43 43 43 44 44 44 44 44 44
ARTICLE	E 7 TAX MATTERS	45
7.1 7.2 7.3 7.4	Mutual Cooperation Payroll Tax Reporting Allocation of Certain Taxes Transfer Taxes	45 45 45 46
ARTICLE	E 8 INTELLECTUAL PROPERTY MATTERS	46
8.1 8.2 8.3	IP Transfers Use of Retained Marks Trade Secret License	46 47 47
ARTICLE	E 9 CONDITIONS TO CLOSING	47
9.1 9.2 9.3	Conditions to the Obligations of the Sellers and the Buyer Conditions to the Obligations of the Buyer Conditions to the Obligations of the Sellers	47 48 49

ARTICLE 1	10 CLOSING	49
10.1 10.2	Closing Deliverables	49 50
ARTICLE 1	11 TERMINATION	51
11.1 11.2	Termination Effect of Termination	51 52
ARTICLE 1	12 POST-CLOSING COVENANTS	52
12.1 12.2 12.3 12.4 12.5 12.6 12.7	Confidentiality Access to Books and Records; Cooperation Wrong-Pocket Transfer Formalities and Costs Insurance Non-Solicitation Additional Post-Closing Covenants	52 54 54 56 56 57
ARTICLE 1	13 INDEMNIFICATION	57
13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11	Indemnification by the Sellers Indemnification by Buyer Indemnification Claims Survival of Representations and Warranties and Covenants Treatment of Indemnification Payments Limitations on Indemnification Exclusivity of Remedy for Monetary Damages Effect of Insurance and Other Recoveries Effect of Investigation Mitigation No Recourse 14 MISCELLANEOUS	57 58 58 60 60 60 61 61 61 61 62
14.1 14.2 14.3 14.4 14.5 14.6 14.7 14.8 14.9 14.10 14.11 14.12 14.13 14.14	Fees and Expenses Further Actions Governing Law Waiver of Jury Trial Submission to Jurisdiction Specific Performance Amendment No Assignment Waiver Withholding Notices Complete Agreement Publicity Severability Third Parties	62 62 62 63 63 63 63 64 64 64 65 66 66

14.16	Waiver of Conflicts	67
14.17	Bulk Transfer Laws	67
14.18	No Other Duties	67
14.19	Counterparts and Delivery	67

Annexes

Annex A Table of Defined Terms

Exhibits

Exhibit A Agreed Accounting Principles

Exhibit B Example Net Working Capital Statement

Exhibit C Form of MNSA

Exhibit D Form of Short-Form IP Assignment Agreement

Exhibit E Form of Site Option Agreement

Exhibit F Form of Spectrum Purchase Agreement Exhibit G Form of Transition Services Agreement

Schedules

Schedule II Specified Assets
Schedule II Specified Liabilities
Schedule III Transferred Contracts

Schedule IV Transferred Intellectual Property

Schedule V Shared Contracts
Schedule VI Certain Information
Schedule 1.1(a) Knowledge Parties

Schedule 1.1(b) Permitted Encumbrances

Schedule 2.3(b)(i)(7) Excluded Liabilities

Schedule 3.4(a) Litigation

Schedule 3.5 Compliance with Law; Permits

Schedule 3.6 Financial Information
Schedule 3.7 Undisclosed Liabilities

Schedule 3.9 Tax Matters

Schedule 3.10 Employee Matters
Schedule 3.12 Intellectual Property
Schedule 3.14 Material Contracts

Schedule 3.16 Principal Suppliers and Principal Distributors

Schedule 5.1 Conduct of Business

Schedule 5.12 Certain Commercial Arrangements

Schedule 6.1 Business Employees
Schedule 6.8 Retention Agreements

Schedule 12.7 Additional Post-Closing Covenants

ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement, dated as of July 26, 2019 (this "<u>Agreement</u>"), is made by and among T-Mobile US, Inc., a Delaware corporation ("<u>TMUS</u>"), Sprint Corporation, a Delaware corporation ("<u>Sprint</u>" or the "<u>Company</u>" and collectively with TMUS, the "<u>Sellers</u>") and DISH Network Corporation, a Nevada corporation (the "<u>Buyer</u>"). Each of TMUS, Sprint and the Buyer is referred to herein as a "<u>Party</u>", and collectively as the "<u>Parties</u>"

RECITALS

WHEREAS, TMUS and Sprint entered into that Business Combination Agreement dated as of April 29, 2018 (as it may be amended from time to time, the "NTM Merger Agreement.") pursuant to which Sprint will merge with a wholly-owned subsidiary of TMUS (such merger, the "NTM Merger");

WHEREAS, the Buyer desires to acquire from the Sellers and the Sellers desire to sell to the Buyer the Company's Boost Mobile, Virgin Mobile and Sprint-branded prepaid wireless businesses and the Company's 800 MHz spectrum holdings;

WHEREAS, concurrently with the entry into this Agreement, the Parties have entered into a Stipulation and Order with the DOJ regarding this Transaction and the NTM Merger;

WHEREAS, in connection therewith, the Buyer and TMUS intend to enter into the Ancillary Agreements at or prior to the Closing Date;

WHEREAS, the Parties desire to make certain representations, warranties, covenants and agreements in connection with this Agreement, the Ancillary Agreements and the transactions contemplated hereby and thereby; and

WHEREAS, the Sellers desire to transfer to the Buyer or its Subsidiaries, and the Buyer desires to assume from the Sellers certain assets, obligations and liabilities as specified in this Agreement, upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the premises and the mutual agreements and covenants herein set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound, the Parties hereby agree as follows:

ARTICLE 1 DEFINITIONS

- 1.1 Certain Defined Terms. As used in this Agreement, the following terms shall have the following meanings:
- "401(k) Plan" means the defined contribution retirement plan intended to qualify under Section 401(a) of the Code in which the Business Employees participate as of immediately prior to the Closing.
- "Affiliate" means, with respect to any specified Person, any other Person that directly or indirectly, through one or more intermediaries, controls or is controlled by or is under common control with such Person, in each case from time to time. The term "control" (including its correlative meanings "controlled by" and "under common control with") means

the possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise); *provided*, *however*, that none of SoftBank Group Corp. ("SoftBank"), Deutsche Telekom AG ("DT") and their respective Affiliates (as defined without giving effect to this proviso), other than the controlled Affiliates of Sprint and the controlled Affiliates of TMUS, shall be deemed to be an Affiliate of Sprint or of TMUS or any of their respective Subsidiaries.

- "Agreed Accounting Principles" means GAAP, subject to the accounting methods, policies, principles, practices and procedures set forth in Exhibit A hereto, using the same accounting methods, policies, principles, practices and procedures, as were used in the preparation of the most recent select balance sheet accounts of the Company; provided that in the event of a conflict between GAAP, on the one hand, and the accounting methods, policies, principles, practices, procedures, classifications, judgments and estimation methodology used in the preparation of the most recent balance sheet of the Company, on the other hand, the terms of Exhibit A and the accounting methods, policies, principles, practices, procedures, classifications, judgments and estimation methodology used in the preparation of the most recent balance sheet of the Company shall apply.
- "Amended and Restated Confidentiality Agreement" means that certain Amended and Restated Confidentiality Agreement entered into among the Buyer, the Company and TMUS, dated as of July 26, 2019, as modified, supplemented or amended from time to time in accordance with its terms.
- "Ancillary Agreements" means the Bill of Sale and Assignment and Assumption Agreement, the Transition Services Agreement, the MNSA, the Site Option Agreement, the Spectrum Purchase Agreement and the Short-Form IP Assignment Agreement.
- "Antitrust Laws" means the HSR Act, the Clayton Act, the Federal Trade Commission Act and all other antitrust, competition or trade regulation Laws, including all Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening competition through merger or acquisition.
- "Boost Business" means the prepaid wireless business conducted by the Company and its Subsidiaries under the Boost Mobile brand, but excluding the prepaid wireless business conducted by the Company under the Assurance Lifeline brand and excluding, for the avoidance of doubt, the prepaid wireless customers of each of Shentel and Swiftel.
 - "Business" means the Boost Business, Virgin Mobile Business and the Sprint Prepaid Business.
- "Business Day" means any day other than a Saturday, Sunday or day on which banks are closed in New York, New York. If any period expires on a day which is not a Business Day or any event or condition is required by the terms of this Agreement to occur or be fulfilled on a day which is not a Business Day, such period shall expire or such event or condition shall occur or be fulfilled, as the case may be, on the next succeeding Business Day.
- "Claim Notice" means a written notification which contains (a) a description of the Losses incurred or reasonably expected to be incurred by the Indemnified Party and the Claimed Amount of such Losses, to the extent then known and (b) a statement of the provisions under the Agreement upon which such claim is based.

- "Claimed Amount" means the amount of any Losses incurred or reasonably expected to be incurred by the Indemnified Party (to the extent then known).
 - "Clayton Act" means the Clayton Antitrust Act of 1914, as amended.
 - "Code" means the Internal Revenue Code of 1986, as amended.
- "Contract" means any contract, agreement, lease, license, commitment, understanding, franchise, warranty, guaranty, mortgage, note, bond or other instrument or consensual obligation that is legally binding, but excluding any Plan.
 - "Controlling Party" means the party controlling the defense of any Third Party Claim.
- "Current Assets" means, as of any particular time, without duplication, the sum of the amounts outstanding as of such time for each of the line items set forth under the heading "Current Assets" in the Example Net Working Capital Statement for the Company and its Affiliates in respect of the Business, as determined in accordance with the Agreed Accounting Principles and the Example Net Working Capital Statement, provided, that Current Assets, as reflected on the Example Net Working Capital Statement, shall not include Tax assets of the Company and its Affiliates in respect of the Business.
- "Current Liabilities" means, as of any particular time, without duplication, the sum of the amounts outstanding as of such time for each of the line items set forth under the heading "Current Liabilities" in the Example Net Working Capital Statement for the Company and its Affiliates in respect of the Business, as determined in accordance with the Agreed Accounting Principles and the Example Net Working Capital Statement, provided, that Current Liabilities, as reflected on the Example Net Working Capital Statement, shall not include Tax liabilities of the Company and its Affiliates in respect of the Business.
 - "DOJ" means the United States Department of Justice or any successor entity thereto.
- "DOJ Consent" means the consent, agreement and approval of the DOJ with respect to the Buyer, this Agreement and the transactions contemplated hereby, or as required under the DOJ Final Judgment or the Stipulation and Order the DOJ may file in any court in connection with the NTM Merger, in form and substance acceptable to the Sellers in their sole discretion, provided that (1) entry by the Sellers into the Stipulation and Order on the date hereof shall mean that such DOJ Consent in effect at such time is acceptable in form and substance to the Sellers and (2) the consummation of the NTM Merger shall mean that such DOJ Consent in effect at such time is acceptable in form and substance to the Sellers.
- "DOJ Final Judgment" means any proposed final judgment the DOJ may file in any court, including a United States District Court, in connection with the NTM Merger, which shall include the DOJ Consent, as such proposed final judgment may be modified by, or with the approval of, any court, in each case in form and substance acceptable to the Sellers in their sole discretion, provided that (1) entry by the Sellers into the Stipulation and Order on the date hereof shall mean that such DOJ Final Judgment in the form contemplated by the Stipulation and Order entered into by the Sellers is acceptable in form and substance to the Sellers and (2) the consummation of the NTM Merger shall mean that such DOJ Final Judgment in effect at such time is acceptable in form and substance to the Sellers.

- "Encumbrance" means any security interest, pledge, lien, license, mortgage, charge, covenant not to sue, purchase option, hypothecation, easement, title defect or other encumbrance of any kind, whether voluntary or imposed by applicable Law, and any agreement to give any of the foregoing.
 - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "Example Net Working Capital Statement" means the example statement setting forth the calculation of Net Working Capital as of March 31, 2019 attached as Exhibit B hereto.
- "Expected Claim Notice" shall mean a notice that, as a result of a Proceeding instituted by or written claim made by a Third Party, an Indemnified Party reasonably expects to incur Losses for which it is entitled to indemnification under Article 13.
 - "FCC" means the Federal Communications Commission or any successor entity thereto.
 - "FCC 214 Application" has the meaning set forth in Section 5.2(c).
- "FCC 214 Approval" means the approval of the FCC 214 Application under the FCC's streamlined processing procedures pursuant to 47 C.F.R. § 63.12, or if the FCC does not apply, or removes the FCC 214 Application from, its streamlined processing procedures, FCC approval shall be reflected either in the FCC's International Bureau Filing System or by Public Notice.
 - "FCC Consent" means the order of the FCC approving the transfers of FCC licenses and spectrum leases associated with the NTM Merger.
- "FCC MNSA Approval" means the approval of the MNSA by the FCC pursuant to Section IV of Attachment 2 of Seller's FCC filing dated May 20, 2019, with such changes as the FCC may reasonably request.
- "Fraud" means an act, committed by or on behalf of a Party and requires a knowingly false representation made herein (with respect to the Sellers, the making of the representations expressly set forth in Article 4 or, in either the case of the Sellers or the Buyer, in any schedule, exhibit or certificate delivered in connection with the Transaction) which constitutes common law fraud under applicable Law.
- "Governing Document" means, with respect to any Person, as applicable, such Person's (a) Articles of incorporation, certificate of incorporation, certificate of formation, Articles of organization, Articles of association, certificate of limited partnership or other applicable similar organizational or charter documents relating to the creation or organization of such Person, and (b) bylaws, operating agreement, shareholder agreement, partnership agreement, or other applicable similar documents relating to the operation, governance or management of such Person.
- "Governmental Authority" means any supra-national, or United States or foreign, national, federal, state, regional, municipal or local government (including any subdivision, court, tribunal, administrative agency or commission or other authority thereof) or any quasi-governmental authority, regulatory or administrative body or other private body exercising any legislative, judicial, regulatory, taxing, importing, self-regulatory or other governmental or quasi-governmental authority (including any stock exchange or antitrust authority).

- "Governmental Order" means any decision, ruling, order, requirement, writ, injunction, decree, stipulation, determination, award, binding agreement or judgment issued or entered by or with any Governmental Authority.
 - "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.
- "Indebtedness" means, with respect to any Person at any specified time, any of the following Liabilities, obligations or undertakings (whether or not contingent and including, without limitation, any and all principal, accrued and unpaid interest, prepayment premiums or penalties, related expenses, commitment and other fees, sale or liquidity participation amounts, reimbursements, indemnities and other amounts which would be payable in connection therewith): (a) for borrowed money or in respect of loans or advances (whether or not evidenced by bonds, debentures, notes, or other similar instruments or debt securities); (b) under or in connection with letters of credit or bankers' acceptances, performance bonds, sureties or similar obligations that have been drawn down, in each case, to the extent of such draw; (c) for capitalized leases (as determined in accordance with GAAP) or to pay the deferred and unpaid purchase price of property or equipment; (d) pursuant to securitization or factoring programs or arrangements; (e) to maintain or cause to be maintained the financing, financial position or covenants of others or to purchase the obligations or property of others; (f) net cash payment obligations of such Person under swaps, options, forward sales contracts, derivatives and other hedging contracts, financial instruments or arrangements that may be payable upon termination or expiration thereof; (g) pursuant to guarantees or other arrangements having the economic effect of a guarantee of any obligation or undertaking of any other Person contemplated by the foregoing clauses (a) through (f) of this definition.
- "Indemnified Party" means the Buyer Indemnified Party or the Seller Indemnified Party, as the case may be, entitled, or seeking to assert rights, to indemnification under Article 13.
 - "Indemnifying Party" means the Party from whom indemnification is sought by the Indemnified Party.
- "Intellectual Property" means all of the following, anywhere in the world: Patents, Marks, Trade Secrets and other proprietary and confidential information and data, copyrights (including Software), database rights, technical drawings, designs, or other intellectual property rights (whether subject to registration or not) and all registrations and applications therefor and all renewals, extensions, restorations and reversions thereof.
- "Inventory" means the inventory of handsets of the Company or any of its controlled Affiliates that relates exclusively or are otherwise allocated by the Company to the Business, wherever located, including all supplies, spare parts, and goods that relate exclusively to the Business or are otherwise allocated by the Company to the Business, whether held at any location or facility owned or leased by the Company or any of its Affiliates or in transit to the Company or any of its controlled Affiliates or held by third parties on behalf of or for use in the Business.

- "Knowledge of the Sellers" or "Knowledge of such Seller" mean, with respect to the Sellers or with respect to such Seller, the actual knowledge, after reasonable inquiry, of the individuals set forth on Schedule 1.1(a).
- "Law" means any United States federal, state or local, foreign, multinational or other law (including common law), statute, rule, ordinance, regulation or Governmental Order of any Governmental Authority.
- "<u>Liabilities</u>" means all liabilities, risks, commitments and obligations of whatever kind and nature, primary or secondary, direct or indirect, asserted or unasserted, absolute or contingent, known or unknown, whether or not accrued.
- "Loss" means any loss (including diminution in value), Liability, claim, damage, expense (including reasonable legal fees and expenses or other professional fees and expenses), court cost, amount paid in settlement, other expense associated with enforcing any right hereunder, expense for investigation and ongoing monitoring and remediation expense; provided, however, that "Loss" shall not include any punitive, indirect, consequential, special or incidental damages (except (x) to the extent of awarded to a Third Party in a Third Party Claim or (y) any indirect, consequential, special or incidental damages that are a reasonably foreseeable consequence of a breach hereunder).
- "Marks" means trademarks, service marks, trade names, trade dress, business and corporate names, logos, trade styles, brand names, product names, domain names and reservations, web addresses, Internet number assignments, slogans, any other source identifiers of any kind or nature, together with all translations, localizations, adaptations, derivations and combinations thereof, including any common law rights, and any applications (including intent to use applications), registrations and renewals for any of the foregoing and the goodwill associated with the foregoing.
- "Material Adverse Effect" means an event, development, circumstance, change or effect that has, individually or in the aggregate, a materially adverse effect on (a) the Transferred Assets and the Assumed Liabilities, taken as a whole, or the assets, Liabilities, results of operations or condition (financial or otherwise) of the Business, taken as a whole or (b) the ability of the Company to consummate the Transactions; provided, however, that for purposes of clause (a) above, no event, development, circumstance, change or effect resulting from or arising out of any of the following shall be deemed to constitute, or shall be taken into account in determining whether there has been, a "Material Adverse Effect" (i) events, developments, circumstances, changes or effects in economic, monetary, or financial conditions in the United States, including changes in prevailing interest rates, credit markets, or financial market conditions in or affecting the United States, (ii) events, developments, circumstances, changes or effects in the prepaid wireless service industry in the United States, (iii) events, developments, circumstances, changes or effects in global or national political conditions, including the outbreak or escalation of war or acts of terrorism, (iv) earthquakes, hurricanes, tsunamis, typhoons, lightning, hailstorms, blizzards, tornadoes, droughts, floods, cyclones, mudslides, wildfires, weather conditions and other natural disasters, (v) changes in applicable Law or the interpretations thereof, (vii) changes in any acceptable accounting standards applicable to the Business or the interpretations thereof, (vii) any failure, in and of itself, by the Company and its Subsidiaries or the Business to meet any internal or published industry analyst projections or forecasts, or estimates of revenue or earnings for any period (it being understood that the facts and circumstances giving rise to such failure in this clause (vii) (that are not otherwise excluded from the definition of Material Adverse Effect by cla

through (vi) or clause (viii) hereof) may be taken into account in determining whether there is or has been a Material Adverse Effect), (viii) events, developments, circumstances, changes or effects directly resulting from the execution, announcement or pendency of this Agreement, the Ancillary Agreements, the transactions contemplated hereby or thereby or the identity of the Buyer or any of its Affiliates as the acquiror of the Transferred Assets (provided that the exception in this clause (viii) shall not apply to any representation regarding non-contravention, including Section 3.3.), and (ix) the consummation of the NTM Merger in accordance with its terms (provided that the exception in this clause (ix) shall not apply to any representation regarding non-contravention, including Section 3.3.)); provided, further, that, with respect to clauses (i) through (vi) such event, development, circumstance, change or effect will be taken into account in determining whether a Material Adverse Effect has occurred or is occurring to the extent it disproportionately adversely affects the Business (taken as a whole) relative to other companies operating in the prepaid wireless service industry in the United States.

- "MNSA" means the mobile virtual network operator agreement in the form attached to this Agreement as Exhibit C, as may it be amended in order to obtain the FCC MNSA Approval.
- "Net Working Capital" means an amount equal to Current Assets (less the carrying value of Rejected Inventory) minus Current Liabilities of the Company and its controlled Affiliates in respect of the Business as of 12:01 a.m. Eastern Time on the Closing Date.
 - "Non -Controlling Party" means the party not controlling the defense of any Third Party Claim.
 - " NTM Network" means the "T-Mobile Network" as such term is defined in the MNSA.
- "Ordinary Course of Business" means, with respect the Company, the ordinary course of business of the Company and its Subsidiaries in respect of the Business, consistent with past practice.
- "Parent Affiliates" means SoftBank and its Affiliates (other than the controlled Affiliates of TMUS) and DT and its Affiliates (other than the controlled Affiliates of Sprint).
 - "Patents" means patents, utility models, design patents and registered designs.
- "Permit" means permit, license, registration, certificate, approval, franchise, variance and similar authorizations in respect of the Transferred Assets and issued by or obtained from any Governmental Authority.
- "<u>Permitted Encumbrances</u>" means, as applicable (i) Encumbrances (A) specifically reflected or specifically reserved against or otherwise disclosed in the Financial Information set forth on <u>Schedule 3.6</u> or (B) incurred in the Ordinary Course of Business since the date of the Financial Information set forth on <u>Schedule 3.6</u>, (ii) mechanics', carriers', workmen's, repairmen's or other like Encumbrances arising or incurred in the Ordinary Course of Business, (iii) Encumbrances arising under original purchase price conditional sales contracts and equipment leases with third parties entered into in the Ordinary Course of Business (but not breaches thereof), (iv) Encumbrances which do not, individually or in the aggregate, materially impair the value of the applicable Transferred Asset or the continued use and operation of the applicable Transferred Asset to which they relate in the conduct of the applicable Business as presently conducted or as anticipated to be conducted, (v) Encumbrances for Taxes and other

governmental charges (A) that are not yet due or payable or which may thereafter be paid without penalty, or (B) which are being contested in good faith and that are, in the case of this clause (B), not material in amount or effect on the Business, (vi) licenses of Transferred Intellectual Property in the Ordinary Course of Business, (vii) Encumbrances arising under Indebtedness or Contracts that will be terminated, be released or will no longer apply upon consummation of the NTM Merger in accordance with its terms, and (viii) Encumbrances set forth in Schedule 1.1(b). For the avoidance of doubt, the Permitted Encumbrances in clause (vii) will not constitute a "Permitted Encumbrance" at the Closing.

- "Person" means any individual, sole proprietorship, partnership, firm, corporation, association, trust, unincorporated organization, joint venture, company, limited liability company or other entity or a Governmental Authority.
- "Personal Information" means any information that identifies or could reasonably be used to identify an individual, browser or device that identifies a specific person and any other personally identifying information that is subject to any applicable Laws.
- "Plan" means each employee benefit plan (within the meaning of Section 3(3) of ERISA, whether or not subject thereto), and each other employee benefit or compensatory plan, program, contract, policy, practice, agreement or arrangement, whether or not in writing and whether or not funded, in each case, sponsored or maintained by the Company or a Subsidiary of the Company (as determined immediately prior to the date of this Agreement) and which provides compensation and/or benefits to any Business Employee, including any employment, consulting, retirement, severance, termination or change in control agreements, and any plans or agreements providing for deferred compensation, equity-based incentives, bonuses, supplemental retirement, profit sharing, medical, welfare, vacation or fringe benefits or other employee benefits or remuneration of any kind, but excluding any plan, program, agreement or arrangement required by applicable Law or regulation (e.g., government mandated severance plans).
 - "Proceeding" means any lawsuit, arbitration, claim, action, administrative or regulatory challenge or proceeding.
- "Records" means books of account, ledgers, general, financial and accounting records, files, invoices, customer and supplier lists, other distribution lists, customer billing and credit records, sales and promotional literature, manuals and marketing studies, communications, accounting, sales and business files and records, property records, Tax records, product records, records related to licenses, records, files and documents with regard to the preparation, filing, prosecution, granting, maintenance and defense of Registered Intellectual Property included in the Transferred Intellectual Property, and other files and records, in each case, whether maintained in electronic or physical form, as applicable.
- "Registered Intellectual Property" means all registrations and applications for Patents, Marks and copyrights that have been filed with any Governmental Authority anywhere in the world, including the United States Patent and Trademark Office and the United States Copyright Office.
- "Regulatory Approvals" means the DOJ Consent, the FCC Consent, the FCC 214 Approval and the FCC MNSA Approval and any other material approvals required by Governmental Authorities in connection with the NTM Merger or the Transaction.

- "Representatives" means, in relation to any Party, its directors, officers, employees, agents, advisers, investment bankers, accountants, consultants and actual or potential debt and equity financing sources.
- "Retained Mark" means any Mark owned by or licensed to Sellers or any of their Subsidiaries that is not included in Transferred Intellectual Property.
- "Shared Contract" means (a) the Contracts set forth on Schedule V, (b) any Contract that both (1) involves, or is reasonably expected to involve, in each case solely with respect to the Business, annual revenue in excess of \$500,000 and (2) is designated by Buyer, by written notice to Sellers prior to the Closing, to be a Shared Contract pursuant to Section 5.4(a) and (c) any Contract identified by Buyer and Seller pursuant to Section 5.4(a) that is entered into by the Company or any of its Subsidiaries with a Third Party to obtain goods or services that relates both to the Business (or the Transferred Assets) and the other business of the Company and its Subsidiaries (or to the Excluded Assets).
 - "Shentel" means Shenandoah Telecommunications Company.
- "Short-Form IP Assignment Agreement" means the short-form IP assignment agreement substantially in the form attached to this Agreement as Exhibit D.
- "Site Option Agreement" means the option to purchase certain decommissioned cell tower and retail store sites in substantially the form attached hereto as Exhibit E.
 - "Software" means software in object code and source code format.
- "Solvent" means, with respect to a Person, that, as of any date of determination, (a) the amount of the fair saleable value of the assets of such Person will, as of such date, exceed (i) the value of all liabilities of such Person, including contingent and other liabilities, as of such date, and (ii) the amount that will be required to pay the probable liabilities of such Person on its existing debts (including contingent liabilities) as such debts become absolute and matured, (b) such Person will not have, as of such date, an unreasonably small amount of capital for the operation of the businesses in which it is engaged or proposed to be engaged following such date, and (c) such Person will be able to timely pay and satisfy its liabilities and other obligations, including contingent and other liabilities, when and as they mature. For purposes of this definition, "not have, as of such date, an unreasonably small amount of capital for the operation of the businesses in which it is engaged or proposed to be engaged" and "be able to timely pay and satisfy its liabilities and other obligations, including contingent and other liabilities, when and as they mature" means that such Person will be able to generate enough cash from operations, asset dispositions or refinancing, or a combination thereof, to meet its obligations as they become due.
- "Specified Assets" means (a) the Current Assets as of 12:01 a.m. Eastern Time on the Closing Date, including all Inventory, except for Rejected Inventory, (b) the Transferred Records, (c) the Transferred Contracts, (d) to the extent transferrable under applicable Law, in whole or in part, all rights to all causes of action, Proceedings, judgments or defenses against third parties and demands of any nature arising on or after the Closing Date, whether arising by way of counterclaim or otherwise, in each case to the extent (and only to the extent) related to any Transferred Asset or Assumed Liability, except for claims or refunds for any Taxes for a period (or portion thereof) ending on or before the Closing Date, (e) to the extent transferable under applicable Law, all Permits (and all applications therefor) used exclusively in, or

obtained exclusively for, the operation of the Business, (f) all purchase orders or other commitments exclusively related to the Business that remain unfulfilled as of the Closing, other than as expressly included in the Excluded Assets; (g) the Transferred Intellectual Property, (h) all personnel Records to the extent (and only to the extent) pertaining to the Transferred Employees, to the extent permitted by applicable Law, (i) all rights, claims and credits (including all guaranties, warranties, indemnities and similar rights) of the Sellers or any of their respective controlled Affiliates to the extent (and only to the extent) related to any Transferred Asset or Assumed Liability, (j) all rights to all insurance causes of action, Proceedings or judgments or other rights the Sellers or any of their respective Affiliates may have for insurance coverage under any past and present occurrence-based policies and occurrence-based insurance contracts insuring the Business or the Transferred Assets, to the extent a covered loss with respect to the Transferred Assets is incurred thereunder, net of any applicable deductible and net of any reasonable and unreimbursed out-of-pocket costs of recovery, in each case including any and all proceeds under any of Sellers' or any of their controlled Affiliates' Third Party insurance policies written prior to the Closing in connection with (A) the damage or destruction of any of the Transferred Assets from and after the date hereof and prior to the Closing that is, or would have been but for such damage or destruction, included in the Transferred Assets or (B) any Assumed Liability; (k) all vehicle leases and personal computers exclusively related to the Business, wherever located, (1) all prepaid wireless customer accounts of the Business, (m) goodwill solely to the extent associated with the Business, (n) funding commitments from handset manufacturers to the extent associated with Inventory that is not Rejected Inventory, (o) purchase orders to the extent identified to Buyer pursuant to Section 5.11(b), (p) a 100% undivided participation interest in the retail installment agreements for handsets held by customers of the Business pursuant to the Participation Agreement contemplated by the Transition Services Agreement, and (q) the other assets set forth on Schedule I.

"Specified Liabilities" means (a) the Current Liabilities as of 12:01 a.m. Eastern Time on the Closing Date, (b) all Liabilities to the extent required to be performed on or after the Closing under Transferred Contracts, Permits, approval or authorization constituting part of the Transferred Assets, including any and all of the Buyer's portion of the Shared Contracts, (c) any and all Liabilities to the extent relating to Taxes attributable to or imposed on the Business or the Transferred Assets for any period (or portion thereof) beginning after the Closing Date; (d) any Transfer Taxes for which Buyer is liable pursuant to Section 7.4; (e) subject to Section 6.8, any and all Liabilities relating to the Transferred Employees with respect to any period (or portion thereof) commencing on or after the Closing Date, including the Retention Agreements; (f) individual customer accommodations (i.e., customer credits or similar customer retention offerings) provided to customers in the ordinary course of business following Closing relating to ordinary course customer complaints whether arising prior to, at or after the Closing; provided that this shall not include any Proceedings initiated by any customer or any putative class action matter, in each case to the extent relating to periods prior the Closing, (g) all insurance premiums, retroactive premiums, premium adjustments and any out-of-pocket costs or expenses associated with collecting the insurance proceeds included in the Specified Assets, (h) all Liabilities associated with purchase orders or other commitments exclusively related to the Business that remain unfulfilled as of the Closing, other than as expressly included in the Excluded Liabilities and (i) the Liabilities set forth on Schedule II.

"Spectrum Purchase Agreement" means the license purchase agreement between TMUS and the Buyer in substantially the form attached hereto as Exhibit F.

- "Sprint Prepaid Business" means the prepaid wireless customers of the Company and its Subsidiaries under the Sprint brand, but excluding the prepaid wireless business conducted by the Company under the Assurance Lifeline brand and excluding, for the avoidance of doubt, the prepaid wireless customers of each of Shentel and Swiftel.
- "Subsidiary" means any Person of which any of TMUS, Sprint, the Buyer or any other Person, as the case may be, owns, directly or indirectly, a majority of the capital stock or other equity interests the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such Person, or otherwise owns, directly or indirectly, such capital stock or other equity interests that would confer control of any such Person, or any Person that would otherwise be deemed a "subsidiary" under Rule 12b-2 promulgated under the Securities Exchange Act of 1934, as amended.
 - "Swiftel" means Swiftel Communications, Inc.
- "<u>Tax</u>" or "<u>Taxes</u>" means all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, customs duty, capital stock, severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise production, value added, occupancy, and other taxes imposed by a Taxing Authority, together with all interest, penalties or additions to tax attributable to such taxes.
- "<u>Tax Return</u>" means any report, return, statement or other written information (including elections, declarations, disclosures, estimates and information returns) supplied or required to be supplied to a Taxing Authority in connection with any Taxes and any schedule thereof or any amendment thereto.
- "Taxing Authority" means any Governmental Authority having jurisdiction over the assessment, determination, collection or other imposition of Taxes.
 - "Termination Date" means the date that is 90 days following the closing of the NTM Merger.
- "Third Party" means, with respect to any specified Person, any other Person who is not an Affiliate of such specified Person (other than Governmental Authority).
- "Third Party Claim" means any Proceeding by a Person other than the Buyer or the Sellers for which indemnification may be sought by an Indemnified Party under Article 13.
- "Trade Secrets" means confidential or proprietary trade secrets meeting the definition of a trade secret under the Uniform Trade Secrets Act or other similar legislation or common laws governing protection of trade secrets or confidential information anywhere in the world, including information, know-how, data and databases.
 - " Transaction " means the transactions contemplated by this Agreement.
 - "Transaction Bonuses" means any transaction bonuses payable by the Sellers or their Affiliates on or after the Closing to the Transferred Employees.
- "Transaction Process" means all matters, whether occurring before or after the date of this Agreement and prior to the Closing, relating to the sale of the Transferred Assets and assumption of Assumed Liabilities and all activities in connection therewith, including matters relating to the drafting, negotiation or interpretation of any of the provisions of this Agreement, or the determination of the allocation of any assets or Liabilities pursuant to the foregoing agreement or the transactions contemplated thereby.

- "Transferred Contract" means (i) any Contract that relates exclusively to the Business, (ii) any Contract set forth on Schedule III, and (iii) that portion of any Shared Contracts that is identified by Buyer and Seller as reasonably necessary for Buyer to continue operation of the Business upon termination of any service provided under the Transition Services Agreement.
- "Transferred Intellectual Property" means (a) all Intellectual Property owned by the Company and its Subsidiaries and exclusively used in the Business, (b) any Software used either in (i) prepaid mobile applications or (ii) websites, in each case of (i) and (ii), that are exclusively related to the Business or that are otherwise set forth on Schedule IV, to the extent the Intellectual Property therein is owned by the Company or one or more of its Subsidiaries, and (c) all Intellectual Property set forth on Schedule IV.
- "Transaction Expenses" means, except as otherwise provided in this Agreement, all fees, costs and expenses of the Sellers incurred or payable by Sellers, their Affiliates or the Business in connection with the Transaction Process, this Agreement (and its performance hereunder) and the consummation of the Transaction or otherwise in connection with Sellers' exploration of alternatives relating to the Business that are unpaid as of the Closing, including (a) to professionals (including investment bankers, attorneys, accountants, brokers and other consultants and advisors) retained by the Sellers or any of their Affiliates that performed services in connection with the negotiation of this Agreement and the Ancillary Agreement and execution of this Agreement, the Ancillary Agreements and the consummation of the Transaction that are unpaid as of the Closing and (b) any expenses incurred by the Sellers in connection with the transactions contemplated by this Agreement for which any Seller or any of their Affiliates is liable.
- "Transferred Records" means all Records in the possession and control of the Company or its Subsidiaries, to the extent a transfer is permitted by Law, to the extent (and solely to the extent) that they relate to (i) the Transferred Assets, (ii) the Assumed Liabilities or (iii) the Business; provided, however, that if any Records contain any information of Sellers or any of their respective Affiliates not related to the Business or not related to the employment of the Transferred Employees, Sellers may elect to redact those portions of such Records to the extent pertaining to such other information or, in Sellers' discretion, Sellers may deliver unredacted copies of such Records containing information not related to the Business or not related to the employment of the Transferred Employees but such information shall be subject to the confidentiality provisions of this Agreement, shall remain the property of the Sellers, and Buyer shall have no rights with respect to such information; provided, further, that Transferred Records shall not include any Tax Return or other Tax records, other than those (or portions thereof) relating exclusively to the Business, the Transferred Assets or the Assumed Liabilities.
 - "Transition Services Agreement" means the transitional services agreement substantially in the form attached to this Agreement as Exhibit G.
- "<u>Virgin Mobile Business</u>" means the prepaid wireless business conducted by the Company and its Subsidiaries under the Virgin Mobile brand, but excluding the prepaid wireless business conducted by the Company under the Assurance Lifeline brand and excluding, for the avoidance of doubt, the prepaid wireless customers of each of Shentel and Swiftel.

- "<u>Willful Breach</u>" means an intentional and willful material breach, or an intentional and willful material failure to perform, in each case that is the consequence of an act or omission by a Party with the actual knowledge that the taking of such act or failure to take such act would cause a breach of this Agreement.
 - 1.2 <u>Table of Defined Terms</u>. A table of defined terms is set forth in <u>Annex A</u>.
 - 1.3 Principles of Interpretation. As used in this Agreement:
 - (a) The meanings of the defined terms are applicable to both the singular and plural forms thereof;
- (b) Reference to any Person includes such Person's successors and permitted assigns but, if applicable, only if such successors and assigns are not prohibited by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually;
 - (c) Reference to any gender includes each other gender;
- (d) Reference to any agreement, document or instrument means such agreement, document or instrument as amended or modified and in effect from time to time in accordance with the terms thereof;
- (e) The terms "hereunder," "hereof," "hereto" and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Article, Section or other provision thereof;
 - (f) Unless the context indicates otherwise, the term "or" is used in the inclusive sense of "and/or";
 - (g) Relative to the determination of any period of time, "from" means "from and including" and "to" means "to but excluding";
- (h) References to documents, instruments or agreements shall be deemed to refer as well to all addenda, Exhibits, Schedules or amendments thereto;
 - (i) The phrase "to the extent" shall mean the degree to which a subject or other theory extends and such phrase shall not mean "if";
 - (j) The word "will" shall be construed to have the same meaning and effect as the word "shall";
- (k) The Preamble, schedules and exhibits of this Agreement form an integral part thereof, and any reference to this Agreement shall include the Preamble, schedules and exhibits thereof;
- (l) The headings used in this Agreement have been adopted by the Parties for ease of reference only and shall not influence the meaning or interpretation of this Agreement;

- (m) Except as specified otherwise or if the context otherwise requires, references to Articles, Sections, Schedules and Exhibits are made to the articles, sections, schedules and exhibits of this Agreement;
- (n) Any phrase introduced by the terms "including", "include", "in particular" or any similar expression shall be construed as illustrative and "without limitation" and shall not limit the sense of the words preceding those terms; and
- (o) The terms "furnished," "provided," "made available" and words of similar import shall refer to those items so provided no later than one (1) Business Day prior to the date of this Agreement.
 - (p) References to "§", "Dollar" or "dollar" are references to the lawful currency of the United States of America.

ARTICLE 2 PURCHASE AND SALE

2.1 Purchase and Sale.

- (a) Upon the terms and subject to the conditions of this Agreement, at the Closing, the Sellers shall, or shall cause their respective Subsidiaries to, sell, assign, transfer, convey and deliver to the Buyer, and the Buyer shall purchase, acquire and accept from Sellers and their Subsidiaries all the right, title and interest of the Company and its Subsidiaries, free and clear of all Encumbrances, other than Permitted Encumbrances, in, to and under (i) all assets held by the Company and its Subsidiaries exclusively used in the Business, other than the Specified Assets, and (ii) the Specified Assets (collectively, the "Transferred Assets"), for (x) the Purchase Price and (y) the Buyer's assumption of all of the Assumed Liabilities to be paid, performed and discharged when due.
- (b) Upon the terms and subject to the conditions set forth in this Agreement, prior to the Closing, the Sellers may, and may cause their respective Subsidiaries to, make such contributions, transfers, assignments and acceptances, such that, upon the consummation of such contributions, transfers, assignments and acceptances, the Sellers or their respective designees shall own the Excluded Assets and shall be responsible for the Excluded Liabilities, without further recourse to the Buyer. In furtherance of the foregoing, (i) the Sellers or their respective designees shall retain, and the Buyer shall not acquire, any interest in the Excluded Assets, and (ii) the Sellers or their respective designees shall retain, and the Buyer shall not assume, any Excluded Liability.

2.2 Excluded Assets.

- (a) Notwithstanding anything to the contrary set forth herein, the Transferred Assets shall not include any Excluded Assets.
- (b) "Excluded Assets" means all assets and properties of the Sellers and their respective Affiliates that are not Transferred Assets, including, in each case with respect to the Sellers and their respective Affiliates:
 - (i) all cash and cash equivalents of the Sellers and their respective Affiliates;

- (ii) all cash and other assets of or relating to any employee benefit plan, program or arrangement or related trust in which any employees of the Sellers or their respective Affiliates participate (including the Plans and related trusts), other than the Retention Agreements;
- (iii) all rights, claims and credits (including all guarantees, warranties, indemnities and similar rights) of the Sellers or any of their respective Affiliates to the extent relating to any Excluded Asset or any Excluded Liability, including any and all such rights, claims and credits arising under insurance policies in favor of the Sellers and their respective Affiliates relating to any Excluded Asset or any Excluded Liability;
- (iv) any Tax assets (including any refund or credit of Taxes, and including any interest paid or credited with respect thereto) of any Seller or their respective Affiliates, or otherwise relating to the Transferred Assets or the Business attributable to any taxable period (or portion thereof) ending on the Closing Date;
- (v) all insurance policies and insurance contracts insuring the Business or the Transferred Assets, other than as expressly included in the Specified Assets;
- (vi) all rights of the Sellers and their respective Affiliates under this Agreement and the other agreements and instruments executed and delivered in connection with this Agreement (including the Ancillary Agreements);
- (vii) all proprietary materials used for the Sellers' or their respective Affiliates' human resource program and supporting documentation thereto;
- (viii) other than the Transferred Intellectual Property, all other Intellectual Property owned or held for use by any Seller or its Affiliates and its and their Third Party licensors;
 - (ix) other than the Transferred Contracts, all Contracts to which any Seller or its Affiliates is a party;
 - (x) all real property held or leased by any Seller or any of its Affiliates;
- (xi) all Rejected Inventory and all purchase orders or other commitments (including funding commitments from handset manufacturers) to the extent such order or commitment contemplates the acquisition of Rejected Inventory;
 - (xii) any Counsel Communications;
- (xiii) the following Records: (1) personnel Records for employees who are not Transferred Employees, (2) Records to the extent relating to any Excluded Asset or Excluded Liability, (3) Tax Returns or work papers relating to Tax or copies thereof, (4) Records prepared in connection with the Transaction Process and the Transaction, including bids received from other parties and analyses relating to the Businesses, (5) Records in the possession and control of the Sellers or any of their respective Affiliates that are not Transferred Records, and (6) copies of the Records (including otherwise Transferred Records) retained by the Sellers or their respective Affiliates in accordance with *bona fide* document retention policies or in order to perform their respective obligations under the Transition Services Agreement or MNSA;

- (xiv) all tangible and intangible wireless network and spectrum assets;
- (xv) all assets to the extent relating to (1) the prepaid wireless business conducted by the Company under the Assurance Lifeline brand and (2) the prepaid wireless customers and distributors of each of Shentel and Swiftel;
- (xvi) all furniture, fixtures, furnishings, equipment, machinery, leasehold improvements, tools and other tangible personal property (other than Inventory), wherever located (including any of the foregoing purchased subject to any conditional sales or title retention agreement in favor of any other Person), other than as expressly included in the Specified Assets;
- (xvii) other than the Transferred Intellectual Property, all computers, computer Software, hardware, servers, workstations, routers, hubs, switches, data communications lines, firmware, networks, and all other information technology equipment, other than as expressly included in the Specified Assets;
- (xviii) all claims of the Company or any of its Affiliates under any Contract in respect of breaches by any party to such Contract (other than the Company and its Affiliates) to the extent such breaches occurred prior to the Closing; and
 - (xix) all Contracts that are retail installment agreements for handsets entered into by the Company or any of its Affiliates.

2.3 <u>Assumption of the Liabilities</u>.

- (a) Upon the terms and subject to the conditions of this Agreement, effective as of the Closing, but subject to <u>Section 2.3(b)</u>, the Buyer shall assume and become responsible to perform, discharge and pay when due, all of the (i) Liabilities arising out of or relating to the ownership, operation or use of the Transferred Assets or the operation or conduct of the Business from and after the Closing Date but solely to the extent such Liabilities actually arise out of or relate to the ownership, operation or use of the Transferred Assets or the conduct of the Business on or after the Closing Date, and (ii) the Specified Liabilities (collectively, the "Assumed Liabilities"); provided, however, that Taxes shall be Assumed Liabilities only to the extent such Taxes are (i) described in sections (d) and (e) of the definition of Specified Liabilities contained herein, or (ii) Assumed Liabilities pursuant to <u>Section 7.3</u>.
- (b) Notwithstanding any other provision of this Agreement, the Buyer shall not assume any Excluded Liability, each of which shall be retained and paid, performed and discharged when due by the Sellers and their respective Affiliates. The term "Excluded Liability" means

(i) any and all Liabilities of the Sellers and their respective Subsidiaries other than the Assumed Liabilities, including the following (except to the extent constituting Specified Liabilities):

- (1) all Liabilities arising out of or relating to the ownership, operation or use of the Transferred Assets or the conduct of the Business before the Closing Date;
- (2) all Indebtedness (other than Indebtedness to the extent actually included in the calculation of Net Working Capital set forth in the Final NWC Statement);
- (3) all Liabilities to the extent arising out of or relating to any Excluded Asset;
- (4) all Transaction Expenses;
- (5) all Taxes of the Sellers and their respective Affiliates that are attributable to taxable periods (or portions thereof) ending on or prior to the Closing Date (determined, as applicable, in accordance with <u>Section 7.3</u>), any income Taxes of Sellers and their respective Affiliates triggered on the sale of the Transferred Assets, and any Transfer Taxes for which Sellers are liable pursuant to <u>Section 7.4</u>;
- (6) all Liabilities arising out of or relating to the employment of any Business Employee (or any dependent or beneficiary of any Business Employee) by the Sellers and/or their Affiliates, to the extent arising out of events occurring prior to the Closing Date, except to the extent that any such Liabilities are expressly assumed by the Buyer under Article 6;
- (7) the Liabilities set forth on Schedule 2.3(b)(i)(7); and
- (8) all Liabilities associated with purchase orders and accounts payable or other commitments to the extent such purchase orders or other commitments contemplate the acquisition of Rejected Inventory; and
- (ii) all Liabilities to the extent relating to Taxes attributable or imposed on the Business or the Transferred Assets for any period (or portion thereof) ending on or prior to the Closing Date, but excluding (A) Liabilities to the extent relating to Taxes that are Assumed Liabilities pursuant to Section 7.3 and (B) any Transfer Taxes for which Buyer is liable pursuant to Section 7.4.
- (c) Each of the Buyer's and the Sellers' obligations under this <u>Section 2.3</u> shall not be subject to offset or reduction by reason of any actual or alleged breach of any representation, warranty or covenant contained in this Agreement or any Ancillary Agreement or any right or alleged right to indemnification pursuant to <u>Article 13</u>.

2.4 <u>Purchase Price</u>. The purchase price to be paid by the Buyer in consideration for all of the Sellers' and their Affiliates' right, title and interest in, to and under the Transferred Assets (the "<u>Purchase Price</u>") shall be \$1,400,000,000, plus the Net Working Capital as of 12:01 a.m. Eastern Time on the Closing Date (as determined by <u>Section 2.5</u> below, which may be a negative number).

2.5 Certain Adjustments to Purchase Price.

- (a) Estimated NWC Statement . The Sellers shall deliver to Buyer no later than ten (10) Business Days after the Closing Date a closing statement (the "NWC Statement") signed by an appropriate officer of TMUS or its Subsidiaries setting forth Sellers' good faith calculation of the amount of Net Working Capital as of 12:01 a.m. Eastern Time on the Closing Date, which shall take into account, and set forth as separate line items, all provisions establishing the basis for such calculation of the Net Working Capital, in each case together with supporting documentation used by the Sellers in calculating such amounts. The NWC Statement (including the calculations therein) shall be prepared in accordance with the Agreed Accounting Principles and the Example Net Working Capital Statement; provided, that Net Working Capital shall also be calculated in accordance with the definition of that term herein (and any other defined terms incorporated therein).
- (b) Reasonable Access. From and after the Closing, the Sellers shall, and shall cause their Affiliates to, on reasonable prior notice to the Sellers and subject to the execution of customary work paper access letters if requested by accountants of the Sellers, (i) provide the Buyer and its Representatives with reasonable access during normal business hours to the facilities, Records and work papers of the Business and (ii) cooperate with and assist the Buyer and its Representatives in connection with the review of such materials, including by making available their employees, accountants and other personnel to the extent reasonably requested, in each case in connection with the Buyer's review of the NWC Statement; provided, that, in the event that Sellers do not provide such access or cooperation reasonably requested by the Buyer or any of its Representatives within two (2) Business Days of any request therefor (or, such shorter period as may remain in the Review Period), the Review Period will be extended by one (1) Business Day for each additional day required for Sellers to fully respond to such request.

(c) Notice of Objection.

(i) If the Buyer has any objections to the NWC Statement, it shall deliver to the Sellers a written statement (a "Notice of Objection") setting forth in reasonable detail the particulars of such disagreement (including the specific items in the NWC Statement that are in dispute and the nature and amount of any disagreement so identified) not later than thirty (30) days after the Buyer's receipt of the NWC Statement (such thirty (30)-day period, subject to adjustment pursuant to Section 2.5(b) above the "Review Period"). If the Buyer fails to deliver a Notice of Objection within the Review Period (or applicable later date if such period is tolled), the NWC Statement and the amount set forth therein shall be deemed to have been accepted by the Buyer and shall be deemed final and binding upon all of the Parties. If the Buyer delivers a Notice of Objection to the Sellers within the Review Period, the Sellers and Buyer shall work in good faith to resolve the Buyer's objections within the thirty (30)-day period following the delivery of the Notice of Objection.

- (ii) If the Buyer has no objections to the NWC Statement, the Buyer may deliver to the Sellers a written statement stating that it has no objections to the NWC Statement during the Review Period (such notice, a "Notice of No Objections"). In the case that the Buyer delivers a Notice of No Objection, there shall be no Review Period and the NWC Statement delivered by the Sellers shall be deemed the Final NWC Statement.
- (d) Resolution of Dispute. In the event that the Sellers and the Buyer are unable to resolve in writing any of the Buyer's objections in the Notice of Objection within the thirty (30)-day period (or such longer period as may be agreed by the Buyer and the Sellers) following the delivery of a Notice of Objection, the resolution of all unresolved items ("Disputed Items") shall be submitted to Ernst & Young LLP (or such other independent accounting firm of recognized national standing in the United States as may be mutually selected by the Buyer and the Sellers) (such accounting firm, the " Accounting Firm") to resolve any remaining disagreements, provided, however, that if Ernst & Young LLP would not have a substantial conflict at the time of its appointment due to one or more accounting relationships with the Buyer or the Sellers, the Buyer and the Sellers shall mutually select an alternative independent accounting firm of recognized national standing in the United States. The Buyer shall execute any agreement reasonably required by the Accounting Firm for its engagement hereunder. The Sellers shall, promptly (but in any event within ten (10) Business Days) following the formal engagement of the Accounting Firm, provide the Accounting Firm (copying the other upon submission) with a written presentation setting forth its calculations of and assertions regarding the Disputed Items and shall allow the Accounting Firm to conduct an independent analysis and audit of the Disputed Items using the Agreed Accounting Principles and the Example Net Working Capital Statement and taking into account the definition of that term herein (and any other defined terms incorporated therein). The Accounting Firm shall be instructed to render its determination with respect to such Disputed Items as soon as reasonably practicable (which the Parties agree shall not be later than forty-five (45) days following the formal engagement of the Accounting Firm). The Parties agree that the purpose of the adjustment contemplated by this Section 2.5 is to measure the amount of Net Working Capital as of 12:01 a.m. Eastern Time on the Closing Date using the Agreed Accounting Principles and the Example Net Working Capital Statement and taking into account the definition of that term herein (and any other defined terms incorporated therein). The determination of the Accounting Firm, acting as an expert and not an arbitrator, with respect to any such disagreements shall be binding and final for purposes of this Agreement. The term "Final NWC Statement." as used in this Agreement shall mean the NWC Statement that is deemed final in accordance with Section 2.5(c) or the NWC Statement resulting from the determinations made by the Accounting Firm in accordance with this Section 2.5(d), as applicable. The term "Final Purchase Price" as used in this Agreement shall mean the Purchase Price plus the Net Working Capital (which may be a negative number) set forth in the Final NWC Statement.
- (e) <u>Accounting Firm Fees</u>. The Accounting Firm shall allocate its costs and expenses between Buyer and the Sellers based on the percentage of the aggregate contested amount submitted to the Accounting Firm that is ultimately awarded to the Buyer, on the one hand, or the Sellers, on the other hand, such that the Buyer bears a percentage of such costs and expenses equal to the percentage of the contested amount awarded to the Sellers and the Sellers bear a percentage of such costs and expenses equal to the percentage of the contested amount awarded to the Buyer.

- (f) <u>Adjustment Payments</u>. Within two (2) Business Days following the determination of the Final NWC Statement, (x) the Buyer or the Sellers shall make an additional payment in immediately available funds in respect of the Purchase Price (if applicable) as follows:
- (i) if the Net Working Capital reflected on the Final NWC Statement is a positive number, the Buyer shall pay the Sellers the amount of such Net Working Capital.
- (ii) if the Net Working Capital reflected on the Final NWC Statement is a negative number, the Sellers shall pay the Buyer the absolute value of such Net Working Capital; or
- (iii) if the Net Working Capital reflected on the Final NWC Statement is zero, then neither the Buyer nor Seller shall have any payment obligation pursuant to this <u>Section 2.5(f)</u>.

Any payment pursuant to this <u>Section 2.5(f)</u> shall be treated as an adjustment to the Purchase Price for all tax purposes. No Losses may be claimed under <u>Article 13</u> or otherwise by any Indemnified Party to the extent such Liabilities are reflected in the Net Working Capital set forth in the Final NWC Statement pursuant to this <u>Section 2.5</u>.

2.6 Allocation of the Purchase Price.

- (a) Within ninety (90) days after the final determination of the Final Purchase Price pursuant to Section 2.5, the Sellers will provide the Buyer with a statement (or statements) (the "Asset Acquisition Statement") with the Sellers' proposed allocation of the Final Purchase Price (plus any other amounts, including Assumed Liabilities, to the extent properly taken into account as consideration for applicable Tax purposes) among the Transferred Assets and, if applicable, the Ancillary Agreements and any other rights transferred hereunder or thereunder in accordance with Section 1060 of the Code (and any other applicable state, local or non-U.S. Law). The Buyer may, within thirty (30) days after receiving such Asset Acquisition Statement, propose to the Sellers in writing any changes to such Asset Acquisition Statement that are consistent with applicable Law (the "Allocation Notice of Objection"), and if the Buyer does not deliver such a Notice of Objection within such period, the Buyer shall be deemed to have accepted such proposed Asset Acquisition Statement and it shall become final and binding on the Parties. If the Buyer delivers a Notice of Objection, then the Buyer and the Sellers will endeavor in good faith to resolve any differences with respect to the Asset Acquisition Statement within thirty (30) days after the Sellers' receipt of the Notice of Objection. If the Buyer and the Sellers are unable to resolve such differences, the matters in dispute shall be resolved by the Accounting Firm, which determination by such Accounting Firm shall be consistent with this Agreement. The fees, costs and expenses of the Accounting Firm shall be borne by the Buyer and the Sellers in inverse proportion as they may prevail on matters resolved by the Accounting Firm, which proportionate allocations also shall be determined by the Accounting Firm at the time the determination of the Accounting Firm is rendered.
- (b) The Buyer and the Sellers agree that they shall each (and shall cause their respective Affiliates to) file all Tax Returns (including amended returns and claims for refunds) and information reports in a manner consistent with the Asset Acquisition Statement (as finalized pursuant to Section 2.6(a)); provided that nothing contained in this Section 2.6(b) shall prevent any Party (or their Affiliates) from settling, or require any of them to litigate any challenge, proposed deficiency, adjustment or other similar proceeding by any Governmental

Authority with respect to the Asset Acquisition Statement. Upon any adjustment to the Purchase Price in connection with an indemnification claim made pursuant to <u>Article 13</u>, the allocation described in the Asset Acquisition Statement (as finalized pursuant to <u>Section 2.6(a)</u>) shall be subject to adjustment in a manner consistent with <u>Section 2.6(a)</u>.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF THE SELLERS

Except as set forth in the Disclosure Schedules delivered by the Sellers to the Buyer immediately prior to the execution of this Agreement (the "Disclosure Schedules") (it being agreed that disclosure of any item in any Section or subsection of a Disclosure Schedule shall apply only to the corresponding Section or subsection of this Agreement and to any other Section or subsection of this Agreement to the extent that the relevance of such item is reasonably apparent on its face in the Disclosure Schedules), TMUS, with respect to itself (solely with respect to Sections 3.1, 3.2, 3.3, 3.17) and Sprint, with respect to itself, the Business, the Transferred Assets and the Assumed Liabilities, hereby represents and warrants to the Buyer as follows:

- 3.1 <u>Organization</u>. Each Seller is duly organized and validly existing under the laws of the jurisdiction of its organization. Each Seller or any applicable Affiliate of such Seller which has title to any asset reasonably expected to be a Transferred Asset is duly qualified and in good standing in each jurisdiction in which Transferred Assets held by it are currently used by such Seller or such Affiliate and where the ownership, leasing, operation or use of its assets requires such qualification. Each Seller or the applicable Affiliate of such Seller has all requisite corporate power and authority to own, lease, operate or use, as the case may be, the Transferred Assets.
- 3.2 <u>Authorization, Enforceability</u>. Each Seller has all requisite corporate power and authority to enter into, execute and deliver this Agreement and the Ancillary Agreements to which it is a party, to consummate the transactions contemplated by this Agreement and the Ancillary Agreements to which it is a party and to perform its obligations hereunder and thereunder. The execution and delivery by such Seller of this Agreement and the Ancillary Agreements to which it is a party, the performance by such Seller of its obligations under this Agreement and the Ancillary Agreements and the consummation by such Seller of the transactions contemplated hereby and thereby have been duly and validly authorized by all necessary corporate action on such Seller's part, including any consent or approval from DT or SoftBank necessary to authorize such execution, delivery or performance. This Agreement has been, and each Ancillary Agreement to which it is a party will be, duly and validly executed and delivered by such Seller and constitutes or will constitute, assuming the due authorization, execution and delivery by the other parties thereto, a valid and binding obligation of such Seller, enforceable against such Seller in accordance with their respective terms, except to the extent that enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other laws relating to or affecting creditors' rights generally and by general principles of equity (whether considered in a Proceeding in equity or at law) (the "Remedy Exceptions").
- 3.3 No Approvals or Conflicts. Assuming the Regulatory Approvals are sought and obtained in accordance with this Agreement, neither the execution and delivery by each Seller of this Agreement or the Ancillary Agreements to which it is a party, nor the consummation by such Seller or any of its Affiliates of the Transaction, will (a) conflict with or violate any provision of such Seller's Governing Documents, (b) subject to Section 14.12, conflict with

the consummation of the NTM Merger in accordance with its terms or conflict with, or result in any violation of or default (with or without notice, lapse of time, or both) under, or give rise to a right of termination, loss of rights, adverse modification of provisions, cancellation or acceleration of any obligation under, or result in the creation of any Encumbrance (other than Permitted Encumbrances) upon any of the Transferred Assets under any provision of the NTM Merger Agreement, (c) require on the part of such Seller any Permit, (d) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, or create in any Person any right to terminate or cancel any Transferred Contract under clauses (i) or (ii) of the definition of Transferred Contract, or (e) violate any Law or Governmental Order, other than, in the case of clauses (c), (d) and (e), any such conflict, breach, default, termination, cancellation, imposition or violation that, or notice, consent, waiver, or Permit, the failure of which to make or be obtained, would not, individually or in the aggregate, reasonably be expected to be material to the Business.

3.4 Litigation.

- (a) Except as set forth on Schedule 3.4 (a) and Proceedings related to the DOJ Final Judgment associated with the NTM Merger, as of the date of this Agreement, there are no Proceedings pending or, to the Knowledge of the Sellers, threatened in writing in respect of the Transferred Assets or the Business that (a) involves a claim (or series of related claims) in excess of \$1,000,000, or (b) involves a claim for injunctive relief which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect if decided adversely to the Business.
- (b) Neither Seller nor any of its Affiliates is a party to or subject to the provisions of any material judgment, order, writ, injunction, decree or award of any Governmental Authority with respect to the Transferred Assets or the Business other than those entered by Governmental Authorities in connection with the NTM Merger.
- 3.5 <u>Compliance with Law; Permits</u>. Except as set forth on <u>Schedule 3.5</u>, and excluding Tax items which are covered by <u>Section 3.9</u>, and except as would not be material to the Business or the Transferred Assets, taken as a whole:
- (a) the use of the Transferred Assets and the conduct of the Business by the Company or its Subsidiaries is, and since January 1, 2018 has been, in compliance with all applicable Laws and the Company is, and since January 1, 2018 has been, not in violation of any such Laws in respect of any Transferred Assets held by it:
- (b) no investigation or review by any Governmental Authority with respect to Sellers or any of their respective Affiliates is pending or, to the Knowledge of the Sellers, threatened, nor has any Governmental Authority indicated an intention to conduct the same, in each case, with respect to the Business or the Transferred Assets;
- (c) Sellers have not received any notice or communication of any material noncompliance with any applicable Laws, in each case, with respect to the Business or the Transferred Assets; and
- (d) no Permits (including Permits issued by the FCC) are required by the Company and its Subsidiaries for the conduct of the Business or the use of the Transferred Assets as used and conducted by or as anticipated to be used and conducted by the Company and its Subsidiaries in the Ordinary Course of Business.

3.6 Financial Information.

- (a) Set forth on Schedule 3.6 are (x) copies of the unaudited Net Working Capital accounts of the Business as of March 31, 2019 and (y) such additional unaudited financial information of the Business as set forth on Schedule 3.6 for the month ended March 31, 2019 and the 12-month period ended March 31, 2019 (the "Financial Information"). The Financial Information (i) has been prepared from, and is consistent with, the Records of the Business, and, except as noted therein, have been prepared in accordance with GAAP consistently applied, (ii) have been prepared on a basis consistent with prior accounting periods, (iii) present fairly, individually and in the aggregate, in all material respects the balances of the specified working capital accounts of the Business for the period covered thereby, (iv) reflect all of the Current Assets and Current Liabilities of the Business (each as defined herein) as of such date in the unaudited working capital accounts, and (v) reflect and fairly present, individually and in the aggregate, in all material respects (A) the total revenue generated, (B) services revenue generated, (C) equipment revenue generated, (D) expenses, (E) contribution margin and (F) subscribers, in each case, of the Business for each period for which such unaudited Financial Information was provided. There were no material changes in the method of application of the accounting policies of the Company as applied to the Business or changes in the method of applying the use of estimates in the preparation of the Financial Information, in each case, other than the Agreed Accounting Principles, as compared with the consolidated audited financial statements of the Company and its Subsidiaries. No balance sheet, income statement, statement of cash flows, statement of equity or other similar financial statement has been prepared in respect of the Business on a standalone basis, and the Buyer acknowledges that the Business has not been conducted on a standalone basis.
- (b) The Company's system of internal controls over financial reporting is sufficient to provide reasonable assurance (i) that transactions in respect of the Business are recorded as necessary to permit preparation of financial statements in conformity with GAAP, consistently applied, and to derive from such financial statements, financial statements in conformity with GAAP, (ii) that transactions in respect of the Business are executed only in accordance with the fiscal authorization policies of the Company and (iii) regarding prevention or timely detection of the unauthorized acquisition, use or disposition of the assets of the Business.
- (c) All accounts receivable that are, and all accounts receivable on the Closing Date that will be, reflected in the Financial Information are valid receivables and were incurred in the Ordinary Course of Business for *bona fide* products delivered or services rendered and, to the Knowledge of the Sellers, are current and collectible. No such accounts receivable (i) are subject to any pending or threatened set-off, discount or counterclaim, other than for which a reserve has been established on the financial statements of the Company or as set forth in <u>Schedule 3.6(c)</u>, or (ii) have been assigned or pledged to any Person.
- (d) Since March 31, 2019, the Inventory has been maintained in the Ordinary Course of Business. All such Inventories are owned free and clear of all Encumbrances (other than Permitted Encumbrances). All of the Inventories consist of, and all Inventories on the Closing Date will consist of, items of a quality usable or saleable in the Ordinary Course of Business (subject in each case to the extent of reserves established thereunder) and are and will be in quantities sufficient for the normal operation of the Business in the Ordinary Course of Business. The Inventories are, and as of the Closing Date will be, in good and merchantable condition. Schedule 3.6(d) sets out the Inventory of the Business as of May 31, 2019 by SKU and specifies the ability of each such SKU to be used on the NTM Network.

- (e) <u>Schedule 3.6(e)</u> sets forth, as of May 31, 2019, the total number of mobile telephone numbers maintained by the Company or any of its Subsidiaries and assigned to an end user of the prepaid mobile wireless voice, text or data services of the Company or any of its Subsidiaries in respect of (A) the Boost Business, (B) the Virgin Mobile Business and (C) the Sprint Prepaid Business.
- 3.7 No Undisclosed Liabilities. There are no Liabilities arising or resulting from the operation of the Business or the ownership of the Transferred Assets (whether accrued, absolute, contingent or otherwise) other than Liabilities that (i) are future executory Liabilities arising under any Transferred Contract (other than as a result of breach of contract, tort, infringement or violation of applicable Law), (ii) are set forth on Schedule 3.7, (iii) have not had, or would not, individually or in the aggregate, reasonably be likely to have, a Material Adverse Effect, or (iv) constitute Excluded Liabilities.
- 3.8 <u>Absence of Certain Changes</u>. Since March 31, 2019, (a) through the date hereof, the Company and its Subsidiaries have conducted the Business in the Ordinary Course of Business (b) there has not been, with respect to the Business or any Transferred Asset, any event which, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect and (c) through the date hereof, any action taken by the Sellers or any of their respective Affiliates that, if taken during the period from the date of this Agreement through the Closing Date, would constitute a violation or breach of or would otherwise require consent under <u>Section 5.1(b)</u> (other than clauses (ii), (v) and (vi) thereto)).

3.9 <u>Tax Matters</u>. Except as set forth on <u>Schedule 3.9</u>:

- (a) All material Tax Returns required to be filed by such Seller with respect to the Business or the Transferred Assets have been duly and timely filed (taking into account applicable extensions) and all such Tax Returns are true, correct and complete in all material respects.
- (b) Such Seller has timely paid (or caused to be paid) all Taxes that are due and payable (other than Taxes that are being contested in good faith), whether or not shown as due on any Tax Returns, in respect of the Business and the Transferred Assets.
- (c) All material Taxes required to be withheld and remitted by such Seller under any applicable law with respect to the Business or the Transferred Assets have been duly and timely withheld and remitted to the proper Taxing Authority.
- (d) There are no material Encumbrances on the Transferred Assets for any failure (or alleged failure) to pay Taxes (other than Permitted Encumbrances).
- (e) Such Seller is not engaged in any audit, examination or investigation by any Taxing Authority for which such Seller expects a material assessment. No Taxing Authority has asserted any material tax assessments, deficiencies or adjustments related to the Transferred Assets or the Business that has not since been resolved and paid in full.

- (f) No claim has been made in writing by any Taxing Authority in a jurisdiction where such Seller does not file Tax Returns that the Seller is or may be subject to taxation with respect to the Business or the Transferred Assets in that jurisdiction, which claim has not been resolved and, if applicable, paid.
- (g) No written agreement waiving or extending, or having the effect of waiving or extending, the statute of limitations or the period of assessment or collection of any material Taxes, in each case, is currently in effect with respect to the Business or the Transferred Assets.
- (h) Notwithstanding anything else in this Agreement, the representations and warranties made in this Section 3.9 (i) do not refer to any matters other than Tax matters that would give rise to an Encumbrance (other than a Permitted Encumbrance) on the Transferred Assets or a Tax for which the Buyer (or its Affiliates) would be liable and (ii) may not be relied upon with respect to any taxable periods (or portions thereof) beginning after the Closing Date or with respect to Tax positions taken by the Buyer or the Buyer's Affiliates.

3.10 Employee Matters.

- (a) Schedule 3.10(a) includes a correct and complete list, as of the date of this Agreement, of each material Plan.
- (b) <u>Schedule 3.10(b)</u> sets forth a correct and complete listing of all outstanding equity awards covering shares of Company common stock held by Business Employees as of June 12, 2019, setting forth the number of shares subject to each such equity award and the holder, grant date, vesting schedule (including whether the vesting will be accelerated by the execution of this Agreement or consummation of the transactions contemplated hereby) and exercise price with respect to each such equity award, as applicable.
- (c) Except as set forth in <u>Schedule 3.10(c)</u> or as, individually or in the aggregate, would not be reasonably expected to have a Material Adverse Effect, no Plan is (i) a "multiemployer plan" within the meaning of Section 3(37) of ERISA, (ii) a "multiple employer plan" within the meaning of Section 4063 of ERISA or (iii) a defined benefit pension plan that is subject to Title IV of ERISA.
- (d) Except as set forth in Schedule 3.10(d), no Plan provides for the payment of any compensation or benefits to any Business Employee solely as a result of the consummation of the Transaction.
- (e) Each Plan that is intended to be qualified within the meaning of Section 401(a) of the Code (i) has received a favorable determination letter or has received a favorable opinion letter from the Internal Revenue Service (the "IRS") as to its qualification, or (ii) has time remaining under applicable Laws to apply for a determination or opinion letter, and to the Knowledge of the Sellers, nothing has occurred that would adversely affect the qualification of any such Plan.
- (f) Except as required under Section 4980B of the Code and other than continued coverage through the end of the month in which a termination of service occurs, no Plan provides retiree or post-employment medical, disability, life insurance or other welfare benefits to any Business Employee.

- (g) The Sellers are not a party to or bound by any collective bargaining agreement with respect to the Business. Except as would not reasonably be expected to result in any material liability to the Business, (i) the Business has not experienced any strike, lockout, organized labor slowdown or work stoppage within the past three (3) years, (ii) no formal complaint against any Seller is pending or, to the Knowledge of the Sellers, threatened before the National Labor Relations Board, the Equal Employment Opportunity Commission or any similar Governmental Authority by or on behalf of any Business Employee, and (iii) to the Knowledge of the Sellers, no organizational effort is presently being made or threatened by or on behalf of any labor union with respect to the Business Employees.
 - 3.11 Real Property. There is no leased or owned real property included in the Transferred Assets held by the Company or any of its Subsidiaries.
 - 3.12 Intellectual Property.
- (a) <u>Schedule IV</u> sets forth a true and complete list of all Registered Intellectual Property included in the Transferred Intellectual Property, indicating for each item the registration or application number, the registration or application date, and the applicable filing jurisdiction. The Company and its controlled Affiliates exclusively own all, right, title and interest in all the Transferred Intellectual Property, free and clear of all Encumbrances (other than Permitted Encumbrances). The Sellers are not bound by any outstanding judgment, injunction, order or decree or any contractual or other obligation materially restricting the use by the Company of the Transferred Intellectual Property, or materially restricting the licensing thereof to any Person. With respect to the Registered Intellectual Property listed on <u>Schedule IV</u>, (i) all such Intellectual Property is subsisting and in full force and effect and, to the Knowledge of the Sellers, valid and enforceable; (ii) the Company or one or more of its Subsidiaries is the owner of record and has paid all maintenance fees and made all filings that are required to be made prior to the Closing Date to maintain Seller's ownership thereof.
- (b) The conduct of the Business as currently conducted does not infringe, misappropriate, dilute or otherwise violate, and in the past three (3) years has not infringed, misappropriated or otherwise violated, any Intellectual Property rights of any Third Party, except for such infringements, misappropriations or other violations that would not reasonably be expected to have a Material Adverse Effect. No Proceedings are pending and no written notices have been received by the Seller during the past three (3) years (or earlier, if presently not resolved), in each case, alleging any infringement, misappropriation or other violation by the Sellers or any of their Subsidiaries of the Intellectual Property rights of any Third Party, except for such infringements, misappropriations or other violations that would not reasonably be expected to have a Material Adverse Effect. Except as set forth in Schedule 3.12(b), to the Knowledge of the Sellers, during the past three (3) years (or earlier, if presently not resolved) no person or entity has infringed, misappropriated, diluted or otherwise violated any of the Transferred Intellectual Property, except for such infringements, misappropriations, dilutions or other violations that would not reasonably be expected to have a Material Adverse Effect, and none of the Company or any of its Subsidiaries has made or asserted any claim, demand or notice against any person or entity alleging any such infringement, misappropriation, dilution or other violation. During the past three (3) years, there has been no litigation, opposition, cancellation, Proceeding, or claim pending, asserted or threatened concerning the ownership, or the right to use any Transferred Intellectual Property, or the validity, registrability, or enforceability of any Registered Intellectual Property included in the Transferred Intellectual Property, except where such litigation, opposition, cancellation, Proceeding, or claim would not, individually or in the aggregate, reasonably be expected to have a Material

- (c) The Sellers have taken all commercially reasonable measures to protect the confidentiality of all Trade Secrets included in the Transferred Intellectual Property and such Trade Secrets have not been disclosed by the Sellers to any Person except pursuant to written non-disclosure agreements that, to the Knowledge of the Sellers, have not been breached by such Person.
- (d) The Sellers have obtained from each Person (including current and former employees and independent contractors) who has created or developed for or on behalf of Sellers or any of their Subsidiaries any Transferred Intellectual Property that is material to the Business a written, present and valid assignment of such Intellectual Property to the Sellers or their Subsidiaries.
- (e) To the Knowledge of the Sellers, in the past three (3) years, there has been no material unauthorized access to or material unauthorized use of any confidential or proprietary information or data that is both in the Sellers' or any of their Subsidiaries' possession or control and material to the Business.
- (f) With respect to any Software included within Transferred Intellectual Property, (i) such Software is free from any material bugs, viruses or other malicious code, (ii) the source code for such Software has not been disclosed to any Third Party, and (iii) such Software does not contain, derive from or link to any open source Software in a manner that requires the disclosure of any source code, limits the ability to charge fees, or grants any license to any Third Party to make derivative works.
- (g) Neither the execution, delivery or performance of this Agreement, nor the consummation of the transactions contemplated by the Ancillary Agreements will result in the loss or impairment of any of the Transferred Intellectual Property.

3.13 Data Protection.

- (a) During the past three (3) years, the Sellers and their Subsidiaries have (i) complied in all material respects with all publicly-facing statements or policies, contractual obligations, and all applicable Laws, in each case, regarding privacy, cyber security, data security and the collection, retention, protection, transfer, use and processing of Personal Information relating to the Business except as, individually or in the aggregate, would not reasonably be expected to be material to the Business, and (ii) implemented and maintained commercially reasonable administrative, technical and physical safeguards designed to protect Personal Information relating to the Business against unauthorized access, use, loss and damage.
- (b) During the past three (3) years, (i) to the Knowledge of the Sellers, there has been no unauthorized access to, or misuse of, any Personal Information relating to the Business maintained by or on behalf of the Sellers or any of their Subsidiaries, in each case, except as, individually or in the aggregate, would not reasonably be expected to be material to the Business, and (ii) no Person (including any Governmental Authority) has made any claim or commenced any Proceeding with respect to any unauthorized access to, or misuse of, any Personal Information relating to the Business maintained by or on behalf of the Sellers or any of their Subsidiaries, except as, individually or in the aggregate, would not reasonably be expected to be material to the Business.

3.14 Transferred Contracts.

- (a) <u>Schedule 3.14(a)</u> of the Disclosure Schedules sets forth a list of each Transferred Contract of the types set forth in clauses (i) and (ii) of the definition of Transferred Contract that is of the type set forth below as of the date hereof (each Transferred Contract required to be set forth on <u>Schedule 3.14(a)</u>, a "<u>Material Contract</u>"):
 - (i) each Contract (or group of related Contracts with respect to a single transaction or series of related transactions), other than Contracts with suppliers or distributors, that involves payments, performance or services or delivery of goods or materials to or by either Seller or any of its Affiliates of any amount or value reasonably expected to exceed \$1,000,000 in any future 12 month period;
 - (ii) each Contract or series of Contracts with a Principal Supplier and with a Principal Distributor;
 - (iii) each Contract of Sellers or any of their respective controlled Affiliates with respect to material Transferred Intellectual Property, including (A) any licenses or other rights granted to any Person with respect to Transferred Intellectual Property, and (B) agreements involving Software (other than non-exclusive license agreements entered into in the Ordinary Course of Business);
 - (iv) each Contract concerning the establishment or operation of a partnership, strategic alliance, joint venture, or limited liability company or other similar agreement or arrangement;
 - (v) each Contract that limits or purports to limit the freedom of either Seller or any of its Affiliates to compete in any line of business with any Person or engage in any line of business within any geographic area or acquire the assets or securities of another Person or, or otherwise materially restricts Sellers' or any of their respective Affiliates' ability to solicit or hire any Person or solicit business from any Person, and each Contract that could require the disposition of any material assets or line of business of either Seller or any of its Affiliates;
 - (vi) each Contract related to an acquisition, divestiture, merger or similar transaction that contains financial covenants, indemnities or other similar payment obligations (including "earn-out" or other contingent payment obligations) that are still in effect and would reasonably be expected to result in the receipt or making of future payments in excess of \$2,500,000;
 - (vii) each Contract that contains a put, call or similar right pursuant to which either Seller or any of its Affiliates could be required to purchase or sell, as applicable, any equity interests in or assets (in the case of assets, having a purchase price in excess of \$2,500,000) of any Person;
 - (viii) each Contract that provides for exclusive rights for the benefit of any Third Party, grants "most favored nation" status, contains minimum volume or purchase commitments, or requires the Company or any of its Subsidiaries to provide any minimum level of service, in each case which are, or in a manner that is, material to the Business, taken as a whole;

- (ix) each Contract providing for indemnification by either Seller or any of its Affiliates of any Person, except for Contracts entered into in the Ordinary Course of Business.
- (b) The Sellers have furnished or made available to the Buyer a complete and correct (including all amendments and supplements thereto) copy of each Material Contract. With respect to each Material Contract: (a) such Contract is the valid and binding obligation of the Company or one or more of its Subsidiaries and, to the Knowledge of the Sellers, of each other party thereto, is enforceable in accordance with its terms and is in full force and effect, except to the extent that enforceability may be limited by the Remedy Exceptions, and (b) none of such Seller, its applicable Affiliate or, to the Knowledge of such Seller, any other party thereto, is in material breach or violation of, or default under, any such Contract, or any event, occurrence or condition, which (after notice, passage of time or both) would constitute or give rise to any such breach, violation or default thereunder, and neither Seller has received any cure notice or other written communication alleging that Sellers are in default or breach of such Material Contract in respect of the Business.

3.15 Transferred Assets.

- (a) The Company or one or more of its Subsidiaries has, or as of immediately prior to the Closing will have, good and valid title to, all of the Transferred Assets, free and clear of all Encumbrances (other than Permitted Encumbrances).
- (b) The Transferred Assets that are physical assets (other than Inventory) (i) are in good operating condition and repair in all material respects, reasonable wear and tear excepted and (ii) have been operated and maintained in all material respects in accordance with industry standard. The Transferred Assets that will be owned, leased or licensed by the Buyer immediately following the Closing (not taking into account the exclusion of any Rejected Inventory), together with any services provided by the Sellers or their respective Affiliates pursuant to the Transition Services Agreement and the MNSA (a) will constitute all of the assets (tangible or intangible) that are necessary to or used in the conduct of the Business as it is conducted as of the Closing and (b) will permit Buyer to operate the Business in all material respects in the manner in which it is conducted as of the Closing.

3.16 Principal Suppliers and Principal Distributors.

- (a) <u>Schedule 3.16(a)</u> sets forth a complete and accurate list of (i) the ten (10) largest suppliers of the Business based on the consolidated cost of goods and services paid to such Persons by the Company and its Subsidiaries for the calendar year ended December 31, 2018 (each, a "<u>Principal Supplier</u>") and (ii) with respect to each Principal Supplier, the aggregate amounts paid to each such Principal Supplier for the calendar year ended December 31, 2018. The Company has not received any written notice from any Principal Supplier indicating that any such Person is ceasing, will cease or plans to cease dealing with the Company or its Subsidiaries.
- (b) <u>Schedule 3.16(b)</u> sets forth a complete and accurate list of (i) the ten (10) largest agents, dealers, resellers or other distributors of the Business based on the consolidated revenue paid to such Persons by the Company and its Subsidiaries for the calendar year ended December 31, 2018 (each, a "<u>Principal Distributor</u>") and (ii) with respect to each Principal Distributor, the aggregate amounts received by each such Principal Distributor for the calendar year ended December 31, 2018. The Company has not received any written notice from any Principal Distributor indicating that any such Person is ceasing, will cease or plans to cease dealing with the Company or its Subsidiaries.

- (c) No Principal Supplier or Principal Distributor has given any notice that such Principal Supplier or Principal Distributor, as applicable, intends to, or to the Knowledge of the Sellers, has otherwise threatened to, terminate its business with respect to the Business, or materially reduce the volume of business transacted with respect to the Business
- 3.17 <u>Transactions with Affiliates</u>. There are no Affiliate Transactions, and no Affiliate Transaction shall extend beyond the Closing Date. For purposes of this <u>Section 3.17</u>, an "<u>Affiliate Transaction</u>" means any Transferred Contract (excluding clause (iii) of the definition of Transferred Contract), the terms of which would be required to be disclosed under Item 404 of Regulation S-K promulgated under the Securities Act of 1933 and Securities Exchange Act of 1934 by Seller, between the Sellers or their Subsidiaries, on the one hand, and the Parent Affiliates, on the other hand, including any Transferred Contract with, in favor of or for the benefit of, or involving the making of any payment or transfer of assets to or guarantee of obligation of, any such Parent Affiliate. None of the Parent Affiliates owns or has title to any of the Transferred Assets or any asset that is exclusively used in the Business or any asset that would otherwise meet the definition of Transferred Asset (excluding clause (iii) of the definition of Transferred Contract) if it were held by the Company.
- 3.18 <u>Brokers</u>. Neither Seller nor any of its Affiliates has any liability or obligation to pay any fees or commissions, for which the Buyer or any of its Affiliates would have any responsibility or liability for, to any broker, finder or agent with respect to the transactions contemplated by this Agreement or any of the Ancillary Agreements.
- 3.19 No Other Representations and Warranties. Notwithstanding anything contained in Article 4 or any other provision of this Agreement, it is the explicit intent of each Party, and Sellers agree, that Buyer is not making and has not authorized any Person to make any representation or warranty whatsoever, express or implied, except those representations and warranties expressly set forth in Article 4.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer hereby represents and warrants to the Sellers as follows:

- 4.1 Organization. The Buyer is duly organized and validly existing under the laws of the jurisdiction of its organization.
- 4.2 <u>Authorization, Enforceability</u>. The Buyer has all requisite corporate power and authority to enter into, execute and deliver this Agreement and the Ancillary Agreements to which it is party, to consummate the transactions contemplated by this Agreement and the Ancillary Agreements to which it is a party and to perform its obligations hereunder and thereunder. The execution and delivery by the Buyer of this Agreement and the Ancillary Agreements to which it is a party, the performance by the Buyer of this Agreement and the Ancillary Agreements and the consummation by the Buyer of the transactions contemplated hereby and thereby have been duly and validly authorized by all necessary corporate action on the Buyer's part. This Agreement has been, and each Ancillary Agreement to which it is a party will be, duly and validly executed and delivered by the Buyer and constitutes or will constitute, assuming the due authorization, execution and delivery by the other parties thereto, a valid and binding obligation of the Buyer, enforceable against the Buyer in accordance with their respective terms, except to the extent that enforceability may be limited by the Remedy Exceptions.

- 4.3 No Approvals or Conflicts. Assuming the Regulatory Approvals are sought and obtained in accordance with this Agreement, neither the execution and delivery by the Buyer of this Agreement or the Ancillary Agreements to which it is a party, nor the consummation by the Buyer of the transactions contemplated hereby or thereby, will (a) conflict with or violate any provision of the Buyer's Governing Documents, (b) require on the part of the Buyer any Permit, (c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, or create in any Person any right to terminate or cancel any material contract to which it is a party, or (d) violate any Law or Governmental Order, other than, in the case of clauses (b), (c) and (d), any such conflict, breach, default, termination, cancellation, imposition or violation that, or notice, consent, waiver, the failure of which to make or be obtained, would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of the Buyer to consummate the transactions in accordance with the terms hereof ("Buyer Material Adverse Effect").
- 4.4 <u>Litigation</u>; <u>Regulatory Approvals</u>. There is no Proceeding pending or, to the Buyer's knowledge, threatened against the Buyer or any of its Affiliates that seeks to, and the Buyer is not subject to any Governmental Order which, individually or in the aggregate, would reasonably be expected to materially impair or delay the ability of the Buyer to effect the consummation of the transactions contemplated by this Agreement or any of the Ancillary Agreements or otherwise prevent Buyer from performing in all material respects its obligations under this Agreement or any of the Ancillary Agreements. To Buyer's knowledge, there is no fact relating to Buyer's or any of its Affiliates' respective businesses, operations, assets, financial condition or legal status, including any officer's, director's or current employee's status, that would reasonably be expected to materially impair or delay the ability of the Buyer to promptly obtain the Regulatory Approvals.
- 4.5 <u>Financing</u>. The Buyer has available on the date of this Agreement (through cash on hand and available credit), and will have available at the Closing, sufficient cash on hand to enable it to pay (a) the Purchase Price, (b) any amounts payable by the Buyer at the Closing pursuant to the Ancillary Agreements, and (c) any fees and expenses of the Buyer or its Affiliates relating to the transactions contemplated by this Agreement and the Ancillary Agreements. Buyer acknowledges and agrees that the Closing is not conditioned on the availability of any expected financing or financing arrangement.
- 4.6 <u>Brokers</u>. Neither the Buyer nor any of its Affiliates has any liability or obligation to pay any fees or commissions, for which the Sellers or any of their respective Affiliates would have any responsibility or liability for, to any broker, finder or agent with respect to the transactions contemplated by this Agreement or any of the Ancillary Agreements.
- 4.7 <u>Solvency</u>. Assuming (a) the satisfaction of the conditions to Buyer's obligation to consummate the Transaction and pay the Purchase Price and (b) the accuracy of the representations and warranties of Seller set forth in <u>Article 3</u> in all material respects, Buyer and its Subsidiaries, on a consolidated basis taken as a whole, will be Solvent immediately after the Closing.

- 4.8 No Other Representations or Warranties. Notwithstanding anything contained in Article 3 or any other provision of this Agreement, it is the explicit intent of each Party, and Buyer agrees, that the Sellers and their respective Affiliates are not making and has not authorized any Person to make any representation or warranty whatsoever, express or implied, except those representations and warranties expressly set forth in Article 3. The Buyer further acknowledges and agrees that none of Sellers, their respective Affiliates or any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding the Business, the Transferred Assets or the Transaction or the transactions contemplated by the Ancillary Agreements, and, none of the Sellers, any of their respective Affiliates or any other Person will have or be subject to any liability to Buyer or any other Person resulting from the distribution to the Buyer or its Representatives or the Buyer's use of information, including (a) any confidential information memoranda or management presentations distributed on behalf of the Sellers or other publications or data room information provided to the Buyer or its Representatives, or any other document, information or projection in any form provided to the Buyer or its Representatives in connection with the Transaction, or (b) the financial information, projections, forecasts, estimates, statements of intent, statements of opinion or other forward-looking statements relating to the Business or the Transferred Assets, in each case, in expectation or furtherance of the transactions contemplated by this Agreement and the Ancillary Agreements. The Buyer further understands and acknowledges that the Sellers have not provided, and will not provide, any financial statements relating to the Business (other than the Financial Information provided in accordance with Section 3.6, the Example Net Working Capital Statement and the NWC Statement).
- 4.9 No Reliance. In entering into this Agreement and acquiring the Transferred Assets from the Sellers, the Buyer expressly acknowledges and agrees that it is not relying on and specifically disclaims reliance on any statement, representation or warranty made by or on behalf of the Sellers, any of their Affiliates or any other Person, including those which may be contained in any presentation or similar materials containing information regarding the Business or any of the Transferred Assets or in any materials provided to the Buyer or any other Person during the course of its due diligence investigation of the Business and the Transferred Assets, other than those representations and warranties made solely by Sellers expressly set forth in Article 3. Except for the representations and warranties expressly set forth in Article 3, (i) the Sellers are selling the Transferred Assets to the Buyer "as is" and "where is" and with all faults, and make no warranty, express or implied, as to any matter whatsoever relating to the Business, the Transferred Assets, the Sellers or any other matter relating to the transactions contemplated by this Agreement or the Ancillary Agreements including as to (x) merchantability or fitness for any particular use or purpose, (y) the operation of the Business or the Transferred Assets after the Closing in any manner, or (z) the probable success or profitability of the Business or the Transferred Assets after the Closing in any manner, or (z) the probable success or

ARTICLE 5 PRE-CLOSING COVENANTS

5.1 Conduct of Business Prior to the Closing.

(a) From the date hereof to the earlier of the termination of this Agreement and the Closing, except as (w) set forth on Schedule 5.1(a), (x) required by applicable Law, (y) consented to in writing by the Buyer (which consent shall not unreasonably be withheld, delayed or conditioned), or (z) otherwise required by the terms of this Agreement, the Sellers shall (A) cause the Business to be conducted in the Ordinary Course of Business, including maintaining customary levels of Inventory, and (B) use their reasonable best efforts (1) to retain and solicit customers and to preserve the Business's relationships with agents, dealers, distributors, resellers, suppliers, management, employees and other Persons having business

relationships with the Business or in respect of the Transferred Assets and the Assumed Liabilities and (2) not taking any action that would reasonably be expected to decrease the percentage of NTM Network-compatible handsets sold by the Business (measured as a percentage of all handsets sold by the Business for the three month period ending May 31, 2019).

- (b) From the date hereof to the earlier of the termination of this Agreement and the Closing, except as (w) set forth on Schedule 5.1(a), (x) required by applicable Law, (y) consented to in writing by the Buyer (which consent shall not unreasonably be withheld, delayed or conditioned), or (z) otherwise required by the terms of this Agreement, no Seller shall, and shall not permit its controlled Affiliates to, do any of the following with respect to the Business, the Transferred Assets or the Assumed Liabilities:
- (i) undertake any merger, spin-off, contribution of assets or other form of similar reorganization with respect to the Transferred Assets, other than in connection with the NTM Merger, other than such transactions by and among the Sellers and their Affiliates;
- (ii) (A) increase the base salary or base wage, bonus, incentive or other compensation, pension, welfare, fringe or other benefits, severance or termination pay of any Transferred Employee, other than in the Ordinary Course of Business and, with respect to increases in connection with annual merit increases or promotions in the Ordinary Course of Business that apply to more than ten percent (10%) of all Transferred Employees, after giving prior notification to the Buyer or (B) terminate any Transferred Employee (other than for cause);
- (iii) sell, lease or otherwise dispose of any Transferred Assets which are material, individually or in the aggregate, to such Business, except for (A) sales of raw materials, work-in-process, finished goods, supplies, parts, spare parts and other Inventories in the Ordinary Course of Business, (B) sales, leases, or other dispositions of Intellectual Property in the Ordinary Course of Business or (C) assets that are obsolete or no longer used in such Business;
- (iv) waive, release, grant or transfer any right of material value solely to the extent relating any Transferred Asset or any Assumed Liability, other than rights that solely constitute Excluded Assets;
- (v) (A) subject to Section 5.4, amend any material term of, waive any material right under, or voluntarily terminate (other than upon expiration in accordance with its terms) any Material Contract, or (B) enter into any Contract that would be a Material Contract if entered into prior to or on the date hereof, other than, in the case of clause (B) in the Ordinary Course of Business;
- (vi) make, change or revoke any material Tax election, (B) change any material method or period of accounting for Tax purposes, (C) prepare any material Tax Return in a manner inconsistent with elections made, positions taken or methods used in preparing or filing similar Tax Returns in prior periods, or (D) agree to an extension or waiver of the statute of limitations with respect to the assessment or determination of material Taxes, in each case, to the extent such action could reasonably be expected to result in any material Liability for the Buyer or its Affiliates or otherwise materially and adversely affect the Business;

(vii) sell, license, or otherwise dispose of, or abandon, cancel, or allow to lapse or expire, any Transferred Intellectual Property, except for immaterial rights in, or immaterial registrations or applications for Transferred Intellectual Property or for non-exclusive licenses granted in the Ordinary Course of Business;

- (viii) (A) delay or accelerate collection of accounts receivable in advance of or beyond their regular due dates or the dates when the same would have been collected in the Ordinary Course of Business or (B) delay or accelerate payment of any accounts payable in advance of its due date or the date such liability would have been paid in the Ordinary Course of Business;
- (ix) fail to maintain Inventory, at levels (by device type) consistent with the Company's and its Subsidiaries' Ordinary Course of Business or otherwise vary any Inventory practices in any material respect from the Ordinary Course of Business of the Company and its Subsidiaries;
- (x) participate in activity of the type sometimes referred to as "trade loading" or "channel stuffing" or any other similar activity that reasonably could be expected to result in an increase, temporary or otherwise, in the demand for the Inventory offered by the Business prior to the Closing, including sales of Inventory (i) with payment terms longer than terms offered in the Ordinary Course of Business for such Inventory, (ii) at a greater discount from listed prices than offered for such Inventory in the Ordinary Course of Business, other than pursuant to a promotion of a nature previously used in the Ordinary Course of Business for such Inventory publicly announced prior to the Closing Date other than in the Ordinary Course of Business, (iv) in a quantity greater than the reasonable resale requirement of the particular customer, agent, dealer, reseller, distributor or wholesaler, or (vi) in conjunction with other material benefits to the customer, agent, dealer, reseller, distributor or breviously offered in the Ordinary Course of Business;
 - (xi) take any action set forth on Schedule 5.1(b)(xi); or
 - (xii) authorize or otherwise commit to take any of the actions above.
- (c) Nothing in this <u>Section 5.1</u> is intended to give the Buyer or any of its Affiliates, directly or indirectly, the right to control or direct the business or operations of either Business prior to the Closing, and prior to the Closing, the Sellers shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over their respective Businesses.
- (d) For the purpose of any consent which may be requested from the Buyer pursuant to Section 5.1(a) or Section 5.1(b), it is specifically agreed that failure by the Buyer to respond in writing (including by e-mail) to a written consent request made by or on behalf of a Seller within five (5) Business Days of the submission of such request pursuant to Section 14.11 shall be deemed for all intents and purposes under this Agreement as a grant of the requested consent.

5.2 Regulatory Approvals.

(a) In connection with the Transaction, each of the Buyer and the Sellers shall cooperate with the other Parties and shall use its reasonable best efforts to obtain the DOJ Consent, FCC 214 Approval and FCC MNSA Approval as promptly as practicable. The Buyer shall not take any action to impair or materially delay the other Regulatory Approvals.

- (b) Without limiting the generality of the foregoing, subject to the terms and conditions herein, the Parties shall use their respective reasonable best efforts to consummate and make effective the Transaction and to cause the conditions set forth in Section 9.1(a) to be satisfied as promptly as reasonably practicable after the date hereof, including (i) using reasonable best efforts to promptly obtain all necessary consents, approvals, waivers and authorizations of, actions or nonactions by, and promptly making all required filings and submissions, with the DOJ, the FCC or any other Governmental Authority regarding the Transaction, (ii) cooperating with the other Parties in promptly making all necessary filings and promptly seeking all consents, approvals, permits, notices or authorizations.
- (c) In furtherance of and without limiting the generality of the foregoing, the Parties shall cooperate and use their respective reasonable best efforts to file with the FCC, at an appropriate time following the date of the DOJ Final Judgment as mutually agreed by the parties (but, in any event, not earlier than ten (10) Business Days after the date of the DOJ Final Judgment or later than 65 calendar days after the date of the DOJ Final Judgment), (i) an application for approval under Section 214 of the Communications Act of 1934, as amended to assign the Business's U.S.-international telecommunications customer accounts to the Buyer (the "FCC 214 Application") and (ii) the MNSA for approval by the FCC. The Parties shall cooperate and use their respective reasonable efforts to obtain the FCC 214 Approval and FCC MNSA Approval and shall each bear their own costs for such preparation, filing and prosecution.
- (d) In furtherance of and without limiting the generality of the foregoing, the Parties shall, and shall cause their respective Affiliates to, (i) use their respective reasonable best efforts to (x) obtain the DOJ Consent and the approval of the Transaction by the DOJ as required by the DOJ or the DOJ Final Judgment, including promptly submitting to the DOJ all documents and information that the DOJ reasonably requests from a Party, whether written or oral, and if the DOJ notifies, orally or in writing, the Sellers, the Buyer or their respective Affiliates that this Agreement or the Transaction is not acceptable unless certain language of this Agreement is modified, the Buyer and the Sellers shall consider in good faith such modifications and shall not unreasonably withhold, condition or delay their agreement to such modifications, and (y) take promptly any and all steps necessary to avoid or eliminate each and every impediment and obtain all consents under any Antitrust Law that may be required by any Governmental Authority, in each case with competent jurisdiction or by any DOJ Final Judgment, so as to enable the Parties to close the Transaction as promptly as practicable, (ii) use their respective reasonable best efforts to (x) obtain the FCC MNSA Approval and FCC 214 Approval, and any state public utility commission approval of the Transaction required by Law, including promptly submitting to the FCC or relevant state public utility commission notifies, orally or in writing, the Sellers, the Buyer or their respective Affiliates that this Agreement or the Transaction is not acceptable unless certain language of this Agreement is modified, each of the Parties shall consider in good faith such modifications and, subject to Section 11.1(c)(iii), shall not unreasonably withhold, condition or delay their agreement to such modifications, and (y) take promptly any and all steps necessary to avoid or eliminate each and every impediment and obtain all consents regarding the Transaction that may be required by any Gov

as to enable the Parties to close the Transaction as promptly as practicable, (iii) vigorously contest (x) any civil, criminal or administrative Proceedings brought by any Governmental Authority or any other Person against such Party or its Affiliates seeking to enjoin, restrain, prevent, prohibit or make illegal the consummation of the Transaction or seeking damages or to impose any terms or conditions on the Transaction, and (y) any Governmental Order that enjoins, restrains, prevents, prohibits or makes illegal the consummation of any of the Transaction or imposes any damages, terms or conditions on the Transaction, and (iv) use their respective reasonable best efforts to (x) resolve any objections the DOJ, FCC or any other Governmental Authority may assert under any applicable Law with respect to the Transaction, and obtain the Governmental Authority approvals of the Transaction required by Law, and (y) resolve any objections that any Third Party may assert relating to the obtaining of any consent, approval, waiver or authorization required from such Third Party in connection with the Transaction; provided that notwithstanding anything in this Agreement to the contrary, the Parties shall not be obligated to (and without the other Parties' consent, a Party shall not with respect to the Business) agree to or accept any divestiture or other structural or conduct relief or new contractual commitments, including network or spectrum sharing arrangements, that were not previously agreed to by the Party as of the date hereof, in order to obtain the Regulatory Approvals or any approval of any other Governmental Authority in connection with the Transaction.

Notwithstanding anything to the contrary in this Agreement, nothing in this Section 5.2 or Section 14.2 shall be deemed to amend, modify or expand the obligations of TMUS or the Company under the NTM Merger Agreement.

- (e) Each Party shall promptly furnish to the other Parties copies of any notices, requests or written communications, and shall promptly describe to the Sellers any oral communications, received by such Party or any of its Affiliates or Parent Affiliates from any third party or any Governmental Authority with respect to the Transaction, and each Party shall permit counsel to the other Party an opportunity to review in advance, and the Buyer shall consider in good faith and reflect the views of such counsel in connection with, any proposed communications by the Buyer or its Affiliates to any Governmental Authority concerning the Transaction; provided, however that (i) materials may be redacted to remove references concerning the valuation, projections, business plans or prospects of the Transferred Assets or Business or as necessary to address reasonable attorney-client or other privilege concerns and (ii) this sentence shall not require any sharing of documents or information not permitted by the applicable Governmental Authority. The Buyer shall provide the Sellers the opportunity, on reasonable advance notice, to participate in any substantive meetings or discussions, either in person or by telephone, between the Buyer or any of its Affiliates, or their respective Representatives, on the one hand, and any Governmental Authority, on the other hand, concerning or in connection with the Transaction, where permitted by such Governmental Authority.
- (f) Without limiting the foregoing, the Parties shall regularly review with each other the progress of obtaining the Governmental Authority approvals of the Transaction and any notifications or filings (including, where necessary, seeking to identify appropriate commitments or submissions to address any concerns identified by any Governmental Authority) pertaining to such approvals and discuss with each other the scope, timing and tactics of any such commitments or submissions with a view to obtain approval of the Transaction by all relevant Governmental Authorities, at the earliest time possible. Buyer shall not take any action, or permit their respective Affiliates to take any action, that would reasonably be expected to materially delay or prevent the consummation of the Transaction, including with respect to the Financing. Sellers shall not take any action, or permit their

respective Affiliates to take any action, that would reasonably be expected to materially delay or prevent the consummation of the Transaction, except as may be permitted or required under the NTM Merger Agreement, <u>provided</u>, that any such actions taken under the NTM Merger Agreement would not materially and adversely affect the Transferred Assets or the Assumed Liabilities.

5.3 Third Party Consents.

- (a) Notwithstanding anything in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign, directly or indirectly, any asset (including any Contract), claim or right, or any benefit arising under or resulting from such asset, claim or right, if an attempted direct or indirect assignment thereof, without the consent or waiver of a Third Party (each, a "Consent"), would constitute a breach or other contravention of the rights of such Third Party, would be ineffective with respect to any party to a Contract concerning such asset, claim or right or would in any way adversely affect the rights of the Sellers or their respective Affiliates or, upon transfer, the Buyer under such asset, claim or right. If any direct or indirect transfer or assignment by the Sellers or their respective Affiliates to, or any direct or indirect assumption by the Buyer of, any interest in, or liability, obligation or commitment under, any asset, claim or right requires a Consent, then such transfer, assignment or assumption shall be made subject to such Consent being obtained, provided that to the extent such Consent is not able to be obtained in accordance with the provisions of this Section 5.3, the provisions of Section 12.3 shall apply to such asset, claim or right.
- (b) Following the date hereof until the Closing Date, the Parties shall cooperate with each other and use their respective reasonable best efforts, subject to and without limiting anything contained in this Agreement, to obtain all material Consents, including, in the case of the Company, using reasonable best efforts to transfer to Buyer all Transferred Contracts, including agency, dealer, reseller and other distribution agreements and all supplier and customer agreements that are Transferred Assets.
- (c) Neither the Sellers nor any of their respective Affiliates shall have any obligation to (i) pay any consideration to any Person for the purpose of obtaining any Consent, or (ii) pay any costs or expenses of any Person in connection with obtaining any Consent, unless all of such costs and expenses (if any) will be borne exclusively by the Buyer. Nothing in this provision shall affect any obligation in any Ancillary Agreement to the extent expressly provided therein.
- (d) The Buyer further expressly acknowledges that none of the Consents is a condition to the consummation of the Transaction and the Buyer acknowledges that no condition shall be deemed not satisfied as a result of (i) the failure to obtain any Consents, (ii) any termination of any Contract (including any Transferred Contract) as a result of the Transaction or (iii) any Proceedings commenced or threatened by any Person arising out of or relating to the failure to obtain any Consent or avoid any such termination, in each case, except to the extent resulting directly from a breach by the Sellers of (x) their express obligations under this Section 5.3 or (y) of any representation regarding non-contravention, including their representation set forth in Section 3.3.
- (e) Without limiting the foregoing, the Buyer shall provide each Seller (or their applicable Affiliates) such information and references (including regarding its creditworthiness) as may reasonably and timely be requested by any relevant Third Party for the purposes of obtaining the required Consents and shall enter into such undertakings or procure such guarantees in favor of any relevant Third Party as may be reasonably requested by such relevant Third Party for the purposes of obtaining the required Consents.

(f) The provisions of this <u>Section 5.3</u> shall not apply with respect to the Shared Contracts, with respect to which <u>Section 5.4</u> shall apply.

5.4 Shared Contracts.

- (a) Except as otherwise agreed by the Sellers and the Buyer or as otherwise expressly provided in this Agreement or the Ancillary Agreements (including with respect to any Contract identified by Buyer and Seller as reasonably necessary for Buyer to continue operation of the Business upon termination of any service provided under the Transition Services Agreement), until the expiration or termination date of the applicable Shared Contract (assuming, for these purposes, that the then-current term in effect as of immediately prior to the Closing is not renewed or extended), the Parties shall (and shall cause their Affiliates to) use reasonable best efforts to obtain or structure an arrangement for the Buyer to receive the rights and benefits, and bear the obligations and burdens, of the portion of such Shared Contract that the Buyer determines is reasonably necessary for Buyer to continue operation of the Business upon termination of any service provided under the Transition Services Agreement; provided, that the Sellers and their respective Affiliates shall not be required to take any action that would, in the good-faith judgment of the Sellers, constitute a breach or other contravention of the rights of any Person(s), be ineffective under, or contravene, applicable Law or any such Shared Contract or adversely affect the contractual rights of the Sellers or any of their respective Affiliates. The Buyer shall indemnify and hold harmless the Sellers and their respective Affiliates for and against all out-of-pocket Liabilities (including Tax Liabilities) arising out of or relating to each such arrangement. With respect to any Liability pursuant to, under or relating to any Shared Contract, such Liability shall be allocated between the applicable Seller, on the one hand, and the Buyer, on the other hand, as follows: (i) if a Liability is incurred solely in respect of the Business or the other businesses of such Seller, such Liability shall be allocated to the Buyer (to the extent it would otherwise constitute an Assumed Liability) or such Seller (to the extent it would otherwise constitute an Excluded Liability), and (ii) if a Liability cannot be so allocated under clause (i), such Liability shall be allocated to such applicable Seller, or the Buyer, as the case may be, based on the relative proportion of total benefit received by the Business (taking into account the extent to which such Liability would otherwise constitute an Assumed Liability or an Excluded Liability hereunder) and the other businesses of such Seller under the relevant Shared Contract, as reasonably determined by such Seller consistent with this Agreement. Notwithstanding the foregoing, each of the Sellers and the Buyer shall be responsible for any or all Liabilities arising from its (or its Affiliates') direct or indirect breach of any Shared Contract.
- (b) Nothing in this Section 5.4 shall be construed so as to require any of the Parties or their respective Affiliates to pay money to any Third Party, commence any litigation or offer or grant any accommodation (financial or otherwise) to any Third Party in connection with the separation or transfer of, or otherwise in respect of, any Shared Contract. For the avoidance of doubt, neither the Sellers nor the Buyer shall be required to provide credit support for the other Party in respect of such other Party's portion of a Shared Contract.
- 5.5 <u>Insurance</u>. Sellers shall, and shall cause their respective Affiliates to, assign, to the extent assignable, to Buyer any and all proceeds under any of either Seller's or any of its controlled Affiliates' Third Party insurance policies written prior to the Closing to the extent received in connection with (i) the damage or destruction of any of the Transferred Assets from

and after the date hereof and prior to the Closing that is, or would have been but for such damage or destruction, included in the Transferred Assets or (ii) any Assumed Liability with respect to a period prior to the Closing, other than Liabilities included in the calculation of Net Working Capital (other than, in the case of this clause (ii), where insurance proceeds are directly or indirectly funded by either Seller or any of its Affiliates through self-insurance or other similar arrangement). Sellers agree to use their reasonable best efforts to obtain any necessary consents or approvals of any insurance company or other Third Party relating to any such assignment, provided, however, that the Buyer shall promptly pay or reimburse the Sellers for all reasonable costs, fees and expenses (including legal expenses) incurred in connection with the obligations under this Section 5.5 and provided, further that the assignment of such proceeds shall be net of any insurance premiums, retroactive premium adjustments and any out-of-pocket costs or expenses associated with the collecting the such proceeds. If such proceeds are not assignable, Sellers agree to pay any such proceeds received by it or any of its controlled Affiliates to Buyer promptly upon the receipt thereof, net of any insurance premiums, retroactive premiums, premium adjustments and any out-of-pocket costs or expenses associated with the collecting such proceeds.

- 5.6 Exclusivity. From the date of this Agreement until the earlier of the Closing or the termination of this Agreement (the "Exclusivity Period"), the Sellers shall not, and shall cause their respective Subsidiaries and Representatives not to, directly or indirectly, discuss, pursue, solicit, initiate, participate in, facilitate, knowingly encourage or otherwise enter into any discussions, negotiations, agreements or other arrangements regarding or which could lead to, a possible sale or other disposition (whether by merger, reorganization, recapitalization or otherwise) of all or any part of the Business with any other Person other than the Buyer or its Subsidiaries (an "Acquisition Proposal") or provide any information to any Person other than the Buyer and its Subsidiaries and Representatives in connection with, or that would reasonably be expected to lead to, any Acquisition Proposal. During the Exclusivity Period, the Sellers will, and each will cause their respective Subsidiaries and Representatives to (A) immediately cease and cause to be terminated any existing discussions or negotiations with any Persons (other than Buyer) conducted heretofore with respect to any Acquisition Proposal, (B) terminate all physical and electronic data room access previously granted to any Person or its Representatives, and (C) promptly request each Person that has executed a confidentiality agreement in the last twelve (12) months with respect to an Acquisition Proposal to return or destroy all information heretofore furnished by such Person or its Representatives by or on behalf of the Sellers.
- 5.7 <u>Further Assurances</u>. After the date of this Agreement, but prior to the Closing, the Parties will cooperate to identify any other assets or liabilities, if any, that the Parties mutually agree are required for the operation of the Business and shall update Schedules I, II, III, IV and V to this Agreement accordingly.
- 5.8 <u>Access and Information</u>. From the date hereof until the earlier of the Closing Date and the termination of this Agreement in accordance with its terms, the Buyer and its Affiliates shall be entitled, including through its and their Representatives, to make such investigation of the Transferred Assets, Assumed Liabilities and services to be provided pursuant to the Transition Services Agreement and the MNSA and such examination of the Records to the extent related thereto, and to receive such information, including financial information (including the information set forth on Schedule VI), as it reasonably requests and to make extracts and copies of such Records, including reasonable access to customary supporting information, data and documentation necessary for the preparation of the Buyer's

financial reports. Any such investigation and examination shall be conducted after reasonable advance notice under reasonable circumstances and shall be subject to any restrictions under applicable Law and this Agreement. The Sellers shall, and shall cause their controlled Affiliates and Representatives to, cooperate with the Buyer and its Affiliates and their Representatives in connection with such investigation and examination, and the Buyer and its Affiliates and Representatives shall cooperate with the respective Representatives of the Sellers and shall use their reasonable best efforts to minimize any disruption to the business. This Section 5.8 shall not entitle Buyer or its Representatives to contact any Third Party doing business with Sellers, access the properties or Records of any such Third Party or access the properties of the Seller or its controlled Affiliates, in each case without Seller's prior written consent. No investigation or notice under this Section 5.8 or otherwise shall (i) alter any representation or warranty given hereunder by Sellers, any condition to the obligations of the Parties under this Agreement or Buyer's right to indemnification hereunder or (ii) modify any section of the Disclosure Schedules. Notwithstanding anything to the contrary set forth in this Section 5.8, in no event shall the Sellers be required to prepare any financial statements with regard to the Business, the Transferred Assets or the Assumed Liabilities, whether prior to or following the Closing.

5.9 Financing; Financing Cooperation.

- (a) The Buyer shall have sufficient funds available to it at the Closing to satisfy the payment of the Purchase Price in full. In the event that the Buyer determines, in its sole discretion, to arrange or obtain any financing to provide funds in connection with the transactions contemplated by this Agreement (the "Financing"), (i) the Buyer shall use, and shall cause its Affiliates to use, its and their reasonable best efforts to ensure that the Financing is available at Closing and (ii) the Sellers shall provide, and shall cause their respective controlled Affiliates and Representatives to provide to the Buyer reasonable assistance and cooperation as is reasonably requested by the Buyer in connection with arranging, obtaining and syndicating the Financing, as necessary, including assisting the Buyer with preparation of customary documents and other materials reasonably necessary in connection with the Financing. Notwithstanding anything to the contrary set forth in this Section 5.9, in no event shall the Sellers be required to prepare any balance sheet, cash flow statement, income statement or statement of stockholder's equity with regard to the Business, the Transferred Assets or the Assumed Liabilities, whether prior to or following the Closing.
- (b) Notwithstanding the foregoing, the Buyer agrees that (i) on the earlier of the Closing Date or the termination of this Agreement, the Buyer shall promptly reimburse the Sellers for all documented out-of-pocket Third Party costs and expenses incurred by the Sellers in connection with such cooperation pursuant to this Section 5.10; and (ii) the Buyer shall indemnify and hold harmless the Sellers and their respective Affiliates and their respective Representatives from and against any and all Liabilities, Losses, damages, claims, costs, expenses, interest, awards, judgments and penalties suffered or incurred in connection with the arrangement of the Financing or any assistance or activities provided in connection therewith. Notwithstanding anything to the contrary herein, it is understood and agreed that the condition precedent set forth in Section 9.2(b), as applied to the Sellers' obligations under this Section 5.9, shall be deemed to be satisfied unless the Financing has not been obtained as a direct result of the Sellers' Willful Breach of its obligations under this Section 5.9.

- (c) The Buyer understands and acknowledges that under the terms of this Agreement, the Buyer's obligation to consummate the transactions hereunder is not in any way contingent upon or otherwise subject to the Buyer's consummation of any financing arrangements, the Buyer's obtaining of any financing or the availability, grant, provision or extension of any financing to the Buyer. For the avoidance of doubt, if any such financing has not been obtained, the Buyer shall continue to be obligated, until such time as this Agreement is terminated in accordance with its terms and subject to the satisfaction or waiver of the conditions set forth in Article 9, to consummate the transactions contemplated by this Agreement.
- 5.10 Notice of Developments. Prior to the Closing and subject to applicable Laws and as required by any Governmental Authority, Sellers and Buyer shall each keep the other reasonably apprised of the status of matters relating to consummation of the Transaction, including promptly furnishing the other with copies of material notices or other material communications received by Buyer or Sellers, as the case may be, or any of their respective Subsidiaries, from any Third Party and/or any Governmental Authority with respect to the Transaction. Prior to the Closing, each Party shall notify the other Party in writing reasonably promptly after becoming aware of (a) any event, circumstance, fact or occurrence arising subsequent to the date of this Agreement which would result in any material breach of any representation, warranty or covenant of any Party in this Agreement or which could have the effect of making any representation or warranty of the any Party in this Agreement untrue or incorrect in any material respect or (b) any other material developments affecting the Transferred Assets or the Assumed Liabilities.

5.11 Additional Covenants.

- (a) Between the date hereof and the Closing, the Sellers will provide to the Buyer no later than ten (10) Business Days after the first (1 st) day of each calendar month, an Example Net Working Capital Statement as of the last day of the calendar month immediately preceding such month. Such monthly Example Net Working Capital Statements shall be prepared in accordance with the Agreed Accounting Principles and will be presented in a format that is substantially similar to the format of the Example Net Working Capital Statement.
- (b) Between the date hereof and the Closing, the Sellers will provide to the Buyer no later than ten (10) Business Days after the first (1 st) day of each calendar month, a list of Inventory identified by SKU and compatibility with the NTM Network as of the first (1 st) day of such calendar month, as well as a list of purchase orders and accounts payable outstanding (together with the amount of Inventory by SKU under each such purchase order and account payable outstanding). On or prior to the date that is five (5) days prior to the Closing Date, the Buyer shall provide written notice to the Sellers identifying the type of Inventory by SKU and compatibility with the NTM Network that the Buyer wishes to reject any or all Inventory of such type held by the Sellers as of the Closing Date (including the amount of such Inventory to be accepted) (and any or all purchase orders and accounts payable in respect of any Inventory by SKU) (such inventory and purchase orders, the "Rejected Inventory") shall be deemed rejected, shall not be included in the Transferred Assets and shall not be included for purposes of calculating the Net Working Capital as set forth in the NWC Statement. Following the Closing Date, the Buyer may purchase from the Sellers, at a price equal to the value that would have been set forth in the calculation of Net Working Capital had such Inventory not been deemed Rejected Inventory, all or a portion of Rejected Inventory, to the extent that the Seller has not already sold or committed to sell the Rejected Inventory to a Third Party.

- 5.12 <u>Certain Commercial Relationships</u>. Prior to the Closing Date, the Buyer and the Sellers shall discuss and negotiate in good faith such updates and modifications to the Annexes to the Transition Services Agreement as may be mutually agreed by the Parties; it being understood that to the extent that updates and modifications cannot be agreed, the Annexes in the form attached to this Agreement shall be the Annexes to the Transition Services Agreement to be executed by the parties thereto. The Parties shall take such actions set forth on <u>Schedule 5.12</u> with respect to certain commercial arrangements between the Parties.
- 5.13 <u>Cooperation Regarding Onboarding</u>. Subject to applicable Law, between the date of this Agreement and the earlier of the Closing Date and the termination of this Agreement, the Sellers shall use reasonable best efforts to cooperate with and make appropriate employees and contractors reasonably available to the Buyer to assist the Buyer, at the Buyer's expense in respect of actual out-of-pocket onboarding costs, to plan and implement necessary and appropriate policies, procedures and other arrangements in connection with the transition of ownership of the Business and onboarding to TMUS's billing and operational systems (as set forth in Section 4.1 to Annex 2 of the MNSA).

ARTICLE 6 EMPLOYEE MATTERS

6.1 Offer of Employment.

(a) No later than forty (40) days prior to the Closing Date, Sellers shall provide Buyer with a list of the annual base salaries or base wages (as applicable), commission opportunities and hire dates (or service recognition dates, if different) of all Business Employees. At least thirty (30) days prior to the Closing Date, the Buyer shall offer employment, effective as of the Closing Date, to each employee of the Business set forth on Schedule 6.1 (each, a " Business Employee") who is actively employed (including any Business Employee on vacation, holiday, jury duty, approved leave of absence or other similar absence) immediately prior to the Closing Date. For any Business Employee who is on short-term or long-term disability immediately prior to the Closing Date, at least thirty (30) days prior to the Closing Date, the Buyer shall offer employment to such Business Employee, effective as of the date on which such Business Employee returns to active employment following the Closing (provided that such return to active employment occurs within twelve (12) months following the Closing Date). In each case, such offer of employment by the Buyer shall provide, for at least the twelve (12) month period following the Closing Date, for (i) job responsibilities and duties that are substantially comparable to each such Business Employee's job responsibilities and duties immediately prior to the Closing Date, (ii) the compensation and benefits set forth in Section 6.3 and (iii) with a job location that is no more than 50 miles from the applicable Business Employee's job location immediately prior to the Closing, with such employment to be effective as of 12:01 a.m. (U.S. Eastern Time) on the Closing Date (or, for employees on short-term or long-term disability leave as of immediately prior to the Closing, such later date as described above). All such employees to whom the Buyer offers employment and who accept such employment are herein referred to as the " Transferred Employees". Effective as of 12:01 a.m. (U.S. Eastern Time) on the Closing Date (or, for employees on short-term or long-term disability leave as of immediately prior to the Closing, such later date as described above), each Transferred Employees shall terminate employment with the Sellers and any of their respective Affiliates, and, except as provided in Section 6.7, Sellers and their Affiliates shall be responsible for any severance or separation benefits that become due to any Transferred Employee as a result of his or her termination by Sellers or their Affiliates on the Closing Date.

- (b) No later than twenty (20) days prior to the Closing, Buyer shall provide Sellers with a list of the base salaries or base wages (as applicable) and commission opportunities set forth in Buyer's offers of employment to the Business Employees. In addition, Buyer shall promptly provide Sellers with written notice of any Business Employee who rejects Buyer's offer of employment (but in no event more than two (2) Business Days after Buyer receives notice of such rejection from such Business Employee).
- 6.2 <u>Seller Benefit Plans</u>. Effective as of 12:01 a.m. (U.S. Eastern Time) on the Closing Date (or, for employees on short-term or long-term disability leave as of the Closing, such later date as described above in <u>Section 6.1</u>), each Transferred Employee shall cease all active participation in and accrual of benefits under all Plans (other than the Retention Agreements). For purposes of <u>Sections 6.4</u> through <u>6.7</u> below, references to the Closing or the Closing Date shall mean, with respect to employees on short-term or long-term disability leave as of immediately prior to the Closing, such later date as described above in <u>Section 6.1</u>.
- 6.3 <u>Benefit Levels</u>. For a period of twelve (12) months following the Closing Date, the Buyer shall cause each Transferred Employee to be provided with (i) a base salary or base wage and commission opportunities that are no less favorable, in the aggregate, than those in effect immediately prior to the Closing Date, and (ii) other benefits that are no less favorable, in the aggregate, than the benefits provided to similarly situated employees of the Buyer or its subsidiaries as in effect from time to time. Without limiting the immediately preceding sentence (or any other provision hereof), for the twelve (12)-month period following the Closing Date, the Buyer shall provide, or cause to be provided, to each Transferred Employee severance payments and severance benefits that are no less favorable than the greater of (i) the severance payments and severance benefits that were provided to such employee as of immediately prior to the Closing Date and (ii) the severance payments and severance benefits that are provided to similarly-situated Buyer employees from time to time.
- 6.4 <u>Service Credit</u>. The Buyer shall cause each Transferred Employee to be credited with his or her years of service with Sellers and their Affiliates prior to the Closing for all purposes under each employee benefit plan, program, arrangement or policy established, sponsored, maintained or contributed to by the Buyer or any of its Affiliates in which the Transferred Employees are eligible to participate on or after the Closing Date (collectively, the "<u>Buyer Plans</u>"), to the same extent as such Transferred Employee was entitled to credit for such service under comparable Plans immediately prior to the Closing, except for benefit accrual under defined benefit pension plans, for purposes of qualifying for subsidized early retirement benefits or to the extent that providing such credit would result in a duplication of benefits. For clarity, the Buyer and its Affiliates shall recognize the Transferred Employees' service with the Sellers or any of their Affiliates (and their predecessors) prior to the Closing for the purposes of (i) accruals and usage of vacation and paid-time-off following the Closing and (ii) all other employment and service related entitlements and benefits.
- 6.5 <u>Terms of Coverage</u>. The Buyer shall cause each Transferred Employee to be immediately eligible to participate, without any waiting time, in any and all Buyer Plans that are welfare benefit plans, to the extent coverage under such Buyer Plan replaces coverage under a comparable Plan in which such Transferred Employee participated immediately prior to the Closing. In addition, for purposes of each Buyer Plan providing medical, dental, pharmaceutical and/or vision benefits, the Buyer shall (i) waive, or cause to be waived, any limitations on benefits relating to pre-existing conditions and actively-at-work requirements for each Transferred Employee and his or her covered spouse, domestic partner or other dependent thereof, and (ii) take into account or cause to be taken into account any eligible

expenses incurred by each Transferred Employee (and his or her covered spouses and dependents) for any co-payments, deductibles, or other out-of-pocket expenses paid by such Transferred Employee or covered spouse, domestic partner or dependent under any comparable Plan during the plan year in which the Closing occurs as if such amounts had been paid under such Buyer Plan.

- 6.6 <u>401(k) Plan Benefits</u>. Effective no later than the Closing Date, the Buyer shall establish or designate a defined contribution plan that includes a qualified cash or deferred arrangement within the meaning of Section 401(k) of the Code (and a related trust exempt from tax under Section 501(a) of the Code) (the "Buyer 401(k) Plan"). Each Transferred Employee shall be immediately eligible to participate in the Buyer 401(k) Plan as of the Closing Date. The Buyer 401(k) Plan shall (no later than the Closing Date) accept rollover distributions of "eligible rollover distributions" (within the meaning of Section 401(a)(31) of the Code) from the 401(k) Plan(s) (including promissory notes evidencing any outstanding loans). To the extent that any such rollover includes a promissory note, the Buyer agrees to expedite the acceptance of such rollovers by the Buyer 401(k) Plan in order to avoid loan defaults.
- 6.7 <u>Liability</u>. Subject to <u>Section 6.1</u>, and without limiting any other provision of this Agreement, effective as of 12:01 a.m. (U.S. Eastern Time) on the Closing Date, the Buyer shall be solely responsible for any severance or separation benefits that become due (a) to any Transferred Employee as a result of his or her termination by the Buyer after the Closing or (b) to any Business Employee as a result of the Buyer's failure to offer employment to or continue the employment of such Business Employee on terms consistent with this <u>Article 6</u> and in accordance with applicable Law.
- 6.8 <u>Retention Incentives</u>. The Buyer shall, or shall cause its Affiliates to, assume and honor the retention incentive arrangements set forth on <u>Schedule 6.8</u> (the "<u>Retention Agreements</u>") and shall pay the retention bonuses thereunder to the applicable Transferred Employees as and when they become due in accordance with the terms of such Retention Agreements, subject to any required withholding for applicable Taxes.
- 6.9 <u>Transaction Bonuses</u>. The Sellers shall, or shall cause one or more of their Affiliates to, pay any earned Transaction Bonuses to the applicable Transferred Employees through a Seller's or its Affiliate's payroll system, subject to any required withholding for applicable Taxes, as and when such Transaction Bonuses become due and payable in accordance with their terms following the Closing. In order to facilitate the payment of such Transaction Bonuses, no later than five (5) Business Days prior to the applicable payment date(s), the Buyer shall, or shall cause one of its Affiliates to, deliver to the Sellers (together with any other information reasonably requested by Sellers that is necessary to determine whether such Transaction Bonuses have become payable) a list of the Transferred Employees who have remained in employment with the Buyer or an Affiliate thereof through the applicable retention date and are eligible to receive Transaction Bonuses (to the extent that such Transaction Bonuses otherwise become earned and payable to the Transferred Employees pursuant to their terms).
- 6.10 WARN Act. The Buyer shall be solely responsible for complying with the WARN Act and any similar applicable Law requiring notice of plant closings, relocations, mass layoffs, reductions in force or similar actions (and for any failures to so comply), in any case, applicable to the Business Employees as a result of any action by the Buyer and its Affiliates after the Closing. The Buyer shall indemnify and hold harmless the Sellers and their Affiliates against any and all Liabilities arising in connection with any failure to comply with the requirements of this Section 6.10.

6.11 No Third Party Beneficiaries. The provisions of this Article 6 are solely for the benefit of the respective Parties to this Agreement and nothing in this Article 6, express or implied, shall confer upon any employee (or any dependent or beneficiary thereof), any rights or remedies, including any right to continuance of employment or any other service relationship with the Buyer, the Sellers or any of their Affiliates, or any right to compensation or benefits of any nature or kind whatsoever under this Agreement. Nothing in this Article 6, express or implied, shall: (i) interfere with the right of the Buyer or its Affiliates to terminate the employment or other service relationship of any Transferred Employee at any time, or (ii) obligate the Buyer or its Affiliates to adopt, enter into or maintain any Plan or other compensatory plan, program or arrangement at any time.

ARTICLE 7 TAX MATTERS

- 7.1 <u>Mutual Cooperation</u>. The Sellers and the Buyer shall, solely with respect to the Businesses and the Transferred Assets, at the other Party's expense (a) each provide the other with such assistance as may reasonably be requested by either of them in connection with the preparation of any Tax Return, or the defense of any audit or other examination by any Taxing Authority or any judicial or administrative Proceeding with respect to Taxes, and (b) each retain and provide the other with any records or other information which the other may reasonably request that are relevant to such Tax Return, audit, examination or Proceeding. Without limiting the generality of the foregoing, the Buyer and Sellers shall retain, until the applicable statutes of limitations (including any extensions) have expired, copies of all Tax Returns, supporting work schedules and other records or information, if any, that may be relevant to such returns for all taxable periods or portions thereof ending before or including the Closing Date.
- 7.2 <u>Payroll Tax Reporting</u>. The Buyer and the Sellers agree that for the taxable year that includes the Closing Date, they will follow the Standard Procedure of Rev. Proc. 2004-53, 2004-34 IRB 320, so that each of the Buyer and the Sellers shall be responsible for employment tax reporting with respect to the wages and other compensation that it pays to Transferred Employees for such calendar year.
- 7.3 Allocation of Certain Taxes. In the case of any taxable period that begins on or before the Closing Date and ends thereafter (each a "Straddle Period"), any real property, personal property, improvement, assessment, special assessment, ad valorem and similar Taxes with respect to the Transferred Assets (such Taxes, "Covered Taxes") for such Straddle Period shall be allocated (a) to the portion of such Straddle Period ending on the Closing Date in an amount equal to the total amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days in the portion of the Straddle Period, and shall be an Excluded Liability, and (b) to the portion of such Straddle Period beginning after the Closing Date in an amount equal to the total amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days in the portion of the Straddle Period after the Closing Date and the denominator of which is the number of days in the entire Straddle Period, and shall be an Assumed Liability; provided, that, the amount of such Covered Taxes shall be determined without regard to any actions taken by the Buyer (or its Affiliates) or other events occurring

after the Closing. The Sellers shall be liable and responsible for the proportionate amount of such Covered Taxes that is attributable to the portion of any Straddle Period ending on the Closing Date, and the Buyer shall be liable and responsible for the proportionate amount of such Covered Taxes that is attributable to the portion of any Straddle Period beginning after the Closing Date. Upon receipt of any bill for any such Covered Taxes, the Buyer or the Sellers, as applicable, shall present a statement to the other setting forth the amount of reimbursement to which each is entitled under this Section 7.3 (taking into account the amounts of Covered Taxes, if any, that the Buyer or the Sellers remitted to a Governmental Authority for any taxable period (or portion thereof) for which the other is responsible pursuant to this Section 7.3), together with such supporting evidence as is reasonably necessary to calculate the proration and reimbursement amount. The proration amount shall be paid by the Party owing it to the other within ten (10) days after delivery of such statement, absent manifest error. The Party that has the primary obligation to do so under applicable Law shall file any Tax Return that is required to be filed in respect of Taxes described in this Section 7.3. The Parties will cooperate and act in good faith to minimize the amount of Covered Taxes.

7.4 <u>Transfer Taxes</u>. All transfer, documentary, sales, use, registration, value-added and other similar Taxes (including any penalties, interest and additions to Tax) incurred in connection with this Agreement, the Ancillary Agreements, the Transaction or the other transactions contemplated hereby and thereby (collectively, "<u>Transfer Taxes</u>") shall be borne and timely paid fifty percent (50%) by the Buyer and fifty percent (50%) by the relevant Seller. Buyer and such Seller shall each, at its own expense, timely file any Tax Return or other document required to be filed by it with respect to such Transfer Taxes, and Buyer and Seller shall cooperate in the preparation and filing of any Tax Returns that must be filed in connection with any Transfer Taxes. Each Party shall use reasonable best efforts to cooperate upon request as reasonably necessary to minimize the amount of any Transfer Taxes or fees applicable to the Transaction.

ARTICLE 8 INTELLECTUAL PROPERTY MATTERS

8.1 IP Transfers.

- (a) Prior to the Closing Date, the Sellers shall rebrand prepaid customer accounts using the Sprint brand as customers of the Boost Business.
- (b) On the Closing Date, the Buyers will: (i) initiate all domain name transfers for domain names included in the Transferred Intellectual Property with the domain name registrar; and (ii) file the Short-Form IP Assignment Agreement for Registered Intellectual Property that is included in the Transferred Intellectual Property with the applicable Governmental Authorities. Seller shall, and shall cause their Subsidiaries to, execute any additional documents and do other acts that are, in each case, reasonably requested by Buyer in order to effectuate and record the transfer of the Transferred Intellectual Property to Buyer.
- (c) The Company shall use its reasonable best efforts to assign to the Buyer that certain Second Amended and Restated Trademark License Agreement by and between Virgin Enterprises Limited and Virgin Mobile USA, LP dated July 27, 2009 (the "Virgin License"). The Buyer shall be responsible for any or all Liabilities arising from its (or its Affiliates') direct or indirect breach of the Virgin License after the Closing Date; provided that the Company shall be responsible for any or all Liabilities arising from its (or its Affiliates') direct or indirect breach of the Virgin License after the Closing Date in connection with any

sublicense granted by Buyer to the Company pursuant to the following sentence. If the Virgin License is assigned to Buyer and to the extent permitted by the terms of the Virgin License, the Buyer shall non-exclusively sublicense the Virgin License back to the Company for a period of twelve (12) months following the Closing solely to allow notice to the applicable Governmental Authorities to remove the Virgin Mark from the Assurance Lifeline brand. Such sublicense shall be subject to other terms and conditions (e.g., quality control) to be negotiated in good faith between the Buyer and the Company.

- (d) Except as provided in <u>Section 8.1(b)</u>, the Parties acknowledge that, from and after the Closing Date, the Sellers shall have no duties or obligations to renew, support, or otherwise maintain any trademark applications or registrations (including any domain name registrations or associated DNS configurations) included in the Transferred Intellectual Property.
- 8.2 <u>Use of Retained Marks</u>. From and after the Closing, neither the Buyer nor any of its Affiliates shall in any way adopt, use, seek to use, apply to register or register any Retained Mark on or in connection with any product, service, or Marks. In no event shall the Buyer or any of its Affiliates use any such Retained Mark after the Closing in any way, including in any manner likely to cause confusion, or to cause mistake or to deceive as to the affiliation, connection, or association of the Buyer or any Affiliate of the Buyer, or as to the origin, sponsorship, or approval of such products or services.
- 8.3 <u>Trade Secret License</u>. Effective as of the Closing, the Company hereby grants, on behalf of themselves and their Subsidiaries, to Buyer and its Affiliates, an irrevocable, perpetual, worldwide, royalty-free, fully paid-up right and license to use and otherwise exploit solely in connection with the operation of the Business (including the conduct of the Business by Buyer and its Affiliates) all Trade Secrets that are (a) owned by the Company or any of its Subsidiaries, (b) used in connection with the Business as conducted as of the Closing Date (other than any aspect of the operation of the Business that is provided by way of services pursuant to the Transition Services Agreement or MNSA), and (c) not included in the Transferred Intellectual Property. Buyer or any of its Affiliates may sublicense any of the rights granted pursuant to this <u>Section 8.3</u> (i) to any of its contractors, channel partners, resellers, or suppliers, in each case, solely in connection with the operation of the Business (including the conduct of the Business by Buyer and its Affiliates), or (ii) in connection with any divestiture of the Business (including the conduct of the Business by Buyer and its Affiliates), or any portion thereof, to a Third Party. All Trade Secrets licensed pursuant to this <u>Section 8.3</u> shall be treated as Confidential Information subject to the confidentiality provisions of <u>Section 12.1(b)</u> and the Buyer will cause any sublicensee of such Trade Secrets to be bound by confidentiality obligations at least as favorable for the benefit of the Sellers.

ARTICLE 9 CONDITIONS TO CLOSING

- 9.1 <u>C</u> onditions to the Obligations of the Sellers and the Buyer. The obligation of the Sellers and the Buyer to consummate the Transaction is subject to the satisfaction (or, to the extent permitted by applicable Law, waiver by the Sellers and the Buyer), as of the Closing, of the following conditions:
 - (a) NTM Merger. The NTM Merger shall have been consummated in accordance with the terms of the NTM Merger Agreement;

- (b) <u>Regulatory Approvals</u>. (i) The DOJ shall have approved the terms of the Transaction as prescribed in any DOJ Final Judgment, (ii) the DOJ Consent and the FCC Consent shall have been obtained, (iii) the FCC 214 Approval shall have been obtained, (iv) the FCC MNSA Approval shall have been obtained and (v) any required approvals by state public utility authorities shall have been obtained; and
- (c) No Legal Restraint. No applicable Law enacted, modified, supplemented or amended after the date of this Agreement or Governmental Order enacted, entered, promulgated, enforced or issued by, or executed with, any Governmental Authority that prevents or, with respect to any Governmental Order, restrains the Transaction shall be in effect.
- 9.2 <u>C onditions to the Obligations of the Buyer</u>. The obligation of the Buyer to consummate the Transaction is further subject to the satisfaction (or, to the extent permitted by applicable Law, waiver by the Buyer), as of the Closing, of the following conditions:
- (a) Representations and Warranties. (i) The representations and warranties of the Sellers set forth in Article 3 (other than the Seller Fundamental Representations and Section 3.8(b)) shall be true and correct (without giving effect to any qualifications or limitations contained therein as to materiality or Material Adverse Effect, as of the date of this Agreement and as of the Closing Date as though made as of such time, except to the extent such representations and warranties expressly relate to an earlier date (in which case such representations and warranties shall be true and correct as of such earlier date), in each case, other than for failures of such representations and warranties of the Sellers to be so true and correct which do not have or are not reasonably likely to have, individually or in the aggregate, a Material Adverse Effect and (ii) the Seller Fundamental Representations and Section 3.8(b) shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date);
- (b) <u>Covenants</u>. The covenants and obligations that the Sellers are required to perform prior to the Closing shall have been duly performed and complied with by the Sellers in all material respects;
- (c) The Sellers shall have taken all actions required to enable Buyer to, and from and after the Closing the Buyer shall, have the ability to provision any new or existing customer of the Business holding a compatible handset device onto the NTM Network pursuant to the terms of the MNSA.
- (d) The Buyer shall have received at the Closing a certificate signed on behalf of the Sellers by an executive officer of each of TMUS and Sprint to the effect that the conditions set forth in Section 9.2(a) and Section 9.2(b) have been satisfied; and
 - (e) The Buyer shall have received each of the deliveries set forth in Section 10.2(a) required to be delivered to Buyer or any of its Affiliates.

- 9.3 <u>Conditions to the Obligations of the Sellers</u>. The obligation of the Sellers to consummate the Transaction is further subject to the satisfaction (or, to the extent permitted by applicable Law, waiver by the Sellers), as of the Closing, of the following conditions:
- (a) Representations and Warranties. (i) The representations and warranties of the Buyer set forth in Article 4 (other than the Buyer Fundamental Representations) shall be true and correct (without giving effect to any qualifications or limitations contained therein as to materiality or Buyer Material Adverse Effect, as of the date of this Agreement and as of the Closing Date as though made as of such time, except to the extent such representations and warranties expressly relate to an earlier date (in which case such representations and warranties shall be true and correct as of such earlier date), in each case, other than for failures of such representations and warranties of the Buyer to be so true and correct which do not have or are not reasonably likely to have, individually or in the aggregate, a Buyer Material Adverse Effect and (ii) the Buyer Fundamental Representations shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date);
- (b) <u>Covenants</u>. The covenants and obligations that the Buyer is required to perform at or prior to the Closing shall have been duly performed and complied with by the Buyer in all material respects (except that the obligations of the Buyer set forth in <u>Section 10.2(b)(i)</u> shall have been performed in all respects);
- (c) The Seller shall have received at the Closing a certificate signed on behalf of the Buyer by an executive officer of Buyer to the effect that the conditions set forth in <u>Section 9.3(a)</u> and <u>Section 9.3(b)</u> have been satisfied; and
 - (d) The Sellers shall have received each of the deliveries set forth in Section 10.2(b) required to be delivered to Sellers or any of its Affiliates.

ARTICLE 10 CLOSING

10.1 <u>Closing</u>. Subject to the terms and conditions of this Agreement, the closing of the Transaction (the "<u>Closing</u>") shall take place at the offices of Latham & Watkins LLP, 885 Third Avenue, New York, New York 10022, at 10:00 a.m. (U.S. Eastern Time) on the first (1 st.) Business Day of the month immediately following the month in which all conditions set forth in <u>Article 9</u> have been satisfied or waived by the Party entitled to the benefit thereof (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions at the Closing) (the "<u>Month Start Date</u>"), <u>provided</u>, <u>however</u>, that if Month Start Date would occur after the date on which any Regulatory Approval requires the Closing to occur, then the Closing shall occur on the last Business Day specified by such Regulatory Approval. The date on which the Closing actually occurs is referred to in this Agreement as the "<u>Closing Date</u>". The Closing shall be deemed effective as of 12:01 a.m. (U.S. Eastern Time) on the Closing Date.

10.2 Deliverables.

- (a) At the Closing, the Sellers shall deliver, or cause to be delivered, to the Buyer:
- (i) a counterpart of a bill of sale and assignment and assumption agreement in order for the Buyer or its Affiliate to assume and undertake the obligations to pay, perform and discharge, as and when due, the Assumed Liabilities and in order for the Sellers to assign to the Buyer or its Affiliate all of each Seller's right, title and interest in and to the Transferred Assets, in in form and substance reasonably acceptable to the Buyer and the Sellers (the "Bill of Sale and Assignment and Assumption Agreement"), duly executed by the applicable Sellers or Affiliates thereof;
 - (ii) a counterpart of each Transition Services Agreement, duly executed by the applicable Seller or an Affiliate thereof;
 - (iii) a counterpart of the MNSA, duly executed by the Sellers or Affiliates thereof;
 - (iv) a counterpart of the Spectrum Purchase Agreement, duly executed by the Sellers or Affiliates thereof;
 - (v) a counterpart of the Site Option Agreement, duly executed by the Sellers or Affiliates thereof;
 - (vi) a counterpart of the Short-Form IP Assignment Agreement, duly executed by the Sellers or Affiliates thereof; and
 - (vii) an IRS Form W-9 duly executed by each Seller.
 - (b) At Closing, the Buyer shall deliver to the Sellers:
- (i) the Purchase Price (excluding any adjustment for Net Working Capital), by wire transfer of immediately available funds to the account(s) designed by the Sellers prior to the Closing Date;
 - (ii) a counterpart of each Transition Services Agreement, duly executed by Buyer;
 - (iii) a counterpart of the MNSA, duly executed by Buyer;
 - (iv) a counterpart of the Spectrum Purchase Agreement, duly executed by Buyer;
 - (v) a counterpart of the Site Option Agreement, duly executed by Buyer;
 - (vi) a counterpart of the Bill of Sale and Assignment and Assumption Agreement; and
 - (vii) a counterpart of the Short-Form IP Assignment Agreement, duly executed by Buyer.

ARTICLE 11 TERMINATION

- 11.1 Termination. This Agreement may be terminated at any time before the Closing and the Transaction abandoned:
 - (a) by the mutual written consent of the Buyer and the Sellers;
 - (b) by written notice to the Sellers from the Buyer if:
- (i) there is any material breach of any representation, warranty, covenant or agreement set forth in this Agreement on the part of the Sellers, such that the conditions specified in Section 9.1 or Section 9.2 would not be satisfied at the Closing (a "Terminating Sellers Breach"), except that, if such Terminating Sellers Breach is curable by the Sellers through the exercise of their reasonable best efforts, then, for a period of up to forty-five (45) days after receipt by the Sellers of notice from the Buyer of such Terminating Sellers Breach (the "Sellers Cure Period"), such termination shall not be effective, and such termination shall become effective only if (A) the Terminating Sellers Breach is not cured within the Sellers Cure Period or (B) the Terminating Sellers Breach is incapable of being cured during the Sellers Cure Period; provided that this right of termination shall not be available to the Buyer if the Buyer is in material breach of any of its representations, warranties, covenants or agreements contained in this Agreement; or
- (ii) the Closing has not occurred on or before the earlier to occur of (x) the date that is twelve (12) months after the date of this Agreement and (y) the Termination Date; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 11.1(b)(ii)</u> shall not be available to the Buyer's failure to comply with its obligations under this Agreement has materially contributed to the failure of the Closing to occur before such date; <u>provided</u>, <u>further</u>, that, if the Termination Date falls during a Sellers Cure Period, the Termination Date shall automatically be extended until the end of such Sellers Cure Period;
 - (c) by written notice to the Buyer from either Seller if:
- (i) there is any material breach of any representation, warranty, covenant or agreement set forth in this Agreement on the part of the Buyer, such that the conditions specified in Section 9.1 or Section 9.3 would not be satisfied at the Closing (a "Terminating Buyer Breach"), except that, if any such Terminating Buyer Breach is curable by the Buyer through the exercise of its reasonable best efforts, then, for a period of up to forty-five (45) days after receipt by the Buyer of notice from the Sellers of such Terminating Buyer Breach (the "Buyer Cure Period"), such termination shall not be effective and such termination shall become effective only if (A) the Terminating Buyer Breach is not cured within the Buyer Cure Period or (B) the Terminating Buyer Breach is incapable of being cured during the Buyer Cure Period (it being understood and agreed that failure by the Buyer to consummate the Closing within five (5) Business Days of the date on which the Closing is required to occur pursuant to Section 10.1 shall be deemed a Terminating Buyer Breach that is incapable of being cured); provided that this right of termination shall not be available to the Sellers if the Sellers are in material breach of any of their representations, warranties, covenants or agreements contained in this Agreement; or

- (ii) the Closing has not occurred on or before the Termination Date; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 11.1(c)(ii)</u> shall not be available to the Sellers if either (i) the Sellers' failure to comply with their obligations under this Agreement has materially contributed to the failure of the Closing to occur before such date or (ii) the condition set forth in <u>Section 9.2(c)</u> has not been satisfied as of such date (only for so long as the condition set forth in <u>Section 9.2(c)</u> remains unsatisfied); <u>provided</u>, <u>further</u>, that, if the Termination Date falls during a Buyer Cure Period, the Termination Date shall automatically be extended until the end of such Buyer Cure Period;
- (iii) if any Governmental Authority modifies or imposes a modification of the terms of the DOJ Final Judgment, this Agreement or the Ancillary Agreements and such modifications are not acceptable to the Sellers in their sole discretion, <u>provided</u>, that in the event that the Sellers enter into any definitive agreement or amendment approving such modification, or otherwise consent in writing to any such modification, that shall mean that such modification is acceptable in form and substance to the Sellers.
- (d) by any Party by written notice to the other Parties if the NTM Merger has been validly terminated pursuant to the terms of the NTM Merger Agreement; or
- (e) by written notice to the Buyer from the Sellers if the DOJ has informed the Sellers in writing that it will not grant the DOJ Consent, or if the DOJ withdraws or retracts the DOJ Consent and informs the Sellers that it will not re-grant the DOJ Consent.
- 11.2 Effect of Termination. Except as otherwise set forth in this Section 11.2 and except for the second sentence of Section 14.12, in the event of the termination of this Agreement pursuant to Section 11.1, this Agreement shall forthwith become void and have no effect, without any liability on the part of any Party or its Affiliates, officers, directors, members or stockholders, other than liability of a Party for any Willful Breach of this Agreement by such Party occurring prior to such termination.

ARTICLE 12 POST-CLOSING COVENANTS

12.1 Confidentiality.

(a) Effective as of the execution of this Agreement, Buyer and Sellers have entered into the Amended and Restated Confidentiality Agreement. The Buyer acknowledges effective upon, and only upon, the Closing, the Amended and Restated Confidentiality Agreement shall be automatically terminated and without further effect; provided, however, that notwithstanding the foregoing, each Party acknowledges that any and all other information provided to it by or on behalf of the other Party, any of their respective Affiliates or Representatives as concerning the disclosing party, their respective Affiliates (other than the Business) and the terms and conditions of this Agreement and the Ancillary Agreements shall remain subject to the confidentiality terms of the Confidentiality Agreement until the date that the Amended and Restated Confidentiality Agreement would have otherwise been terminated in accordance with its terms and provided, further, that the obligations not to disclose Transaction Information (as such term is defined in the Amended and Restated Confidentiality Agreement) to the individuals identified on Exhibit A of the Amended and Restated Confidentiality Agreement shall survive any termination thereof.

(b) From and after the Closing, the Parties agree that any information provided to the other Parties or their Representatives pursuant to the terms of this Agreement (including Section 5.8 or Section 12.2) or otherwise in connection with the Transaction (including the terms and conditions of this Agreement and the Ancillary Agreements, but, in the case of Buyer, excluding all Business Confidential Information that relates exclusively to the Business) (the "Confidential Information") shall be kept strictly confidential, and the Parties and their Representatives shall not disclose any such Confidential Information in any manner whatsoever, provided, however, that a Party may disclose any such Confidential Information to such of its Representatives who have been informed of the confidential nature of the information and have been instructed to comply with the terms of this Section 12.1(b) as if they were parties hereto (provided the Parties shall not disclose any pricing information under the MNSA to actual or potential debt and equity financing sources; provided, however, that the Buyer may share with such actual and potential debt and equity financing sources financial forecasts and models that reflect such pricing information on an aggregated line item basis), and Trade Secrets subject to Section 8.3 may be disclosed to sublicensees in accordance with Section 8.3. The term "Confidential Information" does not include information that: (i) is or becomes generally available to the public (other than as a result of a disclosure by a Party or any of its Representatives in breach of this Section 12.1(b), (ii) (A) was within such receiving Party's possession prior to it being furnished or otherwise disclosed to such receiving Party hereunder or (B) thereafter becomes available to such receiving Party on a non-confidential basis from a source other than the disclosing Party or its Representatives, in either case of clause (A) or (B) without, to such receiving Party's knowledge, being subject to any contractual or other obligation of confidentiality to the disclosing Party or any of its Representatives with respect to such information, or (iii) is or was independently developed by the receiving Party or any of its Representatives without use of, reference to or reliance upon Confidential Information; provided that the foregoing exceptions (i)-(iii) shall not apply with respect to any Trade Secrets. In the event that a Party or any of its Representatives receives a request from a Governmental Authority of competent jurisdiction (by deposition, interrogatory, request for documents, order, subpoena, civil investigative demand or similar process) to disclose, or is otherwise required by Law or the rules of a national securities exchange to which it is subject, to disclose Confidential Information then the provisions of Paragraph 2.2 of the Amended and Restated Confidentiality Agreement shall apply to any such disclosure and are incorporated by reference into this Section 12.1(b), mutatis mutandis, as if they were set forth in their entirety herein. The confidentiality restriction set forth in this Section 12.1(b) shall expire with respect to any Confidential Information three (3) years after the date such Confidential Information is originally provided to a Party or its Representatives by another Party or its Representatives, provided that this Section 12.1(b) shall not expire (i) with respect to any Trade Secrets or (ii) with respect to the obligation not to disclose any pricing information under the MNSA.

(c) From and after the date of this Agreement, each of the Sellers shall not, and shall cause its Affiliates and its and their Representatives not to, directly or indirectly, disclose, reveal, divulge or communicate to any Person other than authorized officers, directors, managers and employees of the Buyer or its Affiliates or use or otherwise exploit, any Business Confidential Information (x) in a manner that would reasonably be expected to be detrimental to the Business or (y) for the purposes of marketing or customer solicitation for the benefit of anyone other than the Business, the Buyer or its Affiliates. Each of the Sellers and their respective directors, officers, employees or Affiliates shall not have any obligation to keep confidential any Business Confidential Information if and to the extent disclosure thereof is specifically required by Law or requested by FCC staff in connection with the FCC MNSA Approval. To the extent legally permissible, such Seller shall notify the Buyer of its intention

to make such disclosure and provide a list of the Business Confidential Information that such Seller intends to disclose prior to making such disclosure. The Sellers agree to cooperate with the Buyer so that the Buyer may seek, at its sole cost and expense, an appropriate protective order. "Business Confidential Information" means any market studies and forecasts, competitive analyses, target markets, advertising techniques, pricing policies and information, customer lists, customer profiles, customer preferences, other trade secrets and any other documents embodying confidential and proprietary information to the extent related to the Business.

12.2 Access to Books and Records; Cooperation.

- (a) From and after the Closing, each of the Sellers, the Buyer and their respective Affiliates shall retain the books, Records, documents, instruments, accounts, correspondence, writings, evidences, of title and other papers relating to the Business in their possession or control for at least seven (7) years following the Closing Date or for such longer period as may be required by Law or any applicable Governmental Order, and shall, and shall cause their respective Affiliates to provide the other Party and its Affiliates and Representatives reasonable access during regular business hours to such Records (including the right to receive hard or electronic copies thereof), including to the extent necessary to: (i) fully enforce, determine or exercise any rights or obligations in connection with the NTM Merger, or (ii) perform any mandatory obligations towards any Governmental Authorities, government-controlled entities, public international organizations, or non-governmental institutions whose employees are treated because of that status or otherwise as government officials under applicable Laws.
- (b) From and after the Closing, the Buyer shall, and shall cause its Affiliates to, provide such assistance and cooperation to the Sellers and their respective Affiliates and Representatives as any Seller may reasonably request in relation to any Proceedings by or against any Seller or its Affiliates, including Proceedings relating to employee claims. The Sellers shall, and shall cause their respective Affiliates to, provide such assistance and cooperation at the Buyer's cost and expense to the Buyer and its Affiliates and Representatives as the Buyer may reasonably request in relation to required regulatory filings of Buyer or its Affiliates (including securities law filings) in respect of periods ending on or prior to the Closing (including providing such information, including financial information, as is reasonably requested and making extracts and copies of such Records, including reasonable access to customary supporting information, data and documentation necessary for the preparation of the Buyer's financial reports), in connection with any Proceedings, including Proceedings relating to employee claims, or to the extent reasonably necessary or advisable to operate the Business after the Closing. The respective Parties shall be entitled, at their sole cost and expense, to make copies of the Records to which they are entitled to access pursuant to this Section 12.2. Notwithstanding anything to the contrary set forth in this Section 12.2, in no event shall the Sellers be required to prepare any balance sheet, cash flow statement, income statement or statement of stockholders' equity with regard to the Business, the Transferred Assets or the Assumed Liabilities, whether prior to or following the Closing

12.3 Wrong -Pocket. If at any time during the five (5)-year period after the Closing:

(a) The Sellers or any of their respective Affiliates receives (i) any refund or other amount which is a Transferred Asset or is otherwise properly due and owing to the Buyer in accordance with the terms of this Agreement or (ii) any refund or other amount which is related to claims or other matters for which the Buyer is responsible hereunder, and which

amount is not an Excluded Asset, or is otherwise properly due and owing to the Buyer in accordance with the terms of this Agreement, the Sellers promptly shall remit, or shall cause to be remitted, such amount to the Buyer at the address set forth in <u>Section 14.11</u>, net of any out-of-pocket expenses and costs (including Taxes) incurred in connection with determining, collecting or obtaining such refund or other amount; or

- (b) The Buyer or any of its Affiliates receives (i) any refund or other amount which is an Excluded Asset or is otherwise properly due and owing to the Sellers or any of their respective Affiliates in accordance with the terms of this Agreement, or (ii) any refund or other amount which is related to claims or other matters for which the Sellers are responsible hereunder, and which amount is not a Transferred Asset, or is otherwise properly due and owing to the Sellers or any of their respective Affiliates in accordance with the terms of this Agreement, the Buyer promptly shall remit, or shall cause to be remitted, such amount to the Sellers at the address set forth in Section 14.11, net of any out-of-pocket expenses and costs (including Taxes) incurred in connection with determining, collecting or obtaining such refund or other amount.
- (c) If at any time after the Closing, the Buyer or any of its Subsidiaries shall receive or otherwise possess any asset or liability that should belong to the Sellers or any of their Affiliates pursuant to this Agreement, Buyer shall, except to the extent the asset is not transferable as provided in Section 12.3(e) promptly notify and transfer, or cause to be transferred, such asset or liability to the Sellers or any of their Affiliates. If at any time after the Closing, the Sellers or any of their respective Affiliates shall receive or otherwise possess any asset or liability that should belong to Buyer or any of its Subsidiaries pursuant to this Agreement, the Sellers shall, except to the extent the asset is not transferable as provided in Section 12.3(e), promptly notify and transfer, or cause to be transferred, such asset or liability to Buyer or any of its Subsidiaries. Prior to any such transfer of assets pursuant to this Section 12.3(c), the Sellers and the Buyer agree that the Person receiving or possessing such asset shall hold such asset in trust for the Person to whom such asset should rightfully belong pursuant to this Agreement.
- (d) If at any time there exist (i) assets that any Party discovers were, contrary to the agreements among the Parties, by mistake or unintentional or other omission, transferred to Buyer or retained by Sellers or any of their respective Affiliates or (ii) Liabilities that any Party discovers were, contrary to the agreements among the Parties, by mistake or unintentional or other omission, assumed by Buyer or retained by Sellers or any of their respective Affiliates, then the Parties shall cooperate in good faith to effect the transfer or retransfer of such misallocated assets, and/or the assumption or reassumption of misallocated Liabilities, to or by the appropriate Person as promptly as practicable and shall not use the determination that remedial actions need to be taken to alter the original intent of the Parties with respect to the assets to be transferred to or Liabilities to be assumed by Buyer or retained by Sellers or any of their respective Affiliates. Each Party shall reimburse any other Party or make other financial adjustments or other adjustments to remedy any mistakes or omissions relating to any of the assets transferred or any of the Liabilities assumed or retained pursuant to this Section 12.3(d).
- (e) Notwithstanding anything to the contrary in this Agreement or any Ancillary Agreement, neither this Agreement nor any Ancillary Agreement shall constitute an agreement to sell, assign, transfer, convey, deliver or assume any asset that would constitute a Transferred Asset if such asset is not transferable in accordance with applicable Law or with any requisite Consent. If the transfer or assignment of any asset intended to be transferred or

assigned hereunder is not consummated prior to or on the Closing Date, whether as a result of a prohibition on transfer due to a violation or breach of applicable Law or any other requisite Consent, then the Person retaining such asset shall thereafter hold such asset for the use and benefit, insofar as legally permitted and reasonably possible, of the Person entitled thereto until the consummation of the transfer or assignment thereof (or as otherwise mutually determined by the Parties). In addition, the Person retaining such asset shall use its reasonable best efforts to take such other actions as may reasonably be requested by the Person to whom such asset is to be transferred in order to place such Person, insofar as legally permitted and reasonably possible, in the same position as if such asset had been transferred as contemplated hereby and so that all the benefits and burdens relating to such asset, including possession, use, risk of loss, potential for gain, and dominion, control and command over such asset, are to inure from and after the Closing Date to the Person to whom such asset is to be transferred. Notwithstanding the foregoing, any such asset shall still be considered a Transferred Asset. This provision is intended, among other things, to cause all such assets that are considered Transferred Assets to be treated for all income tax and accounting purposes as if transferred as contemplated hereby, such that such assets are and will be, for all income tax and accounting purposes, owned by the Buyer, and the Parties will so treat such assets for all income tax and accounting purposes, except as required by Law. The Person retaining an asset due to the deferral of the transfer and assignment of such asset shall not be obligated, in connection with the foregoing, to expend any money or personnel in connection with the maintenance of the asset unless the necessary funds or expenses or costs associated with such maintenance are advanced by the Person to whom such asset is to be transferred, other than reasonable out-of-pocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by the Person to whom such asset is to be transferred; provided, however, that the Person retaining such asset shall, as promptly as practicable, provide notice to the Person to whom such asset is to be transferred of the amount of all such expenses and fees.

- (f) Each Party hereto shall cooperate with each other Party hereto and shall set up procedures and notifications as are reasonably necessary or advisable to effectuate the transfers contemplated by this $\underline{\text{Section } 12.3}$.
- (g) For the avoidance of doubt, the transfer or assumption of any assets or Liabilities under this <u>Section 12.3</u> shall be effected without any additional consideration payable by any Party hereto.
- 12.4 <u>Transfer Formalities and Costs</u>. Without limiting <u>Section 14.1</u>, all registration, filing and other formalities relating to or required for the transfer of the Transferred Assets shall be the sole responsibility of the Buyer. The Buyer shall pay all costs and expenses associated with or arising from the transfer of the Transferred Assets.
- 12.5 <u>Insurance</u>. Except as set forth in <u>Section 2.3(b)</u>, from and after the Closing, the Business, the Transferred Assets and the Assumed Liabilities shall cease to be insured by Sellers' or their respective Affiliates' insurance policies or by any of their self-insurance programs or other similar arrangements, and Buyer (i) agrees to arrange for its own insurance policies (including self-insurance or similar arrangements funded directly or indirectly by Buyer or any of its Affiliates) with respect to the Business, the Transferred Assets and the Assumed Liabilities covering all periods from and after the Closing and (ii) without prejudice to any right to indemnification under this Agreement or any other Ancillary Agreement, agrees not to seek, through any means, to benefit from any of Seller's or its Affiliates' insurance policies which may provide coverage for claims relating in any way to the Business.

12.6 Non-Solicitation.

- (a) From and after the Closing and until the eighteen (18)-month anniversary of the Closing Date, the Sellers shall not, and shall cause each of their respective Affiliates not to, directly or indirectly, solicit for employment or engagement, knowingly entice away, or hire or engage (whether as an employee, consultant, agent, contractor or otherwise) any Transferred Employee who is at or above a Consultant/Manager position; provided that the Sellers and their respective Affiliates shall not be precluded from soliciting, hiring or engaging any such person: (i) pursuant to a general solicitation of employment not targeted toward employees of the Buyer or its Affiliates, or (ii) whose employment or engagement, as applicable, with the Buyer or its Affiliates has been terminated at least six (6) months prior to such soliciting, hiring or engaging with the Sellers or their respective Affiliates.
- (b) <u>Scope</u>. The Parties acknowledge that the restrictions contained in this <u>Section 12.6(a)</u> are reasonable in scope and duration in light of the nature, size and location of the Business. The Parties further acknowledge that the restrictions contained in this <u>Section 12.6(a)</u> are necessary to protect Buyer's significant investment in the Business, including investment in goodwill. It is the desire and intent of the Parties that the provisions of this <u>Section 12.6(a)</u> be enforced to the fullest extent permissible under applicable Law. If any part of this <u>Section 12.6</u> is held to be excessively broad as to duration, scope activity or subject, such part will be construed by limited and reducing it so as to be enforceable to the maximum extent permitted under applicable Law.
 - 12.7 Additional Post-Closing Covenants. The Parties shall take the actions described on Schedule 12.7.

ARTICLE 13 INDEMNIFICATION

- 13.1 <u>Indemnification by the Seller s</u>. From and after the Closing, subject to the limitations contained in this <u>Article 13</u>, the Sellers shall jointly and severally indemnify the Buyer and its Affiliates and their respective officers, directors, employees, agents successors and assigns (the "<u>Buyer Indemnified Parties</u>") in respect of, and hold each Buyer Indemnified Party harmless against any and all Losses incurred or suffered by such Buyer Indemnified Party resulting from, arising out of or relating to:
- (a) any inaccuracy in any representation or warranty of the Sellers contained in Article 3 as of the date hereof or as of the Closing Date (except to the extent such representation or warranty speaks as of a particular date, in which case such inaccuracy shall be determined as of such particular date) (without giving effect to any Material Adverse Effect or other materiality qualification or any similar qualification contained or incorporated directly or indirectly in such representation or warranty);
 - (b) any breach or failure by the Sellers to perform any covenant or agreement of any Seller contained in this Agreement;
 - (c) any failure to collect in full any amount of accounts receivable included in the Final NWC Statement; or
 - (d) any Excluded Liabilities.

- 13.2 <u>Indemnification by Buyer</u>. From and after the Closing, subject to the limitations contained in this <u>Article 13</u>, the Buyer shall indemnify the Sellers and their respective Affiliates and their respective officers, directors, employees, agents, successors and assigns (the "<u>Seller Indemnified Parties</u>") in respect of, and hold each Seller Indemnified Party harmless against, Losses incurred or suffered by such Seller Indemnified Party resulting from or arising out of:
- (a) any inaccuracy in any representation or warranty of the Buyer contained in <u>Article 4</u> as of the date hereof or as of the Closing Date (except to the extent such representation or warranty speaks as of a particular date, in which case such inaccuracy shall be determined as of such particular date) (without giving effect to any Buyer Material Adverse Effect or other materiality qualification or any similar qualification contained or incorporated directly or indirectly in such representation or warranty);
 - (b) any breach or failure by Buyer to perform any covenant or agreement of Buyer contained in this Agreement; or
 - (c) any Assumed Liabilities.

13.3 Indemnification Claims.

(a) An Indemnified Party shall give written notification to the Indemnifying Party of the commencement of any Third Party Claim. Such notification shall be given within thirty (30) days after receipt by the Indemnified Party of notice of such Third Party Claim, and shall describe in reasonable detail (to the extent known by the Indemnified Party) the facts constituting the basis for such Third Party Claim and the amount of damages claimed therein (if specified); provided, however, that no delay or failure on the part of the Indemnified Party in so notifying the Indemnifying Party shall relieve the Indemnifying Party of any liability or obligation hereunder except and only to the extent that the Indemnifying Party is actually prejudiced by such delay or failure. Within twenty (20) days after delivery of such notification, the Indemnifying Party shall have the right to, upon written notice thereof to the Indemnified Party, assume control of and conduct, at the Indemnifying Party's sole cost and expense, the defense of such Third Party Claim (with counsel of national standing reasonably satisfactory to the Indemnified Party); provided, that (i) as a condition precedent to the Indemnifying Party's right to assume and conduct such defense, within fifteen (15) days after the Indemnified Party has given notice of such Third Party Claim, (A) the Indemnifying Party must notify the Indemnified Party in writing that the Indemnifying Party shall undertake the defense of such Third Party Claim and (B) the Indemnifying Party must agree in writing with the Indemnified Party to unconditionally indemnify the Indemnified Party from and against all such Losses that the Indemnified Party may suffer or incur or to which the Indemnified Party may otherwise become subject and which arise from or as a result of or are connected with such Third Party Claim (subject to the limitations set forth in Section 13.6), and (ii) the Indemnifying Party may not assume control of the defense of or conduct the defense of, any Third Party Claim to the extent such claim constitutes a Third Party Claim (A) involving any criminal or quasicriminal Proceeding, action, indictment, allegation or investigation or seeking to impose any criminal penalty, fine or other sanction, (B) made by any Governmental Authority or to which any Governmental Authority is a named party, (C) in which relief other than monetary Losses is sought, including any injunctive or other equitable relief, (D) which, if adversely determined, would reasonably be expected, in the good faith judgment of the Indemnified Party, to establish a precedent, custom or practice materially adverse to the continuing business interests or prospects of the Indemnified Party or its Affiliates, or (E) that could, in the good faith judgment

of the Indemnified Party reasonably be expected to result in Losses in excess of the Cap (and for purposes of any claims with respect to breaches of the representations and warranties other than the Seller Fundamental Representations or the Buyer Fundamental Representations, as applicable) or otherwise in excess of the maximum liability of the Seller Indemnifying Parties or the Buyer Indemnifying Parties, as applicable under this Article 13. For avoidance of doubt, in the event that a Seller or one or more of its Affiliates and the Buyer or one or more of its Affiliates is named in an Proceeding, such Seller and the Buyer shall be entitled to assume the control of and conduct of their own defense and select counsel of their own choosing to defend their respective interests in such Proceeding.

- (b) If the Indemnifying Party does not so assume or does not have the right to so assume control of the defense of a Third Party Claim, the Indemnified Party shall control such defense. The Non-Controlling Party may participate in such defense, and may hire separate counsel at its own expense. The Controlling Party shall keep the Non-Controlling Party reasonably advised of the status of such Third Party Claim and the defense thereof and shall consider in good faith recommendations made by the Non-Controlling Party with respect thereto. The Non-Controlling Party shall furnish the Controlling Party with such information as it may have with respect to such Third Party Claim (including copies of any summons, complaint or other pleading which may have been served on such party and any written claim, demand, invoice, billing or other document evidencing or asserting the same) and shall otherwise reasonably cooperate with and assist the Controlling Party in the defense of such Third Party Claim, including by (i) furnishing and, upon request, procuring the attendance of potential witnesses for interview, preparation, submission of witness statements and the giving of evidence at any related hearing, (ii) promptly furnishing documentary evidence to the extent available to it or its Affiliates, and (iii) providing access to any other relevant party, including any Representatives of the Non-Controlling Party as reasonably needed. Notwithstanding the foregoing, the fees and expenses of counsel to the Indemnified Party with respect to a Third Party Claim shall be considered Losses for purposes of this Agreement only if (A) the Indemnified Party shall have determined in good faith that an actual or potential conflict of interest makes representation by the same counsel or the counsel selected by the Indemnifying Party inappropriate or (B) the Indemnifying Party shall have authorized in writing the Indemnified Party to employ separate counsel at the Indemnifying Party's expense. The Indemnifying Party shall not agree to any settlement of, or the entry of any judgment arising from, any Third Party Claim without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld, delayed or conditioned), unless the relief consists solely of money Losses to be paid by the Indemnifying Party. The Indemnified Party shall not agree to any settlement of, or the entry of any judgment arising from, any such Third Party Claim without the prior written consent of the Indemnifying Party (which consent shall not be unreasonably withheld, delayed or conditioned).
 - (c) In order to seek indemnification under this Article 13, an Indemnified Party shall deliver a Claim Notice to the Indemnifying Party.
- (d) Within twenty (20) days after delivery of a Claim Notice, the Indemnifying Party shall deliver to the Indemnified Party a written response (the "Response"), in which the Indemnifying Party shall: (i) agree that the Indemnified Party is entitled to receive all of the Claimed Amount (in which case the Response shall be accompanied by a payment by the Indemnifying Party to the Indemnified Party of the Claimed Amount, by check or by wire transfer), (ii) agree that the Indemnified Party is entitled to receive the part, but not all, of the Claimed Amount (the "Agreed Amount") (in which case the Response shall be accompanied

by a payment by the Indemnifying Party to the Indemnified Party of the Agreed Amount, by check or by wire transfer), or (iii) dispute that the Indemnified Party is entitled to receive any of the Claimed Amount (whereupon the Indemnifying Party and the Indemnified Party agree that the dispute shall be resolved in accordance with Section 14.5).

- 13.4 <u>Survival of Representations and Warranties and Covenants</u>. The representations and warranties set forth in Sections 3.1, 3.2, 3.15(a) and 3.18 (the "<u>Seller Fundamental Representations</u>") and Sections 4.1, 4.2 and 4.6 (the "<u>Buyer Fundamental Representations</u>") shall survive the Closing for three (3) years from the Closing Date and then expire. All representations and warranties in this Agreement other than the Seller Fundamental Representations and Buyer Fundamental Representations shall survive the Closing for eighteen (18) months from the Closing Date. All covenants and agreements set forth herein which by their terms contemplate actions or impose obligations prior to or at the Closing shall survive the Closing for eighteen (18) months from the Closing Date and then expire. All covenants and agreements set forth herein which by their terms contemplate actions or impose obligations following the Closing shall survive the Closing and remain in full force and effect in accordance with their terms. If an Indemnified Party delivers to an Indemnifying Party, before expiration of a representation or warranty or covenant or agreement, either a Claim Notice based upon a breach of such representation or warranty or covenant, then the applicable representation or warranty or covenant shall survive until, but only for purposes of, the resolution of the matter covered in reasonable detail by such notice. If the Proceeding or written claim with respect to which an Expected Claim Notice has been given is definitively withdrawn or resolved in favor of the Indemnified Party, the Indemnified Party shall promptly so notify the Indemnifying Party.
- 13.5 <u>Treatment of Indemnification Payments</u>. Any payments made to an Indemnified Party pursuant to this <u>Article 13</u> shall be treated as an adjustment to the Final Purchase Price for Tax purposes and such treatment shall govern for all purposes hereof except to the extent that the applicable Laws or Taxing Authority of a particular jurisdiction require otherwise.

13.6 Limitations on Indemnification.

- (a) No amount shall be payable to an Indemnified Party in satisfaction of any claim for indemnification pursuant to <u>Section 13.1(a)</u> or <u>13.2(a)</u>, as applicable, for which the aggregate amount of Losses of the Indemnified Party arising therefrom, together with all related claims, is less than \$50,000; <u>provided</u>, that the limitation set forth in this <u>Section 13.6(a)</u> shall not apply unless and until the aggregate amount of all Losses of the Indemnified Party resulting from, arising out of or relating to breaches of representations and warranties contained in this Agreement exceeds the Deductible Amount.
- (b) No claim may be made by an Indemnified Party against an Indemnifying Party for indemnification pursuant to Section 13.1(a) or Section 13.2(a), as applicable, unless and until the Indemnified Party has sustained aggregate Losses for which it is entitled to indemnification pursuant to Section 13.1(a) or Section 13.2(a), as applicable, in excess of \$14,000,000 in the aggregate (the "Deductible Amount") and then only to the extent such aggregate amount exceeds the Deductible Amount, provided, however, that the Deductible Amount shall not apply with respect to any Losses resulting from, arising out of or relating to breaches of the Seller Fundamental Representations or the Buyer Fundamental Representations, and none of such Losses shall count towards the satisfaction of the Deductible Amount. The maximum aggregate recovery of the Indemnified Parties from the Indemnifying Parties pursuant to Section 13.1(a) or Section 13.2(a), as applicable, shall not exceed an amount equal to \$140,000,000 the ("Cap."), except with respect to breaches of the Seller Fundamental Representations or Buyer Fundamental Representations, which are subject to Section 13.6(c).

- (c) The maximum aggregate liability or recovery of all Buyer Indemnified Parties from all Seller Indemnifying Parties under this Article 13 or otherwise pursuant to this Agreement, including for Fraud, shall not exceed an amount equal to the Final Purchase Price actually received by the Sellers. The maximum aggregate liability or recovery of all Seller Indemnified Parties from all Buyer Indemnifying Parties under this Article 13 or otherwise pursuant to this Agreement, including for Fraud, shall not exceed an amount equal to the Final Purchase Price actually received by the Sellers and the Sellers shall have no indemnification for Losses under this Article 13 by any Indemnified Party if such Liabilities are taken into account in the calculation of Net Working Capital. No Indemnified Party shall be entitled to be indemnified, paid or reimbursed more than once for the same Losses.
- 13.7 Exclusivity of Remedy for Monetary Damages. The Buyer and the Sellers acknowledge and agree that indemnification pursuant to this Article 13 shall be the sole and exclusive remedy of the Indemnified Parties for money damages for any and all Losses from and after the Closing in connection with any breach of a representation or warranty or non-performance of any covenant or agreement set forth in this Agreement; provided, that in the event of (a) any breach of the representation or warranty of the Company contained in Section 3.15, Buyer shall be entitled to elect to receive such services in accordance with Section 1.5 of the Transition Services Agreement and the Sellers shall have no further liability with respect to any related breach of representation or warranty of the Sellers contained in Section 3.15 hereunder and (b) any breach of any covenant or obligation of the Sellers set forth in Section 2.1, Buyer shall be entitled to elect to receive the transfer and delivery of such assets, rights or properties to cure such breach, and Sellers shall be promptly obligated to comply with such election. In the event such election is made and complied with, such amounts shall not be considered for any purposes set forth in Section 13.6 hereto.
- 13.8 Effect of Insurance and Other Recoveries. The amount of any Losses for which indemnification is provided under this Article 13 shall be reduced by any insurance proceeds actually received by the Indemnified Party or any of its Affiliates in connection with the facts giving rise to the right of indemnification (less the amount of the actual costs and expenses incurred in procuring such proceeds).
- 13.9 Effect of Investigation. The representations, warranties, covenants and agreements of the Indemnifying Party, and the Indemnified Party's right to indemnification with respect thereto, shall not be affected or deemed waived by reason of any investigation made by or on behalf of the Indemnified Party (including by any of its Representatives) or by reason of the fact that the Indemnified Party or any of its Representatives knew or should have known that any such representation or warranty is, was or might be inaccurate or that any such covenant or agreement is, was or might have been breached or not fulfilled or by reason of the Indemnified Party's waiver of any condition set forth in Section 9.2 or Section 9.3, as applicable.
- 13.10 <u>Mitigation</u>. The Buyer and each of the Sellers shall use its respective commercially reasonable efforts to mitigate any claim or liability that a Buyer Indemnified Party or a Seller Indemnified Party, as applicable, asserts under this <u>Article 13</u>. The Indemnified Party shall use reasonable best efforts to recover from insurance policies or other applicable sources of recovery the maximum portion of any Losses of such Indemnified Party.

13.11 No Recourse. This Agreement and the Ancillary Agreements may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the Ancillary Agreements or the transactions contemplated hereby may only be brought against, the Persons that are expressly named as parties hereto or thereto and then only with respect to, and to the extent of, the specific obligations set forth herein and therein with respect to such party. Except to the extent a named party to this Agreement or the Ancillary Agreements (and then only to the extent of the specific obligations undertaken by such named party in this Agreement or the Ancillary Agreements and not otherwise), no past, present or future equity holder, controlling person, director, officer, employee, agent, attorney, Affiliate, member, manager, general or limited partner, stockholder, investor or assignee of any party to this Agreement, nor any past, present or future equity holder, controlling person, director, officer, employee, agent, attorney, Affiliate, member, manager, general or limited partner, stockholder, investor or assignee of any of the foregoing, shall have any liability or obligation (whether in contract, tort, equity or otherwise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of the Sellers or the Buyer under this Agreement or the Ancillary Agreements (whether for indemnification or otherwise) of or for any claim based on, arising out of, or related to this Agreement, the Ancillary Agreements or the transactions contemplated hereby or thereby.

ARTICLE 14 MISCELLANEOUS

- 14.1 Fees and Expenses. Except as otherwise provided in this Agreement, each Party shall bear its own expenses and the expenses of its Subsidiaries in connection with the preparation, negotiation, execution and performance of this Agreement and the transactions contemplated herein, including all fees and expenses of any of its Representatives. Each Party shall each bear the fees and expenses of any broker, finder or investment bank retained by such Party or Parties and their respective Affiliates in connection with the Transaction. For the avoidance of doubt, the Purchase Price represents the consideration agreed among the Parties for the Transferred Assets and the Assumed Liabilities and does not include any costs or expenses, including fees and other charges, incurred in connection with the preparation, negotiation and execution of this Agreement and the consummation of the Transaction which shall be payable in accordance with this Section 14.1.
- 14.2 Further Actions. Each of the Parties hereto shall use reasonable best efforts to take, or cause to be taken, all appropriate action, do or cause to be done all things necessary, proper or advisable under applicable Law and execute and deliver such documents and other papers as may be required to consummate the Transaction and transfer all of the Transferred Assets and Assumed Liabilities. Notwithstanding anything to the contrary in this Agreement, no requirement to use "reasonable best efforts" under this Agreement shall require any Party or its Subsidiaries to pay any consent or similar fees to a Third Party or to agree to any adverse amendment to any Contract or any concession with any Third Party. For avoidance of doubt, this Agreement does not create any obligation on the Sellers to consummate the NTM Merger or take any actions to close the NTM Merger other than those provided in the NTM Merger Agreement and the Sellers shall not have any liability to Buyer as a result of any failure of the NTM Merger to be consummated or as a result of the termination of the NTM Merger Agreement in accordance with its terms.
- 14.3 <u>Governing Law</u>. This Agreement, and all claims or causes of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby, shall be governed by, and construed in accordance with, the Laws of the State of New York, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of Laws of another jurisdiction.

- 14.4 <u>Waiver of Jury Trial</u>. EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY ACTION ARISING OUT OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY HERETO (I) CERTIFIES THAT NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH PARTY WOULD NOT, IN THE EVENT OF ANY ACTION, SUIT OR PROCEEDING, SEEK TO ENFORCE THE FOREGOING WAIVER AND (II) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT, BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS <u>SECTION</u> 14.4.
- 14.5 <u>Submission to Jurisdiction</u>. Any Proceedings based upon, arising out of or related to this Agreement or the Transaction shall be brought exclusively in the federal courts located in the State of New York, and, if it has or can acquire jurisdiction, in the United States District Court for the Southern District of New York located therein, <u>provided</u>, <u>however</u>, that if such federal courts have finally determined that they do not have jurisdiction over such Proceeding, such Proceeding shall be heard and determined exclusively in any New York state court sitting in the Borough of Manhattan of The City of New York, and, in each case, appellate courts therefrom. Each of the Parties hereto irrevocably submits to the exclusive jurisdiction of each such court in any such Proceedings, waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, agrees that all claims in respect of the Proceedings shall be heard and determined only in any such court, and agrees not to bring any Proceedings arising out of or relating to this Agreement or the Transaction in any other court (including state court prior to the time that a final determination of non-jurisdiction has occurred). Nothing herein contained shall be deemed to affect the right of any Party to serve process in any manner permitted by Law or to commence Proceedings or otherwise proceed against any other Party in any other jurisdiction, in each case, to enforce judgments obtained in any Proceeding brought pursuant to this Section 14.5.
- 14.6 Specific Performance. The Parties acknowledge that, in view of the uniqueness of the Business and the transactions contemplated by this Agreement, each of the Sellers and the Buyer would not have an adequate remedy at law for money damages in the event that this Agreement has not been performed in accordance with its terms, and therefore agrees that, in addition to all other remedies available at law or in equity, the other Party shall be entitled to an injunction or injunctions to prevent or restrain breaches or threatened breaches of this Agreement by the other (as applicable), and to specifically enforce the terms and provisions of this Agreement to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of the other (as applicable). Each of the Sellers and the Buyer agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that any other Party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity. Any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with any such order or injunction.
- 14.7 <u>Amendment</u>. This Agreement may not be amended, modified or supplemented except upon the execution and delivery of a written agreement executed by the Parties.

14.8 No Assignment. The Buyer may assign its rights, interest or obligations under this Agreement to any of the direct or indirect Subsidiaries of Buyer, provided that (a) no such assignment shall relieve the Buyer of its obligations to the Sellers hereunder and (b) the assignment will not result in any incremental Taxes or other costs or expenses for which any Seller or any Affiliate of any Seller would be responsible, provided that with respect to clause (b), such Seller's or Affiliate's remedy shall be a reimbursement of such costs and expenses. Either Seller may assign its rights, interest or obligations under this Agreement to any of its direct or indirect Subsidiaries, provided that (a) no such assignment shall relieve such Seller of its obligations to the Buyer hereunder and (b) the assignment will not result in any incremental Taxes or other costs or expenses for which the Buyer or any of its Affiliate would be responsible, provided that with respect to clause (b), the Buyer's or such Affiliate's remedy shall be a reimbursement of such costs and expenses. Other than the preceding, neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any Party without the prior written consent of the Buyer, in the case of assignment by any Seller, and of the Sellers, in the case of any assignment by the Buyer. Subject to the preceding sentence, this Agreement will be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns.

14.9 <u>Waiver</u>. The Parties' rights and remedies are cumulative and not alternative. Any of the terms or conditions of this Agreement which may be lawfully waived may be waived in writing at any time by each Party which is entitled to the benefits thereof. Any waiver of any of the provisions of this Agreement by any Party shall be binding only if set forth in an instrument in writing signed on behalf of such Party. No failure or delay in exercising any right, power, or privilege under this Agreement or the documents referred to in this Agreement shall be deemed to or shall constitute a waiver of such right, power, or privilege, and no single or partial exercise of any such right, power, or privilege shall be deemed to or shall preclude any other or further exercise of such right, power, or privilege or the exercise of any other right, power, or privilege hereof.

14.10 Withholding. The Buyer shall be entitled to deduct and withhold from any payments to be made hereunder only such amounts, if any, that it is required to deduct and withhold on account of Taxes under applicable Law; provided, however, that Buyer shall use reasonable best efforts to give the Sellers written notice (including reasonable detail regarding the basis for such withholding and the amount of the proposed withholding) at least ten (10) Business Days before any such deduction or withholding and cooperate with the Sellers to mitigate or eliminate any such withholding to the extent permitted by applicable Law. To the extent any amounts are so deducted and withheld in accordance with the previous sentence, and duly and timely deposited with the appropriate Governmental Authority by the Buyer, such amounts shall be treated for purposes of this Agreement as having been paid to the Sellers in respect of which such deduction and withholding was made.

14.11 Notices. All notices hereunder will be in writing and in the English language and will be deemed to have been received (a) upon receipt of a registered letter, (b) the next Business Day following proper deposit with an internationally recognized express overnight delivery service or (c) upon receipt of an electronic transmission, upon confirmation of such receipt in writing (which may be via email) by the recipient thereof. Notices will be addressed as follows:

If to Sprint:

Sprint Corporation 6200 Sprint Parkway Overland Park, Kansas 66251 Attention: General Counsel

Email: office.chief.legal.officer@sprint.com

with a copy (which shall not constitute notice) to:

Morrison & Foerster LLP 425 Market Street San Francisco, California 94105 Attention: Brandon C. Parris Email: BParris@mofo.com

If to TMUS, or after the closing of the NTM Merger, to Sprint:

T-Mobile US, Inc. 12920 SE 38th Street Bellevue, Washington 98006 Attention: Dave Miller

Email: dave.miller@t-mobile.com

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue

New York, New York 10022

Attention: David Allinson, Josh Dubofsky, Tom Malone

Email: David.Allinson@lw.com, Josh.Dubofsky@lw.com, Thomas.Malone@lw.com

with a copy (which shall not constitute notice) to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019

Attention: Adam O. Emmerich, David K. Lam

Email: AOEmmerich@wlrk.com, DKLam@wlrk.com

If to the Buyer:

DISH Network Corporation 9601 S. Meridian Boulevard Englewood, CO 80112 Attn: General Counsel

Email: tim.messner@dish.com; jeffrey.blum@dish.com;

thomas.cullen@dish.com

with a copy (which shall not constitute notice) to:

Sullivan & Cromwell LLP 125 Broad Street New York, New York 10004 Attn: Scott Miller and Scott Crofton

Email: millersc@sullcrom.com; croftons@sullcrom.com

or to such other address as any Party may, from time to time, designate in a written notice given in like manner.

- 14.12 Complete Agreement. This Agreement and the other documents and writings referred to herein or delivered pursuant hereto (including the Ancillary Agreements and, with respect to Sprint and TMUS only, the NTM Merger Agreement) contain the entire understanding of the Parties with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings, both written and oral, between the Parties with respect to the subject matter hereof and thereof. This Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors and permitted assigns. Each of Sprint (for itself and its Affiliates and the Parent Affiliates) and TMUS (for itself and its Affiliates and the Parent Affiliates) agrees that the entry into, and performance of its respective obligations under, this Agreement, are consistent with and in furtherance of each party's obligations under the NTM Merger Agreement, and that such actions shall not (i) be deemed to amend, modify the terms of or expand the obligations of the parties under the NTM Merger Agreement, and (ii) so long as the performance of such party's obligations under this Agreement are in accordance with the terms of this Agreement, serve as a basis for a claim of breach of any applicable representation, warranty or covenant by such party in the NTM Merger Agreement.
- 14.13 <u>Publicity</u>. From and after the date hereof, the Parties will consult with each other and will mutually agree upon any publication or press release of any nature with respect to this Agreement or the Transaction and shall not issue any such publication or press release prior to such consultation and agreement except (i) as may be required by applicable Law or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange rule or regulation, (ii) to the extent the information contained in such publication or press release relating to this Agreement or the transactions contemplated hereby is substantially consistent with the information included in a press release or other public statement to which the Parties have previously mutually agreed. Notwithstanding the foregoing and for avoidance of doubt, this <u>Section 14.13</u> shall not limit TMUS's and Sprint's ability to issue publications or press releases relating to the NTM Merger, including publications or press releases that specifically reference the Transaction. The Parties agree that the initial press release to be issued with respect to the Transaction shall be in the form agreed by the Parties.
- 14.14 <u>Severability</u>. Any provision of this Agreement that is determined to be invalid, illegal or unenforceable by any court of competent jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction.
- 14.15 <u>Third Parties</u>. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any Person or corporation, other than the Parties and their permitted successors or assigns, any legal or equitable rights, remedies or claims under or by reason of this Agreement or any provision of this Agreement.

- 14.16 Waiver of Conflicts. Latham & Watkins LLP, DLA Piper, Cleary Gottlieb Steen & Hamilton LLP and Wachtell, Lipton, Rosen & Katz have acted as legal counsel to the TMUS, its Affiliates and its Parent Affiliates prior to the Closing and Morrison & Foerster LLP and Skadden, Arps, Slate, Meagher & Flom LLP has acted as legal counsel to Sprint, its Affiliates and its Parent Affiliates prior to the Closing (each such legal counsel, "Prior Counsel"). Each Prior Counsel intends to act as legal counsel to the Sellers, their respective Affiliates and their respective Parent Affiliates after the Closing. The Buyer hereby waives, on its own behalf and agrees to cause its Affiliates to waive, any conflicts that may arise in connection with Prior Counsel representing the Sellers, their respective Affiliates or their respective Parent Affiliates after the Closing to the extent such representation relates to the Business or the Transaction. All communications involving attorney-client confidences between the Sellers, their respective Affiliates and their respective Parent Affiliates and Prior Counsel in the course of the negotiation, documentation and consummation of the Transaction (the "Counsel Communications") shall be deemed to be attorney-client confidences that belong solely to the Sellers, their respective Affiliates and their respective Parent Affiliates. Accordingly, the Buyer shall not have access to any such communications, or to the files of Prior Counsel relating to its engagement, whether or not the Closing shall have occurred.
- 14.17 <u>Bulk Transfer Laws</u>. The Buyer hereby waives compliance by the Sellers and their respective Affiliates with the provisions of any bulk sales, bulk transfer or other similar Laws of any jurisdiction in connection with the Transaction.
- 14.18 No Other Duties. The only duties and obligations of the Parties under this Agreement are as specifically set forth in this Agreement, and no other duties or obligations shall be implied in fact, Law or equity, or under any principle of fiduciary obligation.
- 14.19 <u>Counterparts and Delivery</u>. This Agreement may be executed in two or more counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement. Any signed counterpart of this Agreement may be delivered by facsimile or other form of electronic transmission (e.g., pdf), with the same legal force and effect as delivery of an originally signed agreement.

[signature page follows]

IN WITNESS WHEREOF the parties hereto have caused this Agreement to be duly executed as of and on the date first above written.

T-MOBILE US, INC.

By: /s/ John J. Legere

Name: John J. Legere

Title: Chief Executive Officer

SPRINT CORPORATION

By: /s/ Marcelo Claure
Name: Marcelo Claure

Title: Executive Chairman

DISH NETWORK CORPORATION

By: /s/ Charles W. Ergen

Name: Charles W. Ergen

Title: Chairman

[Signature Page to Asset Purchase Agreement (Prepaid Business)]

Annex A

Table of Defined Terms

<u>Term</u>	Section
<u>"§"</u>	Section 1.3(o)
" <u>401(k) Plan</u> "	Section 1.1
" Accounting Firm "	Section 2.5(d)
" Acquisition Proposal "	Section 5.6
" Affiliate"	Section 1.1
" Agreed Amount "	Section 13.3(d)
"Agreement"	Preamble
" Allocation Notice of Objection"	Section 2.6(a)
"Amended and Restated Confidentiality Agreement"	Section 1.1
"Ancillary Agreements"	Section 1.1
" Antitrust Laws "	Section 1.1
"Articles"	Section 1.3(m)
" Asset Acquisition Statement "	Section 2.6(a)
" Assumed Liabilities "	Section 2.3(a)
"Bill of Sale and Assignment and Assumption Agreement"	Section 10.2(a)(i)
" Boost Business"	Section 1.1
" Business "	Section 1.1
"Business Confidential Information"	Section 12.1(c)
"Business Day"	Section 1.1
" Business Employee "	Section 6.1
"Buyer"	Preamble
" <u>Buyer 401(k) Plan</u> "	Section 6.6
" Buyer Cure Period"	Section 11.1(c)(i)
"Buyer Fundamental Representations"	Section 13.4
" Buyer Indemnified Parties"	Section 13.1
" Buyer Material Adverse Effect"	Section 4.3
" Buyer Plans"	Section 6.4
" Claim Notice "	Section 1.1
" Claimed Amount"	Section 1.1
" Closing Date"	Section 10.1
"Closing"	Section 10.1
"Code"	Section 1.1
"Company"	Recitals
" Confidential Information"	Section 12.1(b)
" Consent "	Section 5.3(a)
" Contract"	Section 1.1
" Controlling Party"	Section 1.1

<u>Term</u>	Section
"Counsel Communications"	Section 14.16
"Covered Taxes"	Section 7.3
"Deductible Amount"	Section 13.6
" <u>Disclosure Schedules</u> "	Article 3
" <u>DOJ</u> "	Section 1.1
"DOJ Consent"	Section 1.1
" DOJ Final Judgment"	Section 1.1
" <u>Dollar</u> "	Section 1.3(o)
"dollar"	Section 1.3(o)
" <u>DT</u> "	Definition of "Affiliate"
"Encumbrance"	Section 1.1
"ERISA"	Section 1.1
"Example Net Working Capital Statement"	Section 1.1
"Excluded Assets"	Section 2.2(b)
"Excluded Liability"	Section 2.3(b)
"Exhibits"	Section 1.3(m)
"Expected Claim Notice"	Section 1.1
" <u>FCC</u> "	Section 1.1
"FCC 214 Application"	Section 1.1
"FCC 214 Approval"	Section 1.1
"FCC Consent"	Section 1.1
"FCC MNSA Approval"	Section 1.1
"Federal Trade Commission Act"	Section 1.1
"Governing Document"	Section 1.1
"Governmental Authority"	Section 1.1
"Governmental Order"	Section 1.1
"HSR Act"	Section 1.1
"Indebtedness"	Section 1.1
"Indemnified Party"	Section 1.1
"Indemnifying Party"	Section 1.1
"Intellectual Property"	Section 1.1
"Inventory"	Section 1.1
"IRS"	Section 3.10(e)
"Knowledge of the Sellers"	Section 1.1
" <u>Law</u> "	Section 1.1
"Liabilities"	Section 1.1
" <u>Loss</u> "	Section 1.1
"Marks"	Section 1.1
"Material Adverse Effect"	Section 1.1
"MNSA"	Section 1.1
"Month Start Date"	Section 10.1

Term_	Section
"Non-Controlling Party"	Section 1.1
"Notice of Objection"	Section 2.5(c)(i)
"NTM Network"	Section 1.1
"NWC Statement"	Section 2.5(a)
"Ordinary Course of Business"	Section 1.1
"Parent Affiliates"	Section 1.1
"Parties"	Preamble
"Party"	Preamble
"Patents"	Section 1.1
"Permit"	Section 1.1
"Permitted Encumbrances"	Section 1.1
" <u>P</u> erson"	Section 1.1
"Plan"	Section 1.1
" Prior Counsel "	Section 14.16
"Proceedings"	Section 1.1
"Purchase Price"	Section 2.4
"Records"	Section 1.1
"Registered Intellectual Property"	Section 1.1
"Regulatory Approvals"	Section 1.1
"Rejected Inventory"	Section 5.11(a)
"Remedy Exceptions"	Section 3.2
"Representatives"	Section 1.1
"Response"	Section 13.3(d)
"Retention Agreements"	Section 6.8
"Schedules"	Section 1.3(m)
"Sections"	Section 1.3(m)
"Seller Fundamental Representations"	Section 13.4
"Seller Indemnified Parties"	Section 13.2
"Sellers Cure Period"	Section 11.1(b)(i)
"Sellers"	Preamble
" Short-Form IP Assignment Agreement"	Section 1.1
"SoftBank"	Definition of "Affiliate"
"Software"	Section 1.1
"Specified Assets"	Section 1.1
"Specified Liabilities"	Section 1.1
"Sprint Prepaid Business"	Section 1.1
"Straddle Period"	Section 7.3
"Subsidiary"	Section 1.1
"Tax Return"	Section 1.1
" <u>Tax</u> "	Section 1.1
"Taxes"	Section 1.1

<u>Term</u>	Section
"Taxing Authority"	Section 1.1
" Termination Date "	Section 1.1
" Terminating Buyer Breach "	Section 11.1(c)(i)
" Terminating Sellers Breach"	Section 11.1(b)(i)
" Third Party Claim."	Section 1.1
" Third Party "	Section 1.1
" Transaction "	Section 1.1
" <u>Transaction Bonuses</u> "	Section 1.1
" <u>Transaction Process</u> "	Section 1.1
" <u>Transfer Taxes</u> "	Section 7.4
" <u>Transferred Assets</u> "	Section 2.1(a)
" Transferred Contract"	Section 1.1
" <u>Transferred Employees</u> "	Section 6.1
"Transferred Intellectual Property"	Section 1.1
" Transferred Records"	Section 1.1
" Transition Services Agreement"	Section 1.1
" Virgin License"	Section 8.1(c)
" Virgin Mobile Business"	Section 1.1

AMENDMENT NO. 1 TO BUSINESS COMBINATION AGREEMENT

This AMENDMENT NO. 1 (this "<u>Amendment</u>"), dated as of July 26, 2019, to the BUSINESS COMBINATION AGREEMENT (the "<u>Agreement</u>"), dated as of April 29, 2018, by and among T-Mobile US, Inc., a Delaware corporation ("<u>T-Mobile</u>"), Huron Merger Sub LLC, a Delaware limited liability company ("<u>Merger Company</u>"), Superior Merger Sub Corporation, a Delaware corporation ("<u>Merger Sub</u>"), Sprint Corporation, a Delaware corporation ("<u>Sprint</u>"), Starburst I, Inc., a Delaware corporation ("<u>Starburst</u>"), Galaxy Investment Holdings, Inc., a Delaware corporation ("<u>Galaxy</u>"), and, for the limited purposes of the covenants and representations and warranties set forth in the Agreement that are expressly obligations of such persons, Deutsche Telekom AG, an *Aktiengesellschaft* organized and existing under the laws of the Federal Republic of Germany ("<u>DT</u>"), Deutsche Telekom Holding B.V., a *besloten vennootschap met beperkte aansprakelijkheid* organized and existing under the laws of the Netherlands ("<u>DT Holdco</u>"), and SoftBank Group Corp., a Japanese *kabushiki kaisha* ("<u>SoftBank</u>").

RECITALS

WHEREAS, Section 8.3 of the Agreement permits the parties to amend the Agreement by an instrument in writing signed on behalf of each of the parties hereto and duly approved by the parties' respective Boards of Directors (or comparable governing body) or a duly authorized committee thereof; and

WHEREAS, the parties hereto desire to amend the Agreement as provided herein.

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows:

ARTICLE I DEFINITIONS AND TERMS

1.1 <u>Definitions</u>. Unless otherwise specifically defined herein, each capitalized term used but not defined herein shall have the meaning given to such term in the Agreement. Each reference herein to "the date of this Amendment" shall refer to the date first set forth above and each reference to "the date of this Agreement" or similar references shall refer to April 29, 2018.

ARTICLE II AMENDMENTS TO THE AGREEMENT

2.1 Amendment to Section 1.3 of the Agreement. Section 1.3 of the Agreement shall be amended and restated in its entirety as follows:

Unless otherwise mutually agreed in writing between Sprint and T-Mobile, the closing of the Merger Transactions (the "Closing.") shall take place at the offices of Wachtell, Lipton, Rosen & Katz, 51 West 52nd Street, New York, New York 10019, at 9:00 a.m., New York time, on the first day of a calendar month, other than any month that is the third month of a calendar quarter (or, if such day is not a business day, the following business day), which such day is at least three (3) business days (the "Closing Date") following the day on which the last to be satisfied or waived (other than those conditions that are not legally permitted to be waived) of the conditions set forth in Article VII (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions) shall be satisfied or waived in accordance with this Agreement, or at such other time and place as may be mutually agreed in writing among the parties; provided, however, that if the Marketing Period has not ended at the time of the satisfaction or waiver of the conditions set forth in Article VII (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions), the Closing shall occur on the earlier to occur of (a) any date during or after the Marketing Period specified by T-Mobile on no less than three (3) business days' notice to Sprint; provided that to the extent that such date falls on a date that is after the Outside Date, T-Mobile shall obtain Sprint's prior written consent, and (b) the first day of a calendar month, other than any month that is the third month of a calendar quarter (or, if such day is not a business day, the following business day), which such day is at least three (3) business days following the final day of the Marketing Period.

- 2.2 <u>Amendment to Section 2.5(b) of the Agreement</u>. Section 2.5(b) of the Agreement shall be amended by adding, immediately following clause (ii) thereof, the words "and (iii) the authorized share count shall be 4,000,000,000 shares of common stock, par value \$0.01 per share."
- 2.3 Additional Provisions relating to Section 6.3 of the Agreement. Notwithstanding anything to the contrary in Section 6.3 or any other provision of the Agreement, from and after the date of this Amendment, none of T-Mobile, Merger Sub, Merger Company, either DT Party, Sprint or any SoftBank Party shall be required to take, or cause to be taken (or agree to take, or agree to cause to be taken) (and, unless requested by both T-Mobile and DT, neither Sprint nor any SoftBank Party shall take, or cause to be taken (or agree to take, or agree to cause to be taken)), any action authorizing, committing to, agreeing to, implementing or effecting, by consent decree, hold separate order, any agreement or letter (including any agreement or letter to address U.S. national security, law enforcement or public safety interests or to mitigate any foreign ownership, control or influence over the businesses of Sprint or T-Mobile), or otherwise, (x) the sale, divestiture, license or disposition of any of the assets or businesses of Sprint or T-Mobile or their respective subsidiaries or (y) restrictions, limitations or conditions, or actions that would limit Sprint or T-Mobile, or their respective subsidiaries, in its freedom of action or operations or ability to retain any businesses, product lines or assets, or any combination thereof, other than those matters expressly set forth on Schedule A to this Amendment, and the parties hereto expressly agree that any such action (including any matter set forth in clause (x) or (y)), other than those matters specifically set forth on Schedule A to this Amendment

(and the parties hereby agree that no matter set forth on Schedule A to this Amendment constitutes a "Regulatory Material Adverse Condition"). For the avoidance of doubt, any action that would constitute a Regulatory Material Adverse Condition (or any waiver of a Regulatory Material Adverse Condition) shall require the prior written consent of each of T-Mobile, Sprint, DT and SoftBank. In the event of any conflict or inconsistency between this Section 2.3 of this Amendment and any other provision of the Agreement or this Amendment, this Section 2.3 of this Amendment shall control.

2.4 <u>Amendment to Section 8.1(b)(i) of the Agreement</u>. Section 8.1(b)(i) of the Agreement shall be amended and restated in its entirety as follows:

if the Merger Transactions shall not have been consummated on or before November 1, 2019, or, if the Marketing Period has started and is in effect at such date, then January 2, 2020 (the "Outside Date"); provided, that the right to terminate this Agreement pursuant to this Section 8.1(b)(i) shall not be available to Sprint, if Sprint's or any SoftBank Party's, or T-Mobile, if T-Mobile's or any DT Party's, material breach of its obligations under this Agreement, the SoftBank Support Agreement or the DT Support Agreement, as applicable, has been the principal cause of, or principally resulted in, the failure of the Merger to be consummated by such time.

ARTICLE III ADDITIONAL REPRESENTATIONS AND WARRANTIES

- 3.1 Additional Representations and Warranties of T-Mobile, Merger Sub, Merger Company, DT and DT Holdco. Each of T-Mobile, Merger Sub, Merger Company, DT and DT Holdco represent and warrant to Sprint and SoftBank as follows: (i) such party has all requisite power and authority to execute, deliver and perform its obligations under this Amendment, (ii) the execution and delivery of this Amendment by such party have been duly authorized by all necessary action on the part of such party and (iii) this Amendment has been duly executed and delivered by such party and, assuming the due authorization, execution and delivery of this Amendment by the other parties hereto, constitutes the legal, valid and binding obligation of such party, enforceable against such party in accordance with its terms, except that such enforcement may be subject to the Enforceability Exceptions.
- 3.2 Additional Representations and Warranties of Sprint, Starburst, Galaxy and SoftBank. Each of Sprint, Starburst, Galaxy and SoftBank represent and warrant to T-Mobile, Merger Sub, Merger Company and the DT Parties as follows: (i) such party has all requisite power and authority to execute, deliver and perform its obligations under this Amendment, (ii) the execution and delivery of this Amendment by such party have been duly authorized by all necessary action on the part of such party and (iii) this Amendment has been duly executed and delivered by such party and, assuming the due authorization, execution and delivery of this Amendment by the other parties hereto, constitutes the legal, valid and binding obligation of such party, enforceable against such party in accordance with its terms, except that such enforcement may be subject to the Enforceability Exceptions.

ARTICLE IV MISCELLANEOUS AND GENERAL

- 4.1 <u>Ratification; No Further Amendment</u>. Except as expressly amended hereby, all terms, conditions and provisions of the Agreement shall remain unmodified and in full force and effect. This Amendment is limited precisely as written and shall not be deemed to be an amendment to any other term, condition or provision of the Agreement or any of the documents, schedules or exhibits referred to therein. Any amendment or modification of this Amendment shall require the prior written consent of each of T-Mobile, Sprint, the DT Parties and the SoftBank Parties.
- 4.2 Effect of Amendment . This Amendment shall form a part of the Agreement for all purposes. From and after the date of this Amendment, any reference in the Agreement to "this Agreement", "hereof", "herein", and "hereunder" and words or expressions of similar import shall refer to the Agreement as amended by this Amendment. The provisions of Section 8.3 (Amendment), Section 8.4 (Extension; Waiver) and Article X (General Provisions) of the Agreement shall apply *mutatis mutandis* to this Amendment, and to the Agreement as amended by this Amendment, taken together as a single agreement, reflecting the terms therein as modified hereby.
- 4.3 Entire Agreement. This Amendment and the Agreement (including the documents, exhibits, schedules and instruments referred to herein), taken together with the Confidentiality Agreement and the Roaming Agreement, constitute the entire agreement among the parties hereto, and supersede all prior agreements and understandings, both written and oral, among the parties hereto with respect to the Merger Transactions and the other transactions contemplated by the Agreement and this Amendment.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their duly authorized representatives as of the date first above written.

T-MOBILE US, INC.

By: /s/ John J. Legere

Name: John J. Legere

Title: Chief Executive Officer

SUPERIOR MERGER SUB CORPORATION

By: /s/ G. Michael Sievert

Name: G. Michael Sievert

Title: President

HURON MERGER SUB LLC

By: /s/ G. Michael Sievert

Name: G. Michael Sievert

Title: President

SPRINT CORPORATION

By: /s/ Marcelo Claure

Name: Marcelo Claure Title: Executive Chairman

DEUTSCHE TELEKOM AG

By: /s/ Dr. Christian P. Illek

Name: Dr. Christian P. Illek

Title: Board Member for Finance (Chief Financial

Officer)

By: /s/ Philipp Pohlmann

Name: Philipp Pohlmann

Title: Senior Vice President DT Finance US&GD

SOFTBANK GROUP CORP.

By: /s/ Masayoshi Son

Name: Masayoshi Son

Title: Chairman and Chief Executive Officer

STARBURST I, INC.

By: /s/ Ronald D. Fisher

Name: Ronald D. Fisher Title: President

GALAXY INVESTMENT HOLDINGS, INC.

By: /s/ Ronald D. Fisher

Name: Ronald D. Fisher Title: President

DEUTSCHE TELEKOM HOLDING B.V.

By: /s/ Frans Roose

Name: Frans Roose Title: Managing Director

By: /s/ Ton Zijlstra
Name: Ton Zijlstra
Title: Managing Director

T-Mobile and Sprint Receive Clearance from Department of Justice for Merger to Create the New T-Mobile

Companies reach agreements with DISH to divest all Sprint's pre-paid businesses including Boost Mobile and Virgin Mobile, and Sprint's 800 MHz spectrum licenses for aggregate proceeds of \$5B

\$43B in previously-announced synergies from T-Mobile/Sprint merger unchanged

Bellevue, Washington and Overland Park, Kansas – July 26, 2019 – T-Mobile US, Inc. (NASDAQ: TMUS) and Sprint Corporation (NYSE: S) today announced they have taken an important step toward completing their proposed merger to create the New T-Mobile by entering into a consent decree with the U.S. Department of Justice (DOJ). The DOJ action moves the merger one step closer to closing, pending other regulatory approvals and the satisfaction of other closing conditions.

The companies also announced agreements with DISH Network Corporation (NASDAQ: DISH) in which the combined T-Mobile and Sprint, as the proposed New T-Mobile, will divest Sprint's prepaid businesses and Sprint's 800 MHz spectrum assets to DISH. Additionally, upon the closing of the divestiture transaction, the companies will provide DISH wireless customers access to the New T-Mobile network for seven years and offer standard transition services arrangements to DISH during a transition period of up to three years. DISH will also have an option to take on leases for certain cell sites and retail locations that are decommissioned by the New T-Mobile, subject to any assignment restrictions. The transactions with DISH are contingent on the successful closing of T-Mobile's merger with Sprint among other closing conditions.

"The T-Mobile and Sprint merger we announced last April will create a bigger and bolder competitor than ever before — one that will deliver the most transformative 5G network in the country, lower prices, better quality, unmatched value and thousands of jobs, while unlocking an unprecedented \$43B net present value in synergies. We are pleased that our previously announced target synergies, profitability and long-term cash generation have not changed," said T-Mobile CEO and New T-Mobile CEO John Legere.

"Today marks an incredibly important step forward for the New T-Mobile. We are ready to bring this supercharged Un-carrier to consumers and businesses across the country, and this milestone brings us much closer to making that vision a reality for customers everywhere!" said Legere. "Our goal was to ensure that the DOJ's concerns were addressed while enabling us to deliver on every aspect of the synergies we promised to unlock... and we did it. It may have taken longer than expected by some, but today's results are a win-win for everyone involved. We cannot wait to get to work bringing this pro-consumer, pro-competition New T-Mobile 5G network to U.S. customers from coast to coast!"

"This is an important day for our country and, most important, American consumers and businesses," said Sprint Executive Chairman Marcelo Claure. "Today's clearance from the DOJ, along with our anticipated approval from the FCC, will allow the U.S. to fiercely compete for 5G leadership. We plan to build one of the world's most advanced 5G networks, which will massively revolutionize the way consumers and businesses use their connected devices to enhance their daily lives. The powerful combination of 5G, artificial intelligence and the Internet of Things will unleash endless possibilities."

New T-Mobile and DISH Agreements

Agreement to Divest Sprint's Prepaid Businesses

Effective upon the successful completion of T-Mobile's merger with Sprint, the New T-Mobile will be committed to divest Sprint's entire prepaid businesses including Boost Mobile, Virgin Mobile and Sprint-branded prepaid customers (excluding the Assurance brand Lifeline customers and the prepaid wireless customers of Shenandoah Telecommunications Company and Swiftel Communications, Inc.), to DISH for approximately \$1.4 billion. These brands serve approximately 9.3 million customers in total.

A greements Upon Closing of Prepaid Divestiture

Master Services Agreement for Network Access

With this agreement, Boost Mobile, Virgin Mobile, and Sprint-branded prepaid customers, as well as new DISH wireless customers, will have full access to the legacy Sprint network and the New T-Mobile network in a phased approach. Access to the New T-Mobile network will be through an MVNO arrangement, as well as through an Infrastructure MNO arrangement enabling roaming in certain areas until DISH's 5G network is built out.

Transition Services Agreement to Support Prepaid Customers

The New T-Mobile will offer standard transition services arrangements to DISH for up to three years following the close of the divestiture transaction. The transition services provided by the New T-Mobile will result in the orderly transfer of prepaid customers to DISH and will also ensure the continued and seamless operation of Boost Mobile, Virgin Mobile, and Sprint-branded prepaid businesses following transition to DISH's ownership.

Agreement to Divest Sprint's 800 MHz Spectrum Licenses to DISH

DISH has agreed to acquire Sprint's portfolio of nationwide 800 MHz spectrum for a total value of approximately \$3.6 billion in a transaction to be completed, subject to certain additional closing conditions, following an application for FCC approval to be filed three years following the closing of T-Mobile's merger with Sprint. This will permit the New T-Mobile to continue to serve legacy Sprint customers during network integration, pending later FCC approval of the license transfer. The companies have also entered into an agreement providing the New T-Mobile the option to lease back a portion of the spectrum sold to DISH for an additional two years following closing of the spectrum sale.

Option for DISH to Take Over Decommissioned Cell Sites and Retail Locations

Following the closing of T-Mobile's merger with Sprint and subsequent integration into the New T-Mobile, DISH will have the option to take on leases for certain cell sites and retail locations that are decommissioned by the New T-Mobile for five years following the closing of the divestiture transaction, subject to any assignment restrictions.

Agreement to Engage in Negotiations Regarding T-Mobile Leasing DISH's 600 MHz Spectrum

The companies have also committed to engage in good faith negotiations regarding the leasing of some or all of DISH's 600 MHz spectrum to T-Mobile.

The completion of the T-Mobile and Sprint combination remains subject to remaining regulatory approvals and certain other customary closing conditions. T-Mobile and Sprint expect to receive final federal regulatory approval in Q3 and currently anticipate that the merger will be permitted to close in the second half of 2019. Additional information can be found at www.NewTMobile.com.

Advisors

In connection with the DISH transaction, Goldman Sachs and PJT Partners acted as financial advisors to T-Mobile, and J.P. Morgan acted as financial advisor to Sprint, Latham & Watkins LLP acted as lead legal counsel to T-Mobile and Morrison & Foerster LLP acted as lead legal counsel to Sprint, with Wachtell, Lipton, Rosen & Katz, Cleary Gottlieb Steen & Hamilton LLP and DLA Piper also acting as counsel to T-Mobile in connection with the DISH transaction and the combination of T-Mobile and Sprint.

About T-Mobile

As America's Un-carrier, T-Mobile US, Inc. (NASDAQ: TMUS) is redefining the way consumers and businesses buy wireless services through leading product and service innovation. Our advanced nationwide 4G LTE network delivers outstanding wireless experiences to 83.1 million customers who are unwilling to compromise on quality and value. Based in Bellevue, Washington, T-Mobile US provides services through its subsidiaries and operates its flagship brands, T-Mobile and Metro by T-Mobile. For more information, please visit http://www.t-mobile.com.

About Sprint

Sprint (NYSE: S) is a communications services company that creates more and better ways to connect its customers to the things they care about most. Sprint served 54.5 million connections as of March 31, 2019 and is widely recognized for developing, engineering and deploying innovative technologies, including the first wireless 4G service from a national carrier in the United States; leading no-contract brands including Virgin Mobile USA, Boost Mobile, and Assurance Wireless; instant national and international push-to-talk capabilities; and a global Tier 1 Internet backbone. Today, Sprint's legacy of innovation and service continues with an increased investment to dramatically improve coverage, reliability, and speed across its nationwide network and commitment to a 5G mobile network in the U.S. You can learn more and visit Sprint at www.sprint.com or www.facebook.com/sprint and www.twitter.com/sprint.

Important Additional Information

In connection with the proposed transaction, T-Mobile US, Inc. ("T-Mobile") has filed a registration statement on Form S-4 (File No. 333-226435), which was declared effective by the U.S. Securities and Exchange Commission (the "SEC") on October 29, 2018, and which contains a joint consent solicitation statement of T-Mobile and Sprint Corporation ("Sprint"), that also constitutes a prospectus of T-Mobile (the "joint consent solicitation statement/prospectus"), and each party will file other documents regarding the proposed transaction with the SEC. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE JOINT CONSENT SOLICITATION STATEMENT/PROSPECTUS AND OTHER RELEVANT DOCUMENTS FILED WITH THE SEC WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. The documents filed by T-Mobile may be obtained free of charge at T-Mobile's website, at www.t-mobile.com, or at the SEC's website, at www.sec.gov, or from T-Mobile by requesting them by mail at T-Mobile US, Inc., Investor Relations, 1 Park Avenue, 14th Floor, New York, NY 10016, or by telephone at 212-358-3210. The documents filed by Sprint may be obtained free of charge at Sprint's website, at www.sprint.com, or at the SEC's website, at www.sec.gov, or from Sprint by requesting them by mail at Sprint Corporation, Shareholder Relations, 6200 Sprint Parkway, Mailstop KSOPHF0302-3B679, Overland Park, Kansas 66251, or by telephone at 913-794-1091.

No Offer or Solicitation

This communication shall not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended.

Cautionary Statement Regarding Forward-Looking Statements

This communication contains certain forward-looking statements concerning T-Mobile, Sprint and the proposed transaction between T-Mobile and Sprint. All statements other than statements of fact, including information concerning future results, are forward-looking statements. These forward-looking statements are generally identified by the words "anticipate," "estimate," "expect," "intend," "may," "could" or similar expressions. Such forward-looking statements include, but are not limited to, statements about the benefits of the proposed transaction, including anticipated future financial and operating results, synergies, accretion and growth rates, T-Mobile's, Sprint's and the combined company's plans, objectives, expectations and intentions, and the expected timing of completion of the proposed transaction. There are several factors which could cause actual plans and results to differ materially from those expressed or implied in forward-looking statements. Such factors include, but are not limited to, the failure to obtain, or delays in obtaining, required regulatory approvals, and the risk that such approvals may result in the imposition of conditions that could adversely affect the combined company or the expected benefits of the proposed transaction, or the failure to satisfy any of the other conditions to the proposed transaction on a timely basis or at all; the occurrence of events that may give rise to a right of one or both of the parties to terminate the business combination agreement; adverse effects on the market price of T-Mobile's or Sprint's common stock and on T-Mobile's or Sprint's operating results because of a failure to complete the proposed transaction in the anticipated timeframe or at all; inability to obtain the financing contemplated to be obtained in connection with the proposed transaction on the expected terms or timing or at all; the ability of T-Mobile, Sprint and the combined company to make payments on debt or to repay existing or future indebtedness when due or to comply with th

adverse changes in the ratings of T-Mobile's or Sprint's debt securities or adverse conditions in the credit markets; negative effects of the announcement, pendency or consummation of the transaction on the market price of T-Mobile's or Sprint's common stock and on T-Mobile's or Sprint's operating results, including as a result of changes in key customer, supplier, employee or other business relationships; significant transaction costs, including financing costs, and unknown liabilities; failure to realize the expected benefits and synergies of the proposed transaction in the expected timeframes or at all; costs or difficulties related to the integration of Sprint's network and operations into T-Mobile; the risk of litigation or regulatory actions, including the antitrust litigation brought by the attorneys general of thirteen states and the District of Columbia; the inability of T-Mobile, Sprint or the combined company to retain and hire key personnel; the risk that certain contractual restrictions contained in the business combination agreement during the pendency of the proposed transaction could adversely affect T-Mobile's or Sprint's ability to pursue business opportunities or strategic transactions; effects of changes in the regulatory environment in which T-Mobile and Sprint operate; changes in global, political, economic, business, competitive and market conditions; changes in tax and other laws and regulations; and other risks and uncertainties detailed in the Form S-4, as well as in T-Mobile's Annual Report on Form 10-K for the fiscal year ended December 31, 2018 and in its subsequent reports on Form 10-Q, including in the sections thereof captioned "Risk Factors" and "Cautionary Statement Regarding Forward-Looking Statements," as well as in its subsequent reports on Form 8-K, all of which are filed with the SEC and available at www.sec.gov and www.t-mobile.com, and in Sprint's Annual Report on Form 10-K for the fiscal year ended March 31, 2019 and in its subsequent reports on Form 10-Q, including in the sections thereof captioned "Risk Factors" and "MD&A — Forward-Looking Statements," as well as in its subsequent reports on Form 8-K, all of which are filed with the SEC and available at www.sec.gov and www.sprint.com. Forward-looking statements are based on current expectations and assumptions, which are subject to risks and uncertainties that may cause actual results to differ materially from those expressed in or implied by such forward-looking statements. Given these risks and uncertainties, persons reading this communication are cautioned not to place undue reliance on such forward-looking statements. T-Mobile and Sprint assume no obligation to update or revise the information contained in this communication (whether as a result of new information, future events or otherwise), except as required by applicable law.